Active Management of PV-Rich Low Voltage Networks

A thesis submitted to The University of Manchester for the degree of

Doctor of Philosophy

in

the Faculty of Science and Engineering

2017

Andreas T. Procopiou

Electrical Energy and Power Systems Group School of Electrical and Electronic Engineering The University of Manchester

Contents

	LIST OF FIGURES	7
	LIST OF TABLES	9
	ABSTRACT	10
	DECLARATION	11
	COPYRIGHT STATEMENT	12
	DEDICATION	14
	ACKNOWLEDGEMENT	15
1	FUTURE LOW VOLTAGE NETWORKS	16
	1.1 RESIDENTIAL-SCALE PV SYSTEMS	16
	1.2 FROM PASSIVE TO ACTIVE LV NETWORKS	18
	1.2.1 Technical issues faced in passive LV Networks	19
	1.2.2 Active LV Networks	21
	1.2.3 Centralised and Decentralised Control Approaches	21
	1.2.3.1 Centralised Control	21
	1.2.3.2 Decentralised Control	23
	1.2.4 Challenges	25
	1.2.4.1 Realistic Modelling and time-series Analyses	25 26
	1.2.4.2 Denavioural and Locational Uncertainties	20 27
	1.3 ACTIVE MANAGEMENT OF LV NETWORKS PROJECT	
	1.4 AIMS AND OBJECTIVES	
	1.5 MAIN CONTRIBUTIONS OF THE THESIS	29
	1.5.1 Management of Residential-scale PV Systems	29
	1.5.2 Management of Residential-scale Battery Energy Storage Systems	30
	1.5.3 Management of OLTC-fitted LV Transformers	
	1.5.4 List of Publications	
	1.5.4.1 Journal Papers	31
	1.5.4.2 Conference Papers	32
	1.5.4.3 Technical Reports	32
	1.6 Thesis Outline	33
2	MANAGING IMPACTS AND STATE OF THE ART	36
	2.1 INTRODUCTION	36
	2.3 RESIDENTIAL-SCALE PV SYSTEMS	39
	2.3.1 Embedded Active and Reactive Power Control Functions	39
	2.3.1.1 Active Power Limit	40
	2.3.1.2 Volt-Watt	40
	2.3.1.3 Volt-Var	41
	2.5.1.4 Reactive power capability of normal residential-scale inverters	42

	2.3.2 State of the Art	43
	2.3.2.1 Decentralised Control	43
	2.3.2.2 Centralised Control	50
	2.3.3 Summary of Gaps in Literature	53
	2.4 RESIDENTIAL-SCALE BATTERY ENERGY STORAGE SYSTEMS	55
	2.4.1 State of the Art	57
	2.4.1.1 Network-oriented Research	57
	2.4.1.2 Customer-oriented Research	61
	2.4.2 Summary of Gaps in Literature	62
	2.5 LV OLTC-FITTED TRANSFORMERS	64
	2.5.1 Transformer and OLTC Operating Principles	65
	2.5.1.1 Transformer Operating Principle	65
	2.5.1.2 OLTC Operating Principle	66
	2.5.2 State of the Art	67
	2.5.3 Summary of Gaps in Literature	71
3	CONTROL SCHEMES FOR PV-RICH LV NETWORKS	73
	3.1 INTRODUCTION	73
	3.2 RESIDENTIAL-SCALE PV SYSTEMS	73
	3.2.1 Decentralised Voltage Control	73
	3.2.1.1 Volt-Var Set-points	73
	3.2.1.2 Volt-Watt Set-points	
	3.2.2 Centralised Thermal Control (CTC)	76
	3.2.2.1 Control Architecture	76
	3.2.2.2 Control Logic	78
	3.2.3 Combined Centralised Thermal and Decentralised Voltage Control	83
	3.2.3.1 Control Logic	84
	3.3 RESIDENTIAL-SCALE BATTERY ENERGY STORAGE	85
	3.3.1 Normal Operation Mode	85
	3.3.1.1 Description	85
	3.3.1.2 Illustrative Example	86
	3.3.1.3 Limitations	87
	3.3.2 Advanced Operation Mode	88
	3.3.2.1 Illustrative Example	93
	3.4 LV OLTC-FITTED TRANSFORMERS	95
	3.4.1 Architecture	95
	3.4.2 Control Logic	96
4	CASE STUDY PART 1: MODELLING	99
	4.1 INTRODUCTION	99
	4.2 Real French LV Network	99
	4.2.1 Transformer	100
	4.3 REAL SMART METER DATA	100
	4.3.1 Construction of Domestic Load Profiles	102
	4.3.2 Allocation of Domestic Load Profiles	102
	4.4 REALISTIC PV GENERATION PROFILES	105
	AAI = Comparison with Paul Franch DV Concretion Profiles	105
	4.11 Comparison with Keat Fielder F V Ocheration Fromes	100
	4 4 1 2 Power Amplitude	107
	4.4.2 Allocation of Residential-scale PV Systems and Generation Profiles	108
	1.1.2 Anocation of Residential scale i v Systems and Generation i Tomes.	100

	4.4.2.1 Specifications of PV Inverters	109
	4.5 MONTE CARLO METHODOLOGY	110
	4.6 Performance Metrics	111
	4.6.1 Voltage Issues	112
	4.6.2 Thermal Issues	113
	4.6.3 Energy Metrics	113
	4.6.4 Control Actions	
5	CASE STUDY PART 2: SIMULATION ANALYSES FOR RESIDENTI	AL-
3	SCALE PV SYSTEMS	115
	5.1 INTRODUCTION	115
	5.2 DECENTRALISED VOLTAGE CONTROL	115
	5.2.1 Volt-Var Analysis	116
	5.2.1.1 Voltage Issues	117
	5.2.1.2 Thermal Issues	118
	5.2.1.3 Energy Metrics	120
	5.2.1.4 Summary	120
	5.2.2 Volt-Watt Analysis	121
	5.2.2.1 Voltage Issues	121
	5.2.2.2 Thermal Issues	
	5.2.2.3 Energy Metrics	
	5.2.2.4 Summary	
	5.3 CENTRALISED THERMAL CONTROL (CTC)	125
	5.3.1 Time-Series Performance	125
	5.3.2 Stochastic Analysis	128
	5.3.2.1 Voltage Issues	
	5.3.2.2 Thermal Issues	129
	5.3.2.3 Energy Metrics	
	5.5.2.4 Control Actions	133
	5.5.5 Summary	134
	5.4 COMBINED CENTRALISED THERMAL AND DECENTRALISED VOLTAGE	124
	CONTROL	134
	5.4.1 Time-Series Performance	135
	5.4.2 Stochastic Analysis	13/
	5.4.2.1 Voltage Issues	138
	5.4.2.2 Inermal Issues	
	5.4.2.5 Energy Metrics	1 1.39
	5.4.2.4 Control Actions	
	5.- . 5 - - - - - - - - - -	
	CASE STUDY PART 3: SIMULATION ANALYSES FOR RESIDENTI	AL-
0	SCALE BES SYSTEMS	
	6.1 INTRODUCTION	142
	6.2 TECHNICAL SPECIFICATIONS	
	6.2.1 BES Sizing	
	6.3 Additional Performance Metrics	
	6.3.1 Grid Dependency Index (GDI)	144
	6.3.2 LV Network Energy Imports	145
	633 State of Charge (SOC)	145
	634 Adopted Advanced Operation Mode Settings	145
	s.s.i Ruopica Ruvancea Operation mode beamgs	

	6.4 TIME-SERIES PERFORMANCE	145
	6.4.1 Normal Operation Mode	146
	6.4.2 Advanced Operation Mode	149
	6.5 STOCHASTIC ANALYSIS	152
	6.5.1 Voltage Issues	152
	6.5.2 Thermal Issues	154
	6.5.2 Fnergy Metrics	155
	654 Fnergy Imported	156
	6541 Grid Dependence Index (GDI)	158
	6.5.5 State of Charge (SOC)	159
	6.6 SUMMARY	161
		101
_	CASE STUDY PART 4: SIMULATION ANALYSES FOR LV OLTC-FIT	TED
7	TRANSFORMERS	163
	7.1 INTRODUCTION	163
	7.2 MODELLED LV OLTC-FITTED TRANSFORMER	163
	7.3 TIME-SERIES PERFORMANCE	165
	7.4 STOCHASTIC ANALYSIS	167
	7.4.1 Voltage Issues	167
	7.4.2 Control Actions	168
	7.5 Summary	169
	7.6 COMPARISON OF INVESTIGATED CONTROL SCHEMES	170
8	CONCLUSIONS AND FUTURE WORK	172
	8.1 INTRODUCTION	172
	8.2 RESEARCH AREA OVERVIEW AND MAIN GAPS	172
	8.3 THESIS MAIN CONTRIBUTIONS	174
	8.3.1 Management of Residential-scale PV Systems	175
	8.3.2 Management of Residential-scale Battery Energy Storage Systems	176
	8.3.3 Management of OLTC-fitted LV Transformers	176
	8.4 Conclusions	177
	8.4.1 Residential-Scale PV systems	177
	8.4.1.1 Decentralised Voltage Control	177
	8.4.1.2 Centralised Thermal Controller (CTC)	178
	8.4.1.3 Combined Centralised Thermal and Decentralised Voltage Control	179
	8.4.2 Management of LV OLTC fitted LV Transformers	1 /9
	8.4.5 Management of LV OLIC-fitted LV fransformers	101
	8.5 FUTURE WORK	101
	8.5.1 Seasonal Analyses and different representative LV networks	101
	8.5.2 Different BES systems and controls	182
	0.5.5 Economic Assessment.	182
	6.5.4 Consideration and combination of other LC18	183
	6.5.5 Comparison with Optimal Solution	183
	6.5.0 Combination of Proposed Active Network Management Schemes	183
	REFERENCES	195
		105
	APPENDIX	197

APPENDIX A	INVESTIGATED VOLT-VAR CURVES	. 198
APPENDIX B	INVESTIGATED VOLT-WATT CURVES	. 199

Word Count (main body): 45,274

Word Count (total): 51,879

List of Figures

Figure 1-1 Solar PV Global Capacity and Annual Additions, 2005-2015	17
Figure 1-2 Passive LV network without LCTs	19
Figure 1-3 Passive LV network with residential-scale PV systems	20
Figure 1-4 Active LV network	22
Figure 2-1 Voltage rise and drop on a passive network	37
Figure 2-2 Active Power Limit function	40
Figure 2-3 Generic Volt-Watt curve	41
Figure 2-4 Generic Volt-Var Curve	42
Figure 2-5 Power capability of normal residential-scale PV inverter	43
Figure 2-6 Investment cost of different voltage control strategies, on a German L	V
network [29]	46
Figure 2-7 Customer voltages with 50% penetration of PV systems without (left)	and
with Volt-Var control (right) [66]	47
Figure 2-8 Investigated scenarios in [38]	49
Figure 2-9 Interaction between residential-scale PV and BES system	55
Figure 2-10 Conventional storage strategy and proposed strategy in [45]	60
Figure 2-11 Voltage variation in radial distribution feeder [21]	65
Figure 2-12 Basic transformer operation (ideal transformer)	66
Figure 2-13 Voltage control using OLTC	67
Figure 3-1 Volt-Var curve adjustable settings	74
Figure 3-2 Volt-Watt curve adjustable settings	75
Figure 3-3 Monitoring and Control Architecture	77
Figure 3-4 Centralised Thermal Control Flowchart (Stage 1)	82
Figure 3-5 Centralised Thermal Control Flowchart (Stage 2)	83
Figure 3-6 Illustrative example of normal operation mode	86
Figure 3-7 Charging and discharging period definition	89
Figure 3-8 Advanced operation mode flowchart	92
Figure 3-9 Illustrative example of advanced operation mode	93
Figure 3-10 OLTC proposed control architecture	95
Figure 3-11 Controller example	97
Figure 3-12 OLTC control logic	98
Figure 4-1 French LV network	. 100
Figure 4-2 Composition of Smart Data by (a) Type and (c) Subscribed power	. 101
Figure 4-3 Distribution of EDF's customers among the different types [129]	. 102
Figure 4-4 Load Profile Creation Example	. 102
Figure 4-5 Smart meter data categorisation	. 103
Figure 4-6 Conversion of a load profile from (a) 30-min to (b) 1-min resolution	. 103
Figure 4-7 Daily PV profiles of 3 and 6kWp systems (a) January and (b) June	. 106
Figure 4-8 Capacity Factor for Real and Modelled PV Profiles	. 107
Figure 4-9 Minute to Minute % Difference of PV Power Amplitude	. 108
Figure 4-10 Monte Carlo Methodology	. 111
Figure 5-1 Reactive capability of 10% over-rated residential-scale PV inverter	. 116
Figure 5-2 Voltage issues with Volt-Var control	. 118
Figure 5-3 Transformer utilisation level with Volt-Var control	. 119
Figure 5-4 Feeder utilisation level with Volt-Var control	. 119
Figure 5-5 Total energy losses with Volt-Var control	. 120
Figure 5-6 Number of EN 50160 non-compliant customers	. 122
Figure 5-7 Transformer and feeder utilisation level with Volt-Watt	. 123

3
4
5
б
7
7
8
9
0
1
3
4
5
б
б
7
8
9
9
0
1
3
б
7
8
9
0
0
1
2
3
5
5
б
б
7
8
0
б
б
7
9

List of Tables

Table 6-2 Types of customers in France 10
Table 6-3 Number of connections for each type of customer [129]
Table 6-4 Modelled PV Inverter - Basic specifications 10
Table 6-1 Modelled BES System - Basic Specifications [81]14
Table 6-2 Network Energy Imports – BES Normal Operation
Table 6-3 Network Energy Imports and Exports - BES Advanced Operation 15
Table 7-1 Voltage regulation of the modelled LV OLTC-fitted transformer 16
Table 7-2 Control Schemes Summary 17

ABSTRACT

PhD Thesis: Active Management of PV-Rich Low Voltage Networks Andreas Procopiou, The University of Manchester, July 2017 Doctor of Philosophy (PhD)

The increased penetration of residential-scale photovoltaic (PV) systems in European-style low voltage (LV) networks (i.e., long feeders with high number of connected customers) is leading to technical issues such as voltage rise and thermal overload of the most expensive network assets (i.e., transformer, cables). As these issues significantly limit the ability of LV networks to accommodate higher PV penetrations, Distribution Network Operators (DNOs) are required to proceed with expensive and time-consuming investments in order to reinforce or replace these assets. In contrast to this traditional approach of network reinforcement, which potentially leads to massive capital expenditure, the transition towards active LV networks where controllable elements, existing (i.e., PV systems) and likely to be adopted (i.e., battery energy storage systems, LV on-load tap changer transformers), can be managed in real-time, poses an attractive alternative.

Although several active network management schemes have been recently proposed to increase the hosting capacity of PV-rich LV networks, they are mostly based on managing voltage issues only; and, in general, aim to solve technical issues separately. Integrated solutions aiming at managing simultaneously voltage and thermal issues are required, as recent studies demonstrate that both issues can coexist in PV-rich LV networks. More importantly the majority of studies, which commonly neglect the characteristics of real LV networks (e.g., unbalanced, three-phase, radial, multiple feeders with several branches, different types of customers), use complex optimisation techniques that require expensive communication infrastructure and extensive or full network observability (currently not available in LV networks). However, considering the extensiveness of LV networks around the world, practical, cost-effective and scalable solutions that use limited and already available information are more likely to be adopted by the industry.

Considering the above gaps in the literature, this Thesis contributes by proposing innovative and scalable active network management schemes that use limited network monitoring and communication infrastructure to actively manage (1) Residential-scale PV systems, (2) Residential-scale Battery Energy Storage (BES) systems and (3) LV on-load tap changer (OLTC)-fitted transformers. The adoption of the proposed active network management schemes, which makes use of already available devices, information and requires limited monitoring (i.e., secondary distribution substation), allows making the transition towards active LV networks more practical and cost-effective.

In addition, to tackle the challenges related to this research (i.e., lack of realistic LV network modelling with high resolution time-series analyses), this Thesis, being part of the industrial project "Active Management of LV Networks" (funded by EDF R&D) and having access to French data, contributes by considering a fully modelled typical real residential French LV network (three-phase four-wire) with different characteristics and number of customers. Moreover, realistic (1-min resolution) daily time-series household (from real smart meter data) and PV generation profiles are considered while a stochastic approach (i.e., Monte Carlo) is adopted to cater for the uncertainties related to household demand as well as PV generation and location.

DECLARATION

No portion of the work referred to in this thesis has been submitted in support of an application for another degree or qualification of this or any other university or other institute of learning.

COPYRIGHT STATEMENT

i. The author of this thesis (including any appendices and/or schedules to this thesis) owns certain copyright or related rights in it (the —Copyright) and he has given The University of Manchester certain rights to use such Copyright, including for administrative purposes.

ii. Copies of this thesis, either in full or in extracts and whether in hard or electronic copy, may be made only in accordance with the Copyright, Designs and Patents Act 1988 (as amended) and regulations issued under it or, where appropriate, in accordance with licensing agreements which the University has from time to time. This page must form part of any such copies made.

iii. The ownership of certain copyright, patents, designs, trademarks and other intellectual property (the —Intellectual Property) and any reproductions of copyright works in the thesis, for example graphs and tables (—Reproductions), which may be described in this thesis, may not be owned by the author and may be owned by third parties. Such Intellectual Property and Reproductions cannot and must not be made available for use without the prior written permission of the owner(s) of the relevant Intellectual Property and/or Reproductions.

iv. Further information on the conditions under which disclosure, publication and commercialisation of this thesis, the Copyright and any Intellectual Property and/or Reproductions described in it may take place is available in the University IP Policy, in any relevant Thesis restriction declarations deposited in the University Library, The University Library's regulations and in The University's policy on presentation of Theses.

"Let the future tell the truth, and evaluate each one according to his work and accomplishments. The present is theirs; the future, for which I have really worked, is mine"

– Nikola Tesla

DEDICATION

To Maria,

my loving wife,

who always picked me up on time and encouraged me to go on every adventure and her love and confidence is my constant source of inspiration and encouragement.

To Tasos and Skevi,

my wonderful parents, who have raised me to be the person I am today and their love and support gave me strength to chase my dreams.

ACKNOWLEDGEMENT

I deeply wish to thank my supervisor, Professor Luis F. Ochoa, the most amazing leader I ever met in my life. His guidance and the feedback he provided me, as well as his experience, have positively influenced not only the document at hand but, also, my way of thinking as an engineer and as a person.

Special thanks and gratitude to my industrial sponsor, Électricité de France (EDF) R&D, and especially to the awesome people, Luc, Leticia, Soufien, Mathieu, Sebastien, Belen, Mathilde, Thierry, Borris and Guillaume, who I worked with at the department of Measurements and Information Systems of Electrical Networks (MIRE) in Saclay (France).

Finally, I would like to thank all of my team and friends at the University of Manchester for their endless support and friendship during the past four years. You mean a lot to me and I value each one of you: Andrea, Jairo, Kyriacos, Luis, Christos, Chao, Tuba, Fredy, Sahban, Alejandro, Tiago, Jonathan, Sait, Zedong, William, Valentin, Francesco, Mathaios, Vidy, Dimitris and Michalis.

Thank you all!

Chapter 1

Future Low Voltage Networks

One of the main challenges our society is facing today is the reduction of carbon emissions. As a result, during the last years, major efforts to increase environmental awareness have been made with governments all over the world and, in many cases, introducing targets to decrease emissions. The European Union (EU), for example, has set a target to source 20% of its total consumption from renewables by 2020 and 27% by 2030 [1]. To achieve these challenging targets, European countries have promoted through different incentives the adoption of residential-scale low carbon technologies (LCTs). These technologies are related to distributed generation, such as residential-scale photovoltaic (PV) systems, electro-thermal technologies, such as the electric heat pumps (EHPs), and the electrification of transport, such as the electric vehicles (EVs).

1.1 Residential-scale PV Systems

Residential-scale PV systems, in particular, became one of the most rapidly adopted LCTs in low voltage (LV) distribution networks due to the significant cost reduction of the technology itself and the incentives (e.g., feed in tariff schemes, subsidies) rolled out in several countries around the world [2].

Indeed, Figure 1-1, which shows the global annual PV additions and capacity from 2005 to 2015 [2], demonstrates that PV technology is getting more prevalent each year. Looking back a decade, the annual addition of PV capacity (i.e., 5.1GW) in 2005 increased by ten times (i.e., 50GW) in 2015 and more than half of the total capacity was installed within the last three years. Moreover, out of the total 227GW of global PV capacity, almost half (i.e., 100GW) is installed within European countries.



Figure 1-1 Solar PV Global Capacity and Annual Additions, 2005-2015

Germany for example, who very early introduced attractive incentives for PV system installations is leading in Europe and the world with a total of 40GW installed PV capacity by the mid of 2016 from where the 74% are considered to be residential-scale PV system installations [3, 4]. The current capacity, which corresponds to the 6.4% of total German gross power consumption [5], is expected to increase up to 52GW to achieve the government's target of having 10% of total power generation coming from PV systems by 2020.

Another example, France, which is within the top four EU countries with the highest installed PV capacity [6], has already reached 6.9GW by the mid of 2016 [7]. Out of the total PV installations in the mainland of France, the majority (91.1%) of them are installed in LV networks with sizes ranging between 3 and 9kWp. These numbers are expected to increase even more aiming to achieve the government's recently reviewed energy goals to triple the current PV installed capacity by the end of 2023 reaching 20.2GW [8].

The installation of residential-scale PV systems has also been increased in countries where there is not as much of PV potential (i.e., low sun radiation) but attractive incentives exist. For example, as a result of the UK Government's feed-in-tariff program, the total installed PV capacity in the UK reached 11GW by the end of September 2016. Interestingly, over 7GW of PV capacity was installed during the last two years which clearly demonstrates the rapid adoption of the technology [9]

given the ambitious aim of the Government to double (i.e., 22GW) the capacity by 2020 [10].

Given the numerous advantages that the PV technology is offering (i.e., simple to install, non-polluting, emits no noise and requires little maintenance) [11] and the strong will of European countries to meet their renewable energy targets, the penetration levels of residential-scale PV systems in LV networks are expected to increase even more. Indeed, according to the European Photovoltaic Industry Association (EPIA), the total installed PV capacity, in Europe, is expected to increase by 50% in 2018 [12]. LV networks, however, have been traditionally designed without controllable elements (i.e., passive), assuming unidirectional power flow (i.e., from the power plant to the customer) and with no provision of PV systems increase, technical issues (e.g., voltage rise, thermal overload of assets) are expected to occur; thus, limiting the network's ability to host more LCTs [13-15].

1.2 From Passive to Active LV Networks

As mentioned above and shown in Figure 1-2, LV networks have been traditionally designed and operated as passive circuits assuming a unidirectional power flow, where the power flows (i.e., blue arrows) from the Bulk Supply Point then through the transmission, distribution and finally reaching the LV customers (i.e., households) which are connected and supplied through the LV feeders (i.e., overhead or underground cables) of the corresponding LV network [16-19]. All feeders of an LV network are connected at the same point, the LV busbar, which is located at the secondary distribution substation (i.e., secondary side of the distribution transformer) and is also the final stage of the delivery of electric power.

Due to the limited observability and controllability closer to the customers (i.e., lower voltage levels), existing LV networks were designed according to the worst case loading conditions (i.e., maximum demand) without considering any kind of LCT and in particular PV systems.

Nowadays, as illustrated in Figure 1-3, with the increasing adoption of residentialscale PV systems on the consumers' side (i.e., LV networks), the traditionally assumed unidirectional power flow system is changing to a bidirectional one where



Figure 1-2 Passive LV network without LCTs

both consumers and producers interact with the LV network. However, the presence of a large number of residential-scale PV systems in LV networks, and in particular those designed considering relatively small loads (i.e., heating coming from gas and no air-conditioning systems), may result in some technical issues (i.e., voltage rise, feeder and transformer thermal overloads), particularly during periods of maximum generation and minimum demand [15, 20, 21].

1.2.1 Technical issues faced in passive LV Networks

Due to the low reactance to resistance ratio (X/R) in distribution networks, and particularly in LV (i.e., ≤ 0.5), cable resistances are high which makes them prone to voltage drops and rises when current is flowing through them. Voltage rise, for example, is considered to be one of the most dominant technical issues in LV networks [15, 22, 23] as it occurs due to the fact that residential-scale PV systems are injecting current to the network that results in significant reverse current flows (opposite direction, towards the substation). For simplicity and better understanding, considering the fundamental equation of Ohm's law, $V = I \times R$, a negative current flow will lead to a negative voltage drop meaning that the voltage will rise.



Low Voltage Network

Figure 1-3 Passive LV network with residential-scale PV systems

In power systems terms, the power generated from residential-scale PV systems (green arrows) is reducing the customer's load demand which is then increasing the voltage at the PV connection point. This is also illustrated in Figure 1-3 where the demand supplied to the customers (blue arrows) is significantly lower compared to Figure 1-2. However, as the penetration of residential-scale PV systems increases in LV networks (i.e., significant current flowing in the opposite direction), voltages can rise above the statutory limit which in turn limits the hosting capacity of the network. Additionally, high voltages can significantly reduce the lifespan of most household appliances [24] while also increasing the risk of damaging sensitive electronic equipment that may be interconnected to the network (e.g., EVs, wind turbines, PV panels).

In an even worse scenario, high penetrations of local generation (i.e., residentialscale PV systems) can significantly exceed the local demand resulting to large amounts of reverse power flow (green arrows). This phenomenon, however, might result to the thermal overloading of the most important and expensive assets of the network (i.e., feeder cables and transformer) [15, 23]. To adequately cater for the aforementioned technical issues, Distribution Network Operators (DNOs) are now required to proceed with costly investments in order to reinforce the existing network assets (i.e., larger transformers and feeder cables). Hence, this passive "fit & forget" connection approach represents a barrier to the adoption of high penetrations of low carbon technologies and, in particular, residential-scale PV systems.

In contrast to the network reinforcement which will potentially lead to massive capital expenditure on the ageing network, the transition towards active LV networks where controllable elements can actively be managed, poses an attractive alternative. The vast opportunity to use monitoring data and automation to avoid unnecessary maintenance or replacement of the assets can help to increase the hosting capacity while using the existing LV network infrastructure.

1.2.2 Active LV Networks

To increase the hosting capacity of LV networks and hence enable the acceleration of the decarbonisation of the electricity distribution sector, efficient and cost-effective solutions that reduce the need of expensive and time consuming network reinforcements are required. To achieve this, DNOs are required to move from passive to active LV networks, where controllable devices and low carbon technologies are actively managed to solve technical issues such as voltage rise/drop and thermal overloads [22, 25, 26].

1.2.3 Centralised and Decentralised Control Approaches

Emerging LCTs offer numerous controllability options which can be used in either a "centralised" (i.e., transformer as the 'hub') or "decentralised" (local) way to manage voltage and thermal issues.

1.2.3.1 Centralised Control

Considering a "centralised" control approach, a central controller (i.e., 'hub' controller), located at the secondary distribution substation, can obtain data from several remote monitoring points across the LV network. Gathering all monitoring data at the 'hub' controller allows for a better visualisation of the LV network state

Active Management of PV-Rich Low Voltage Networks



Figure 1-4 Active LV network

which then enables the provision of adequate control signals to the corresponding LCTs and control devices to manage potential voltage and/or thermal issues.

To provide an example, Figure 1-4 demonstrates a simple schematic of an active LV network where depending on the control cycle (i.e., data sampling rate), monitoring data from across the LV network can be sent through communication devices (either wireless or wired) to the "hub" controller. Once the data are received, the "hub" controller would then process it and based on a control algorithm can instruct PV systems to operate (e.g. reduce maximum generation output) in a way that improves the LV network constraints and satisfies an operational goal (e.g., thermal and voltage limits). Most studies found in the literature (discussed in section 2.3.2), due to the increased complexity (i.e., large number of PV systems with different ratings, location and behaviour of LV customers), apply complex optimisation algorithms to calculate optimal settings for the PV systems. This, however, requires extensive LV network visibility and information (i.e., network topology) which is not usually available to DNOs, making the application of such approaches difficult to

implement. Thus, centralised control approaches that are scalable and practical to implement need to be developed.

Under the same control approach (i.e., centralised) the use of even more mature technologies, commonly used in higher voltage levels, can also be adopted in LV networks. Such technologies are the on-load tap changers (OLTCs) which can be adopted at secondary distribution substations (i.e., MV/LV) and actively managed to alleviate voltage issues, hence increasing the network's hosting LCT capacity and reducing the need of PV generation curtailment [27]. Although the use of LV OLTCfitted transformers has been recently studied in order to increase the flexibility of voltage management closer to LV customers [28, 29], traditional OLTC control concepts (i.e., fixed voltage target) might not be adequate to manage voltages issues as the voltage level among different feeders (connected at the same transformer) might vary due to the particular characteristics of LV networks (i.e., uneven LCT penetration, load unbalance, etc.). Therefore, remote monitoring data closer to the LV customers and in particular critical network points (i.e., farthest customer connections) is required to visualise the state of the network thus allowing taking more accurate decisions. Moreover, considering that contrasting voltages might occur in LV networks (i.e., one feeder with high PV penetration and other heavily loaded) OLTC control methods able to satisfy both issues need to be investigated.

Despite the benefits provided by a centralised control approach, the lack of observability in LV networks requires a significant capital investment in information and communication infrastructure (ICT) that allows a two-way communication between the "hub" controller and the corresponding control and remote monitoring devices. Furthermore, the corresponding challenges associated with the increased amount of remote monitoring data should be taken into account as the cost of gathering, storing, and analysing them can be significant; making the centralised control approach a less suitable solution for DNOs.

1.2.3.2 Decentralised Control

On the other hand, adopting a "decentralised" control approach stands as another attractive alternative allowing controllable devices to operate autonomously based on their local controller that responds to local network conditions (e.g., voltage level) and requirements (e.g., statutory limits). Decentralised control approaches offer the advantage of "plug and play" as there is no need of a central controller and more importantly, as actions are taken locally, no communication infrastructure is required making the adoption of such an approach cost-effective.

Under this control approach (i.e., decentralised), modern inverters installed in residential-scale PV systems can be used as they have a fast (i.e., fraction of a second) and flexible internal control mechanism that can implement several local control functions [30] aiming to mitigate technical issues (i.e., voltage rise/drop, transformer and feeder thermal overloads), thus increasing the ability of LV networks to host more LCTs. Such decentralised control functions are the Volt-Var and Volt-Watt control of PV inverters which can adjust the reactive and active power output of the inverter based on the voltage level at the connection point of the PV system, respectively. Although the adoption of decentralised control of residentialscale PV inverters has been investigated in the literature (discussed in section 2.3.2.1), a significant number of these studies [31-35] apply complex optimisation algorithms to calculate the optimal settings for each PV inverter active and reactive power controller. Despite the need of extensive information/visibility of the network which is not commonly available, individual settings for each inverter are required, increasing the implementation challenges. Moreover, several studies [36-38] assume PV inverter ratings that can be high enough to maintain the full solar output (kWp) and the requested vars. In reality, however, residential-scale PV inverters are currently sized economically to the end user (not for network support) and therefore might have limited capability to produce or absorb the required reactive power (i.e., vars) in periods of high power output (i.e., sunny weather condition) as the available reactive power headroom is varying throughout the day (i.e., variable solar irradiance).

Consequently, the adoption of these decentralised PV control functions (i.e., Volt-Var, Volt-Watt) has to be thoroughly investigated considering the corresponding specifications or limitations (e.g., limited reactive capability) associated with the commercially available PV systems. More importantly, scalable and easy to implement solutions, such as the use of universal PV inverter settings, are required to accelerate the adoption of these decentralised control functions that will help manage technical issues.

Additional flexibility can be achieved with residential-scale battery energy storage (BES) systems which are considered to be one of the key technologies in future LV networks [39]. Recently, residential-scale BES systems (up to ~20 kWh) have been introduced by several manufacturers (e.g., Tesla, Siemens, ABB) with the primary goal of helping householders maximise the usage of PV generation by storing excess generation during the day to use later at night [39, 40]. However, BES systems have the potential –yet untapped– to mitigate impacts associated with the excess of PV generation (e.g., voltage and thermal issues), thus providing DNOs with an alternative solution to reinforcement. Although the adoption of residential-scale BES systems has been introduced only just recently, several studies try to address this issue [41-46]. However, they focus only on DNO benefits (i.e., managing voltage and thermal issues) without considering the primary goal of customers (i.e., lower grid dependency). Therefore, as BES systems are expected to be attached to residential-scale PV systems, it is important to develop control methods that can provide benefits to both customers (i.e., lower grid dependency) and DNOs (i.e., managing voltage and thermal issues).

1.2.4 Challenges

This section identifies and discusses the main challenges associated with the research on the active management of PV-rich LV networks.

1.2.4.1 Detailed Modelling and time-series Analyses

With the increasing uptake of residential-scale PV systems there is a clear need to study their impacts in LV networks while also developing suitable active network management solutions to mitigate those. To successfully perform these studies, detailed, close to reality models of LV networks are required as they allow capturing their particularities (e.g., load unbalance, cable characteristics) compared to synthetic or generic models. However, the passive nature of LV networks (i.e., neither monitored nor controlled) as well as the fact that DNOs have historically worked with synthetic (or generic) networks (to assess the requirements of the LV network

planning stage) it is uncommon for the DNOs to hold accurate network data. This poses a critical challenge in realistic modelling.

Along with the above, traditionally performed (by DNOs and researchers) power flow analyses considering only snapshots of load and generation profiles pose another significant challenge when analysing LV networks as these are unable to capture the time-series behaviour of customer demand and PV generation. It is therefore important to perform realistic time-series analyses that allow not only capturing the behaviour of demand and generation but also their correlation through the day. Hence, high granularity (e.g., scale of seconds and minutes) time-series demand and generation profiles are essential to allow realistic modelling and timeseries analyses.

This PhD project, however, as part of the industrial project "Active Management of LV Networks" funded by the Électricité de France (EDF) R&D (i.e., the largest R&D centre in Europe), had access to real French data (i.e., network, demand and generation) which allowed to carry out realistic studies and cover all the aforementioned challenges. Hence, this research, taking advantage of these data, uses a real typical residential, French LV network (three-phase four-wire) with different characteristics and number of customers, adopting realistic (1-min resolution) daily time-series household (from real smart meter data) and PV profiles.

1.2.4.2 Behavioural and Locational Uncertainties

The rapid adoption of residential-scale LCTs and particularly PV systems, in future LV distribution networks raises significant challenges due to their behavioural and locational uncertainties [15]. Consequently, commonly adopted deterministic analyses or specific scenarios (i.e., worst case) are not suitable considering the intermittent behaviour of load and PV generation. Hence, to cope with the corresponding uncertainties and to truly understand the impacts on the electricity infrastructure to which the PV systems will connect to, as well as potential solutions, stochastic approaches must be adopted.

To tackle this challenge and cater for the uncertainties related to household demand as well as PV generation and location, this PhD Thesis adopts a stochastic approach (i.e., Monte Carlo methodology). This allows extracting more meaningful results as well as understanding the extent to which an investigated active network management control method can cope under different network scenarios.

1.2.4.3 Network Observability

To effectively manage LV networks, extended visibility is required. This will enable DNOs to better understand the state of the networks and therefore utilise their assets more efficiently in real time. Unfortunately, LV networks as of today (i.e., passive), are not equipped with monitoring or communication devices [47], thus making the adoption of active management techniques even more challenging. Although the already planned and on-going installation of advanced metering infrastructure (AMI) technologies (i.e., smart meters) in many European countries [48] might cover this gap of observability in LV networks, it is expected to take several years more to be fully implemented, hence allowing DNOs to use measurement data in real-time. Furthermore, it is uncertain whether AMI data can actually be used for operational purposes as this depends on how and who manages all these data. Thus interim solutions are required to enable the adoption of active management techniques in LV networks.

For the above reason, in the last few years, several DNOs have shown an interest in increasing the network visibility closer to the customers by installing remote monitoring devices at critical points (i.e., feeders' end points) within their LV networks [47, 49]. The use of remote monitoring devices and the corresponding communication infrastructure, although ensuring accurate visibility of the LV network, results in higher capital and operational costs. Therefore, considering the extensiveness of LV networks around the world (e.g., circa 500,000 in the UK), the installation of monitoring devices might not be economically feasible [50, 51].

Consequently, it is essential to explore the extent to which limited observability with existing (or installed) secondary substation monitoring (i.e., LV busbar) could be used to augment the network visibility, particularly at critical points of the LV network. Considering the above, this PhD Thesis aims to provide practical and scalable solutions to actively manage PV-rich LV networks using limited observability (e.g., LV busbar, critical points) and available network information.

1.3 Active Management of LV Networks Project

The work presented in this Thesis has been sponsored by the industrial project "Active Management of LV Networks". This was a three and half year project funded by the Électricité de France R&D (i.e., the largest R&D centre in Europe) and run by The University of Manchester under the Industrial Cooperative Awards in Science & Technology (iCASE) fund. The aim of this project was to investigate the effect of applying active management techniques to increase the ability of LV networks to accommodate higher number of LCTs, in particular, residential-scale PV systems and EVs. The project was divided into three main tasks: (1) the coordinated control of residential-scale LCTs (i.e., PV systems, EVs), (2) the control of residential-scale BES systems and (3) the utilisation of LV OLTC-fitted transformers to control voltages and congestion in LV networks. More importantly, the project provided the unique opportunity to access real French LV data (i.e., network, demand and PV generation) that allowed carrying detailed realistic studies.

1.4 Aims and Objectives

The adoption of LCTs in LV networks can help achieve the targets set by many countries to reduce the carbon emissions. Despite the benefits gained, the continuously increasing penetration of LCTs, and, in particular, PV systems, can also result in technical issues (i.e., voltage rise, thermal overloads), hence limiting the number of LCTs that can be connected. In this context, and consistent with the objective of the Active Management of LV Networks (but not limited to it), the main purpose of this Thesis is to provide innovative, practical and scalable solutions to actively manage PV-rich LV networks in order to increase their hosting capacity. Thus, to achieve the main purpose stated above, the following objectives have been established:

- 1. The use of realistic models of LV networks from real data (GIS) to perform the analyses.
- 2. The creation of realistic daily high resolution load (based on real smart meter data) and PV generation profiles in order to carry out time-series analyses.

- 3. The investigation, development and implementation of scalable solutions to actively manage residential-scale PV systems (in either a decentralised or centralised way) in order to solve technical issues in LV networks.
- 4. The investigation, development and implementation of scalable solution to actively manage residential-scale BES systems (installed along with PV systems) in order to solve technical issues in LV networks.
- 5. The utilisation of OLTC-fitted LV transformers to manage voltage issues in LV networks considering both local monitoring (at the busbar) and remote monitoring points (e.g., end nodes).

1.5 Main Contributions of the Thesis

This section summarises the main and original contributions of this Thesis.

1.5.1 Management of Residential-scale PV Systems

Decentralised Voltage Control (Volt-Watt). It is demonstrated that a universal, single set of Volt-Watt settings is possible to be identified with the proposed method as the most adequate to effectively manage voltage issues, regardless the penetration, in a given LV network. This brings benefits to the DNO as it increases the scalability of the corresponding control method due to the fact that these settings are universal for all PV inverters, require a one-off setup (e.g., on the installation day) and are valid for any PV penetration.

<u>Decentralised Voltage Control (Volt-Var)</u>. The limitations and drawbacks of adopting such a control method, with commercially available residential-scale PV inverters, are demonstrated for the first time. Findings show that the benefits of this control approach are minimal due to the limited reactive capability of PV inverters at peak generation periods, where voltages are higher. This allows DNOs to take more informed decisions when adopting decentralised voltage control with reactive compensation (i.e., Volt-Var).

<u>Centralised Thermal Controller</u>. An innovative and scalable centralised control logic which bridges the gaps in the literature (e.g., adoption of complex optimisation techniques, requirement of extensive network monitoring, not scalable, etc.) is developed and proposed in this Thesis. The proposed control logic aims to manage thermal overloads in PV-rich LV networks (for any penetration) while using limited network observability (i.e., substation measurements only). The proposed control logic, with its performance successfully assessed considering different PV penetration levels (i.e., 0-100%), offers significant benefits to DNOs as it does not require the installation of remote monitoring points and reduces the need of costly communication infrastructure. More importantly, its generic nature allows it to be deployed and work on any PV-rich LV network its performance is expected to be the same as the proposed method and will always curtail the amount of energy required to keep the utilisation of assets within limits. It should also be highlighted that although the proposed control method is designed considering residential-scale PV systems as the controlled technology, it can easily be adapted to manage any other technology that can support active power control.

<u>Combined Centralised Thermal and Decentralised Voltage Control</u>. To provide a complete active network management aiming to solve both thermal and voltage issues at any PV penetration level, this Thesis proposes the combination of the centralised thermal controller with the decentralised voltage control (i.e., Volt-Watt). Unlike other solutions that are not easily implementable (due to complex optimisation algorithms, increased network visibility, expensive communication infrastructure), the proposed control method offers a more cost-effective solution requiring limited network observability. As previously mentioned, the generic nature of the proposed method, allows it to be deployed and work on any PV-rich LV network.

1.5.2 Management of Residential-scale Battery Energy Storage Systems

For the first time, an advanced operation mode that provides benefits both to the customer (i.e., increased self-consumption) and the DNO (i.e., voltage and thermal management) is proposed for residential-scale BES systems. The proposed operation mode, which can be programmed on residential-scale BES systems, offers the advantages of the decentralised control approach thus requiring no communication infrastructure. Based on local measurements (i.e., inverter connection point) the advanced operation mode can progressively charge/discharge the BES system in a way that increases the customer's self-consumption while effectively eliminating all

voltage issues up to high PV penetration levels beyond which it can significantly reduce them. The decentralised and generic nature of the proposed operation mode makes it replicable on any PV-rich LV network; however, it should be noted that the overall performance might be influenced by the size of the installed BES systems. For instance, although smaller sizes of BES systems will still bring benefit to the customers (i.e., reduced energy bills) they might not bring enough benefits to the DNO (i.e., voltage and thermal management) due to their limited storage capability.

1.5.3 Management of OLTC-fitted LV Transformers

A scalable and adaptive OLTC logic that aims to manage contrasting issues of voltage rise (due to the presence of generation) and drop (due to loads) is proposed in this research. Compared to traditional OLTC control approaches (i.e., fixed voltage target) that do not consider the diversified voltage levels among different feeders (due to uneven penetration of LCTs, load unbalance, etc.), the proposed method allows the automatic update of the voltage target (at the busbar) according to network conditions while using limited network observability coming from critical remote monitoring points (i.e., feeders' end points). Crucially, this provides the significant benefit of easily adapting to network changes (i.e., additional installation of PV systems or loads) without the need of reconfiguring OLTC settings. Considering the case study, the proposed OLTC control logic, which was assessed considering different PV penetrations, was found to effectively eliminate all voltage issues up to high PV penetration levels beyond which it can significantly reduce them.

1.5.4 List of Publications

The development and application of the proposed active management solutions in different analyses and studies have led to the publication of the following journal paper, conference papers and technical reports.

1.5.4.1 Journal Papers

 A. T. Procopiou and L. F. Ochoa, "Voltage Control in PV-Rich LV Networks without Remote Monitoring," *IEEE Transactions on Power Systems*, vol. 32, no. 2, pp. 1224-1236, March 2017. [51] DOI Link: https://doi.org/10.1109/TPWRS.2016.2591063 A. T. Procopiou, J. Quiros-Tortos, and L. Ochoa, "HPC-Based Probabilistic Analysis of LV Networks with EVs: Impacts and Control," *IEEE Transactions on Smart Grid*, vol. 8, no. 3, pp. 1479-1487, May 2017. [52] DOI Link: https://doi.org/10.1109/TSG.2016.2604245

1.5.4.2 Conference Papers

- A. T. Procopiou, C. Long, and L. F. Ochoa, "On the effects of monitoring and control settings on voltage control in PV-rich LV networks," in *Power & Energy Society General Meeting*, 2015 IEEE, 2015, pp. 1-5. [53] DOI Link: http://dx.doi.org/10.1109/PESGM.2015.7285791
- L. Chao, A. T. Procopiou, L. F. Ochoa, G. Bryson, and D. Randles, "Performance of OLTC-based control strategies for LV networks with photovoltaics," in *Power & Energy Society General Meeting*, 2015 IEEE, 2015, pp. 1-5. [54] DOI Link: http://dx.doi.org/10.1109/PESGM.2015.7285618
- A. T. Procopiou, L. Chao, and L. F. Ochoa, "Voltage control in LV networks: An initial investigation," in *Innovative Smart Grid Technologies Conference Europe (ISGT-Europe)*, 2014 IEEE PES, 2014, pp. 1-6. [55] DOI Link: http://dx.doi.org/10.1109/ISGTEurope.2014.7028971

1.5.4.3 Technical Reports

All technical reports were produced for the industrial iCASE project "Active Management of LV Networks" and submitted to the Electricite de France (EDF) R&D, who will use to make more informed decisions in terms of the solutions that could be implemented to increase the penetration of LCTs in LV network. The following reports are under a non-disclosure agreement (NDA) between the authors (Andreas T. Procopiou, Luis F. Ochoa) and the company (EDF R&D).

- 1. **A. T. Procopiou** and L. F. Ochoa, Deliverable 2.4 "Benefits of adopting Storage Devices in LV networks", Active Management of LV Networks, prepared for EDF R&D, October 2016 [56]
- 2. **A. T. Procopiou** and L. F. Ochoa, Deliverable 2.3 "Benefits of controlling EVs and PV in combination with other technologies", Active Management of LV Networks, prepared for EDF R&D, June 2016 [57]
- 3. **A. T. Procopiou** and L. F. Ochoa, Deliverable 2.2 "Benefits of controlling EVs and PV", Active Management of LV Networks, prepared for EDF R&D, October 2016 [58]

- 4. **A. T. Procopiou** and L. F. Ochoa, Deliverable 2.1 "EVs and PV: Literature review and initial modelling", Active Management of LV Networks, prepared for EDF R&D, April 2015 [59]
- 5. **A. T. Procopiou** and L. F. Ochoa, Deliverable 1.3 "Monitoring Aspects and Benefits of OLTC in LV Networks", Active Management of LV Networks, prepared for EDF R&D, April 2015 [59]
- 6. **A. T. Procopiou** and L. F. Ochoa, Deliverable 1.2 "Benefits of OLTC transformers in LV", Active Management of LV Networks, prepared for EDF R&D, November 2014 [60]
- A. T. Procopiou and L. F. Ochoa, Deliverable 1.1 "Literature review and initial modelling", Active Management of LV Networks, prepared for EDF R&D, May 2014 [61]

1.6 Thesis Outline

This Thesis consists of eight chapters in total. The remaining seven chapters that follow are outlined below.

Chapter 2 - Managing Impacts and State of the Art

Chapter 2 presents the research that has been carried out in the literature considering active network management schemes aiming to solve voltage and thermal issues in LV networks due to high penetrations of residential-scale PV systems. This includes decentralised and centralised control of residential-scale PV systems, BES systems and the use of LV OLTC-fitted transformers. The gaps of the current research in the literature, which are addressed by this Thesis, are outlined and discussed.

Chapter 3 - Control Schemes for PV-Rich LV Networks

This chapter provides the details of all the control schemes proposed in this Thesis aiming at managing voltage and/or thermal issues in PV-rich LV networks. First, section 3.2, details the control schemes proposed to actively manage residential-scale PV systems aiming at managing both voltage and thermal issues. An "advanced operation" mode for residential-scale battery energy storage systems is proposed and presented in section 3.3 to bring benefits to both customers (i.e., increased self-consumption) and DNO (i.e., voltage and thermal management). Lastly, the proposed control logic for LV OLTC-fitted transformers aiming at managing contrasting voltages issues (rise and drop) using limited network observability coming from

critical remote monitoring points (i.e., farthest customer connections or feeders' end points) is proposed and presented in section 3.4.

<u>Chapter 4 – Case Study Part 1: Modelling</u>

Taking advantage of having access to real French data (due to this research being part of an industrial project), this chapter, presents the modelling of different elements essential to perform realistic time-series analyses and assess the performance of the proposed active network management schemes. First, the details (i.e., number of feeders, connected customers and transformer) and topology of the real residential French LV network used in this research are presented. Then, the modelling aspects of the French domestic load profiles (based on real smart meter data) and PV generation profiles are presented along with load and generation profile allocation procedures adopted in this work. Finally, the Monte Carlo methodology and performance metrics, used in this Thesis to assess the performance of the proposed control schemes, are presented.

<u>Chapter 5 – Case Study Part 2: Simulation Analyses for Residential-scale PV</u> <u>Systems</u>

This chapter presents the simulation analyses carried on the real French LV network to assess the performance of the proposed control schemes for residential-scale PV systems under different PV penetration levels during summer (i.e., worst case scenario - low demand with high generation). First, a stochastic analysis is performed to assess the performance of the most common decentralised voltage control approaches (i.e., Volt-Var and Volt-Watt). Thereafter, the performance of the proposed centralised thermal control is investigated, along with the performance of the proposed combined centralised thermal and decentralised voltage control. Finally, a summary of the most important findings is presented.

<u>Chapter 6 – Case Study Part 3: Simulation Analyses for Residential-scale BES</u> <u>Systems</u>

The proposed "advanced operation" mode for residential-scale BES systems is adopted in this chapter on the real French LV network to assess its performance under different PV penetration levels during summer. Its performance is compared against the "normal operation" mode adopted based on available specifications from BES system manufacturer. A time-series analysis is first performed on the two BES operation modes. Then, their performances are assessed adopting a stochastic analysis considering different PV penetration levels. Finally, a summary of the most important findings is presented.

<u>Chapter 7 – Case Study Part 4: Simulation Analyses for LV OLTC-fitted</u> <u>Transformers</u>

This chapter investigates and presents a quantitative assessment of the benefits when adopting OLTC-fitted transformers with the proposed control logic in order to manage voltage issues in PV-rich LV networks. First the OLTC-fitted transformer adopted for the corresponding analyses is presented. Then, the time-series performance of the proposed OLTC control logic is assessed using the real French LV network. Lastly, its performance is stochastically investigated considering different penetration levels of residential-scale PV systems for different days in summer, levels of monitoring and control cycles. Moreover, this chapter summarises and compares the technical benefits (i.e., voltage and thermal management) of the proposed control schemes investigated in this Thesis, along with different implementation aspects (i.e., potential required investments, effects on customers).

Chapter 8 – Conclusions and Future Work

This chapter presents a summary of the main findings and conclusions gained from this research as well as potential improvements and future work that can follow the work presented in this Thesis.

Chapter 2

Managing Impacts and State of the Art

2.1 Introduction

This chapter presents the research that has been carried out in the literature considering active network management schemes aiming to solve voltage and thermal issues in LV networks due to high penetration of residential-scale PV systems. The gaps of the available research in the literature, which are addressed by this Thesis, are outlined and discussed.

2.2 Impacts of PV Systems on Passive LV Networks

Given the rapid adoption of residential-scale PV systems in passive LV networks (designed with relatively small considering relatively small loads, and without any provision for accommodating any kind of LCT), it is essential to investigate and understand the potential technical impacts that these might cause to the networks. A complete description of the possible technical impacts occurring from PV systems integration in distribution networks is presented in [62] highlighting the following: voltage rise, voltage fluctuations, voltage and current unbalance, reverse power flow, asset overloading, increased power losses and increased total harmonic distortion.

In order to improve the understanding of the technical impacts associated with the integration of residential-scale PV systems in LV networks, several authors have carried out diverse studies [23, 63-65]. One of the most interesting studies performed in [63, 64], assesses the impacts of different PV penetrations on two real UK LV networks in order to estimate their corresponding hosting capabilities applying a Monte Carlo technique. The analysis, which considers 5-min resolution synthetic data for domestic load and PV generation, highlighted the impacts of voltage rise, reverse power flow, and the overloading of feeder segments which are results of high PV penetration levels. Considering an even more thorough impact study, which was performed on more than 10,000 real LV networks in New Zealand and using real domestic load and generation profiles, the authors in [23] highlight among different
technical impacts (e.g., reverse power flow, voltage unbalance), that voltage rise is one of the most dominant in LV networks.

According to the aforementioned studies it is obvious that high penetrations of residential-scale PV systems in passive LV networks, and particularly those designed considering relatively small loads, will lead to technical issues such as voltage rise and thermal overloads of assets (due to reverse power flow). To illustrate how these issues occur and how they might affect passive LV networks, Figure 2-1 presents a simplified LV network.

For simplicity, the LV network in Figure 2-1 supplies only one household with a demand of P_L and Q_L through a feeder with an impedance of R + jX. At the same connection point, a PV system is also connected with an active and reactive power equal to P_{PV} and Q_{PV} , respectively.

First, assuming that no PV system is present, the voltage at the household connection point, bus 1 (V_1), can be approximated (in per unit) using (2.1). This is illustrated in Figure 2-1 (grey line), where V is the voltage at the substation busbar.



$$V_1 \approx V - (R \cdot P_L) - j(X \cdot Q_L) \tag{2.1}$$

Figure 2-1 Voltage rise and drop on a passive network

As shown in the figure, under load-only conditions ($P_{PV} = 0$), the voltage at the customer's connection point is dropping as it is directly influenced by its demand (i.e., P_L, Q_L) and the impedance (i.e., R + jX) between its connection point and the substation. It should be noted that the reactance to resistance ratio (X/R) in distribution networks, and particularly in LV, is considerably lower (≤ 0.5) than in higher voltage level networks (≥ 10) and therefore the values of cable resistances are high, leading to significant voltage drops or rises along the line [66].

However, once the PV system is connected, the voltage at the household connection point (V_1) , which can now be approximately calculated using (2.2) as in [67], is likely to increase due to the reverse power flowing back to the substation.

$$V_1 \approx V + R(P_{PV} - P_L) + X(Q_{PV} - Q_L)$$
(2.2)

As indicated in (2.2) and observed in Figure 2-1, voltage will increase depending on the level of PV generation. For example, assuming a unity power factor for both demand and generation (i.e., $Q_L = Q_{PV} = 0$), if the PV generation is equal to the customer's demand (i.e., $P_{PV} = P_L$), then voltage is neither expected to drop nor rise as no power is demanded or generated at the customer's connection point (i.e., no power is flowing). However, as soon as the PV generation exceeds the local demand (i.e., $P_{PV} > P_L$), power starts to flow towards the substation which results in the voltage rise at the customer's connection point (see Figure 2-1). With significantly higher PV generation (i.e., $P_{PV} \gg P_L$), this issue becomes even more severe as voltage can rise above the statutory limits (red line, Figure 2-1) due to the excessive power flowing back to the substation.

Furthermore, as previously discussed in section 1.2.1, this excessive amount of reverse power flow (due to high PV generation) might result in the thermal overloading of the most important and expensive assets of the network (i.e., feeder cables and transformer). Overloading the corresponding assets above their maximum current capacity leads to the increment of their insulation temperature above their operational limit. Consequently, this phenomenon may result in damaging or accelerating the ageing of the assets.

Considering this passive nature (i.e., no controllable elements) of traditional LV networks, DNOs are now facing significant challenges as the connection of large

numbers of PV systems might lead to technical impacts (i.e., voltage rise and asset thermal overloading), that unfortunately limit the hosting capacity of LV networks [68]. Although the traditional solution of reinforcing (or replacing) the existing network assets (i.e., larger transformers and feeder cables) allows accommodating higher number of PV systems, it requires significant capital investments [29, 69]. Therefore, this passive "fit & forget" connection approach represents a barrier to the adoption of high penetrations of low carbon technologies and, in particular, residential-scale PV systems.

An alternative to this costly network reinforcement, is the active management of the PV systems themselves and the use of other technologies (i.e., energy storage, OLTC) to solve (or postpone) these technical impacts [29]. The following sections present the most prominent research carried out in the literature aiming at managing voltage and thermal issues by controlling residential-scale PV systems, BES systems and LV OLTC-fitted transformers.

2.3 Residential-scale PV systems

Modern inverters for residential-scale PV systems are considered to be intelligent as they have far more capabilities than just converting DC to AC power. Today, commercially available inverters provide power control functions which are able to limit the active power output (i.e., curtailment) of the corresponding PV system that they are connected to and also inject or absorb reactive power [30, 70, 71]. What is more important, they are equipped with numerous two-way communication interfaces enabling them to be remotely controlled through wired (e.g., power line carrier, fibre optic, dedicated lines) or wireless (e.g., radio/cellular signals, WiFi, WiMAX, ZigBee) connections. Although standards in Europe, as stated in [72], have allowed reactive power compensation from small scale distributed generators (i.e., residential-scale PV systems) to manage voltage, PV inverters are still operated only to convert DC to AC power; thus their real potential is yet untapped.

2.3.1 Embedded Active and Reactive Power Control Functions

To understand how PV inverters can help in actively managing LV networks, this section presents the most common power control functions which are found in modern residential-scale PV systems in Europe (e.g., France, UK) [73]. It should be

noted that although a power factor control function, $\cos \varphi(P)$, is also embedded in modern residential-scale PV systems, only functions with a direct relation to voltage are considered in this Thesis.

2.3.1.1 Active Power Limit

This function limits the inverter's maximum output power by setting a limit as a percentage value (e.g., 50%) to which the maximum rated output power will be limited to. For example, if a PV system with a rated power of 3kWp has an active power limit of 50%, then it will be able to output only up to 1.5kW of power as illustrated in Figure 2-2. This essentially means that any resulting power above 1.5kW is curtailed.



Figure 2-2 Active Power Limit function

Such function can be used in either centralised or decentralised control approaches to limit the maximum output of large number of PV systems aiming to prevent the overloading of assets (i.e., transformer and feeder cables). This, in turn, results, as a natural effect of reducing the active power, in the reduction of high voltages during peak generation.

2.3.1.2 Volt-Watt

Another control function that manages the active power output of the PV systems is the Volt-Watt which tries to maintain the voltage at the terminal of the PV system within predefined voltage limits (e.g., statutory voltage limits). It essentially limits the maximum generation capability of each individual PV system according to 1) the voltage at the point of connection (at the terminals of the PV system), and 2) the predefined Volt-Watt set-points (set either by the DNO or the owner itself). This is illustrated in Figure 2-3. Although the Volt-Watt set-points can be constructed considering multiple points to represent a detailed and accurate curve, modern residential PV inverters usually support up to 6 points to represent a linear curve (more details in section 3.2.1). While this allows PV inverters having a smaller processing unit (i.e., cheaper and smaller), it also increases scalability and practicality as less effort is required to design and setup the corresponding curves.



Figure 2-3 Generic Volt-Watt curve

Considering situations where high PV output and low demand might result in high feeder voltages, the Volt-Watt control function can be used to limit the maximum active power output of the PV systems and, therefore, bring voltages within the statutory limits. This control can also be beneficial in situations where existing controls (e.g., voltage control through OLTC-fitted transformer) are not able to prevent the occurrence of these high voltages. Moreover, this control function can be used to also manage thermal issues (provided that the overvoltages and the thermal overloads occur at the same time) by adopting conservative Volt-Watt set-points while managing voltages.

2.3.1.3 Volt-Var

The Volt-Var control function, similar to the Volt-Watt, tries to maintain the voltage at the terminal of a PV system within predefined voltage limits (e.g., statutory voltage limits). It essentially allows each individual PV system to provide a unique var response according to 1) the voltage at the point of connection (the terminals of the PV system), 2) the available reactive power capacity of the inverter at that point in time, and 3) the predefined Volt-Var set-points (set either by the DNO or the owner itself) as illustrated in Figure 2-4.

Such control function can be applied in a control approach where reactive power is absorbed if the local voltage begins to exceed the pre-determined upper level (as defined by the Volt-Var set-points). On the contrary, if the voltage begins to fall below the pre-determined lower level (e.g., due to the reduction in active power output) reactive power can be injected in the network to help boost the voltage back to normal levels.



Figure 2-4 Generic Volt-Var Curve

2.3.1.4 Reactive power capability of normal residential-scale inverters

It is important, however, to highlight that the Volt-Var control function depends on the reactive capability of the corresponding PV inverter. At each instant *t*, the maximum available reactive power capability, Q_t^{max} , can be calculated using (2.3).

$$Q_t^{max} = \sqrt{(S^{max})^2 - (\hat{P}_t)^2}$$
(2.3)

At every instant *t*, the Q_t^{max} is calculated with respect to the real power generation, \hat{P}_t , and the apparent power rating of the inverter S^{max} . This active and reactive capability, is also illustrated in Figure 2-5 (a) which shows the P-Q capability curve of a normal/common (i.e., here is illustrated considering $S^{max} = 1 p. u.$) residential-scale PV inverter with a power factor capability of 0.9 (leading or lagging). As the figure shows, the reactive power capability of the inverter is directly influenced by the generation of the PV system (\hat{P}_t). More specifically, when considering the behaviour of the PV system through the day (influenced by the sun irradiance), Figure 2-5 (b) highlights that during periods of high generation (full irradiance, \hat{P}_t =

1 p. u.), the inverter is not able to absorb or inject any reactive power posing an important limitation of the corresponding control functions during the periods where reactive compensation is most needed (to reduce voltages). It is worth mentioning that the aforementioned limitation can be overcome by oversizing the inverters and therefore being able to absorb/inject reactive power even at peak generation periods. This, however, entails higher costs to the end customers which might not be willing to cover.



Figure 2-5 Power capability of normal residential-scale PV inverter

2.3.2 State of the Art

Considering the variety of already available control functions and the high flexibility of PV inverters to manage active and reactive power makes them desirable in mitigating technical issues such as voltage rise and thermal overloads. The majority of studies in [31-38, 74-81] focus on managing voltage issues adopting decentralised control methods whereas the number of studies investigating the adoption of centralised control methods is comparably lower [82-86]. These prominent published research studies are discussed in the following sections.

2.3.2.1 Decentralised Control

Given the lack of observability in LV networks and the nature of decentralised control approaches where actions are taken locally without the need of communication infrastructure, researchers have primarily focused on the decentralised management of voltage issues. A couple of studies, and more specifically in [79-81], investigate the adoption of decentralised active power curtailment methods (i.e., Volt-Watt, active power limit) to manage voltage issues where a stronger emphasis is given in the literature on the use of reactive power

compensation through residential-scale PV systems to reduce (or eliminate) the corresponding voltage issues [31-38, 74-78].

2.3.2.1.1 Active Power Control

The adoption of the decentralised Volt-Watt control function (described in section 2.3.1.2) of residential-scale PV systems considering five different curves (from least to most sensitive in voltage changes), is investigated in [79] on three test LV feeders (based on real feeders in Belgium) with 62, 48 and 38 customers, respectively. Based on only three PV penetration level scenarios, the analysis showed that the investigated control method, considering all Volt-Watt settings, is able to keep all customer voltages within the statutory limits while also improving voltage unbalance issues. Nonetheless, while authors highlight the fact that the amount of energy curtailed is influenced by the adopted Volt-Watt settings, no detailed information is provided in terms of the amount of energy curtailed based on the different settings. The provision of such information is important as it allows DNOs taking more informative decisions when adopting the corresponding control scheme.

A more sophisticated decentralised control algorithm for PV systems that aims to manage local voltage issues through the active power curtailment of PV generation is presented in [80]. Using local measurements (i.e., voltage, power, weather, radiation) a short-term (i.e., 15 seconds) forecast (based on a Kalman filtering technique) is performed to check for potential overvoltage issues in the horizon. Using this forecast information, an estimation is also performed to evaluate if the overvoltage issue can be solved with reactive compensation. If not, then the amount of active power that is required to be curtailed is calculated for the corresponding PV system. The proposed method, which is validated using a modified IEEE 34-node test feeder with 2 PV systems, shows that it can effectively manage voltage rise issues. Despite the effectiveness of the proposed method, the validation considered a deterministic case with only two PV systems. Considering the extent of LV networks (e.g., circa 500,000 in the UK) and the number of PV systems installations, a significant effort might be required to model/program the corresponding forecast algorithm on each PV system making the adoption of such control method challenging to implement.

An interesting study in [81] combines the use of decentralised active (i.e., curtailment) and reactive power control of residential-scale PV systems to manage

voltage issues in PV-rich LV networks considering five operation modes. During normal operating conditions ("mode 1") where voltage is below a specified voltage (i.e., 1.07 p.u.) PV systems follow a specific Volt-Var curve to adjust their reactive power. As soon as "mode 1" of any PV system is not enough to maintain the voltage below 1.07 p.u. then "mode 2" is activated and a distress signal is sent to all PV systems that do not face voltage issues. Upon receiving the signal, all PV systems start absorbing reactive power based on a linear relation between the reactive power output and the remaining time until "mode 2" deactivation. If within the specified period of time, "mode 2" is not able to solve voltage issues, "mode 3" is activated where all PV systems start curtailing their generation based on a linear relation of the active power output and the remaining time until "mode 3" deactivation. Once the voltage issues are managed, "mode 4" is activated to smoothly restore the active power generation. "Mode 5" follows to restore reactive power in order to allow entering back in "mode 1". The proposed method which was validated considering two test cases (single feeder and 14-feeder LV network), shows that important implementation settings like the duration period of "modes" and the Volt-Var curve settings are case specific as different settings were used for each test case. Consequently, the adoption of this control scheme might face implementation challenges as it will have to be tuned (i.e., settings) for each of the thousands of LV networks in a given region. Moreover, while the proposed method is based on the fact that it can be implemented without the need of extended communication infrastructure, some sort of investment is required to enable PV systems to transmit and receive overvoltage and distress signals, respectively.

While the adoption of active power control (i.e., generation curtailment) in PV-rich LV networks provides an effective voltage management and therefore increases the ability of LV networks to adopt more LCTs, there are a limited number of studies on this area. As this issue might be associated with the fact that the active power control of PV systems leads to reduced yields on PV system owners [87], current literature and researchers have put a significant effort in investigating the use of reactive power control of residential-scale PV systems to manage voltage issue.

2.3.2.1.2 Reactive Power Control

German studies in [28, 29], where the techno-economic comparison of the four most commonly available PV inverter power control functions (presented in section 2.3.1) is performed, highlight that the decentralised reactive power control, and particularly Volt-Var, is one of the most practical and economic approaches to adopt (see Figure 2-6). In addition, recent results from real pilot projects in Germany and Austria [74-76], "DG DemoNet Smart LV Grid" and "Active intelligent low-voltage network", show that the adoption of decentralised Volt-Var control (combined with OLTC-fitted transformer) on German LV networks can provide effective voltage management. Nonetheless, considering that households in Germany and Austria are three–phase connected, it is unclear the extent of the benefits when adopting such control methods in LV networks where customers are mostly single-phase connected (e.g., UK, France, U.S.A). To provide more understanding on this matter, voltage rise caused by a single-phase connected PV system is considered to be approximately six times higher than the voltage rise caused by a three-phase connected PV system of the same power [38, 88, 89].



Figure 2-6 Investment cost of different voltage control strategies, on a German LV network [29]

An investigation of the benefits when adopting a decentralised Volt-Var control to manage voltage issues in single-phase household connections is presented in [77]. This work, having as the main objective to provide evidence on the ability of such control scheme to mitigate voltage rise, investigates the adoption of generic Volt-Var curves on representative U.S. feeders considering 15, 30 and 50% penetration level of residential-scale PV systems. Similarly, using another U.S. feeder with 11 PV

systems, the work in [37] also assesses the performance of Volt-Var control adopting specific inverter set-points (i.e., Volt-Var curves). Although the results (see Figure 2-7) from both studies in [37, 77] although prove that voltage rise issues can be indeed mitigated with the adoption of Volt-Var control, authors do not give details on how adequate settings can be defined.



Figure 2-7 Customer voltages with 50% penetration of PV systems without (left) and with Volt-Var control (right) [77]

In the same vein of decentralised voltage management, authors in [31] move a step forward proposing a method to define optimal Volt-Var curves (i.e., set-points) for every individual PV inverter (22 in total) in a real Irish LV network with 74 singlephase connected customers. The performance of the optimal curves (in terms of voltage regulation), which are defined using a three-phase optimal power flow formulation (TOPF), is compared with a centralised active control scheme where the reactive power settings of the PV inverters are optimally defined in real time via a communication link. Even though results show that the adoption of optimal Volt-Var curves achieves almost the same performance with the centralised active control scheme, the analysis is performed considering only one specific PV penetration level (22 residential-scale PV systems in this case). Hence, it is unclear if the corresponding optimal Volt-Var curves will still be adequate considering a different PV penetration level or location of PV systems.

Adopting a similar approach to [31], the authors in [32] use a multi-objective genetic algorithm to define the optimal settings for each PV inverter in a real LV network (71 households) located in Bornholm Island, Denmark. The study, which aims to increase the hosting capacity of the LV network by adopting and comparing two

decentralised reactive power control functions, $\cos \varphi(P)$ and Q(V) (described in section 2.3.1) shows that the hosting capacity of the LV network can be indeed increased (up to 40% for the specific network) when adopting these methods. Out of the two, it was highlighted that the Volt-Var control, Q(V), provides more effective voltage regulation than the power factor control, $\cos \varphi(P)$, possibly due to the fact that Volt-Var is acting based on the voltage level rather than the PV system's active power output. Although the application of optimisation approaches as in [31-35] (e.g., multi-objective genetic algorithm, TOPF) allows defining the most adequate settings for the adopted PV inverter control (thus better voltage management), it requires extended information of the LV network topology (not commonly available information to DNOs). This, unfortunately, limits their practicality and scalability given that the inverter settings have to be individually defined for the thousands of LV networks in a given region. Additionally, it should also be highlighted that the adoption of such methods might lead to significant computational times and increase the risk of not converging to a solution as the size of the optimisation problem significantly increases with network size and number of control variables.

More recently, a study performed in [38] tries to provide more understanding on different (on residential-scale PV systems) decentralised voltage control functions and concepts while investigating 28 different control settings (i.e., curves) for the Volt-Var controller. The study, which highlights the importance and effectiveness of the Volt-Var control function in managing voltage issues, shows that the most effective settings in managing voltage issues leads to a significant amount of reactive power absorption. Despite the effectiveness of the method, the effect of the increased current flow in the grid (due to reactive power) which might lead to the thermal overloading of network assets (i.e., cables, transformer) is not investigated. Additionally, as shown in Figure 2-8, the analysis is based on four simplified scenarios where the PV systems in the feeder are (1) uniformly distributed, (2) linearly increasing, (3) linearly decreasing and (4) all lumped at the end. More importantly, the analyses consider only two fictional continuous feeders with uniform impedance where all households and PV systems share the same profiles, respectively. Although the simplified case study allows proving the concept, in reality, however, feeders have numerous laterals with households (loads) and PV systems spread unevenly along the feeder. Subsequently, the investigated methods

might overestimate the amount of reactive and active power that needs to be absorbed and curtailed, respectively. More importantly, this analysis, in [38], and several others, in [36, 37], assume PV inverters with ratings that can be high enough (i.e., oversized inverters) to maintain the full solar output (kWp) and the requested vars. However, as described in section 2.3.1.4 and shown in Figure 2-5, normal residential-scale PV inverters might have limited capability to inject or absorb the required reactive power (i.e., vars) in periods of high power output (i.e., sunny weather conditions). The available reactive power headroom is varying throughout the day (i.e., variable solar irradiance) and, therefore, might not be available when needed [77, 90].



Figure 2-8 Investigated scenarios in [38]

Another recent study in [78], aims to provide a solution not only to voltage rise but also voltage drop and voltage fluctuation due to the cloud effects on PV generation. The authors propose and test a three mode reactive power controller for PV inverters on an Australian distribution system. The controller uses a reverse power flow versus reactive power droop characteristic for voltage rise mitigation during high PV generation. During normal operation (i.e., low or no PV periods), controller operates in "mode 1" which is a dynamic var control mode for voltage drop support. In periods of voltage rise (i.e., high PV generation), the controller operates in "mode 2" absorbing reactive power to mitigate overvoltage. "Mode 3" is initiated in moments of high voltage fluctuation (i.e., passing clouds) in order to mitigate fluctuations by ramp-rate control of inverter var output. While the analysis proves that the proposed method is effective in voltage management, the investigation was performed considering a specific penetration level of PV systems and no detailed information was provided on the curve settings used for the PV inverters. More importantly, the adoption of the proposed control function, that requires the PV inverters to use a reverse active power flow versus reactive power droop characteristic, might not be readily available in commercial PV inverters. Thus, the control logic will have to be programmed for each PV inverter, reducing its practicality and scalability.

Despite the evidence provided by the literature that the decentralised reactive compensation can indeed aid in mitigating voltages rise issues and, consequently, increase the hosting capacity of LV networks, none of the studies investigates the effect of increased current flowing in the network due to the absorption of reactive power. As a matter of fact, the majority of available studies either do not consider any thermal impacts or assume that the first impact resulting from high penetration of residential-scale PV systems is the voltage rise. Recent LV network impact studies [15, 23, 91], however, show that thermal overloading of assets (i.e., transformer and feeder segments) might occur simultaneously with voltage rise issues. In even more severe cases, thermal overloading of assets occurs before any voltage rise issue appears. Consequently, although the adoption of reactive compensation might help solving one issue (i.e., voltage rise), it is important to highlight that this may result in the increment of another (i.e., thermal overloading of the assets), as more current is flowing in the network due to the absorption of reactive power.

2.3.2.2 Centralised Control

Centralised control approaches, where control actions and set-points are provided through a central controller (usually located at the secondary substation), have always been an important part of the academic research and one of the most preferred control practices in managing electricity distribution networks. Considering that voltage rise due to the adoption of high penetration of residential-scale PV systems in LV networks is, as reported in [15, 22, 23], one of the most dominant technical issues, the literature has focused in providing centralised solutions with a primary aim to manage only voltage issues [82, 83]. Management of the thermal overloading of assets, however, which is also an important technical impact in PV-rich LV networks, has not been thoroughly addressed [84]. More importantly, centralised solutions that consider both technical impacts simultaneously are limited [85, 86].

Starting with [82], a centralised real-time active power capping method is proposed to manage voltage rise issues in distribution networks with high penetrations of PV systems. To achieve that, every PV system (within a distribution network) considers a voltage threshold where, if violated, a signal is sent to all PV systems instructing them to start recording their active power outputs and voltages. Based on consecutive measurements a linear relationship between the voltage and active power is created for each PV system to then use and forecast the maximum active power output (generation cap) that can be allowed without resulting in voltage rise issues. Using this information, the centralised controller calculates and sends back a new generation limit for all PV systems that aims to share the generation curtailment among all of them (fair curtailment). Authors who highlight the implementation challenges associated with the required communication infrastructure between PV systems and the central controller also propose a technique to adopt the same method on a decentralised approach. However, a communication between neighbouring PV systems is still required.

Having the same goal (voltage management), authors in [83] propose a centralised control method that it can optimally define the active and reactive power output of residential-scale PV systems in unbalanced PV-rich LV networks. Using a multiobjective optimal power flow, the proposed method aims at reducing the corresponding voltage issues while reducing the amount of active power curtailment by providing priority to reactive power absorption. To reduce complexity, the use of the weighted-sum method is adopted to convert the multi-objective problem into an aggregated single-objective. Its performance is assessed on an Australian test LV network (based on Perth Solar City) with 77 single-phase connection customers and 34 residential-scale PV systems (i.e., 32% of PV penetration level). The results show that, indeed, voltages can be effectively managed with minimum curtailment while also reducing the voltage unbalance effect. However, to achieve the corresponding performance, the study considered 60% over-sized PV inverters to enable the absorption of reactive power even at peak generation otherwise not possible with normal inverters (as discussed in section 2.3.1.4). While over-sized PV inverters provide significant benefits to DNOs (e.g., voltage management) they are associated with higher costs (that customers might not be willing to pay) compared to normal and commercially available residential-scale PV inverters. Furthermore, the adoption of this control scheme requires a two-way communication which entails a major investment in communication infrastructure.

While all previously discussed studies aim at managing voltage issues, authors in [84] propose an optimisation method for thermal management in LV networks with high penetrations of LCTs (i.e., PV systems, electric heat pumps). In this study, residential customers who own a PV system are assumed to have a contract with the local DNO where they specify a percentage of their total generation that is allowed to be curtailed (i.e., non-firm capacity) while the rest should be maintained at all times (i.e., firm capacity). Based on this firm and non-firm generation capacity, a real-time optimisation method (using mixed integer programming) is proposed to minimise the amount of active power required to be curtailment in order to keep the utilisation of assets (i.e., transformer, feeder cables) within limits. The performance of the proposed method, which is investigated on a 16-bus Dutch LV network with full penetration of PV systems, is compared with other 4 different curtailment methods and rules in which PV systems should contribute to the curtailment (e.g., selection based on power flowing in the feeders, security constrained OPF). As expected, the proposed method achieves the goal (i.e., thermal management) with minimum curtailment compared to the investigated methods.

To provide a more complete solution, the same authors, propose in [85], the combination of their proposed centralised thermal controller with a decentralised active power control method to manage both thermal and voltage issues in PV-rich LV networks, simultaneously. The decentralised PV control approach which is based on a P-V droop characteristic is acting in real-time to manage voltage issues where the centralised controller, based on 15-min control cycles, is checking for thermal violations. Once identified, the optimisation problem is solved with the current network state to provide the optimal active and reactive settings for the PV systems. The analysis, which is based on the same case study in [84], considerers a multiagent system-based control scheme to perform coordination and communication between the different network components (i.e., transformer, feeders, customers and PV systems). This unfortunately, reduces the scalability and practicality of the corresponding method as it requires complex communication infrastructure and full network observability (currently not available in LV networks).

Using a similar approach (optimisation algorithm), the work in [86] also proposes an on-line centralised controller that aims in managing both voltage and thermal issues

simultaneously. To perform the optimisation method, a state estimation is performed to obtain the voltages at each bus. Then using sensitivity coefficients, the network's active and reactive power injections are estimated to become an input to the optimisation problem and finally define the active and reactive power for each PV system in the network. Although the application of distribution state estimation technique is likely to have a good accuracy, it requires extended information of the LV network topology as well as the number and type of connected low carbon technologies. Consequently, this might pose a significant challenge as the corresponding estimator has to be tailored to each of the thousands of LV networks in a given region.

More importantly, almost all of the centralised control approaches available in the literature apply complex optimisation algorithms to calculate the optimal active and reactive power settings of each PV system [83-86]. As a result, extensive information/visibility of the network is required which in reality is not available. Hence, a significant investment is required to develop the corresponding monitoring and communication infrastructure that will enable the successful implementation of the corresponding control schemes.

2.3.3 Summary of Gaps in Literature

The wide amount of recent studies suggests that the control of residential-scale PV systems to manage voltage issues in PV-rich LV networks is a topic undergoing intense study. While the work on centralised control methods in LV networks is limited, the adoption of decentralised control methods has attracted more attention, giving emphasis on the reactive power control of PV inverters. Based on the studies discussed in section 2.3, the most important gaps found in the current literature and addressed in this Thesis are highlighted and summarised below:

 An in depth analysis on the effects of different set-points (i.e., curves) for both Volt-Watt and Volt-Var control functions is not covered in the literature. While a detailed analysis of the effects will enable DNOs to take more informed decisions when adopting these controls, it will also allow identifying universal settings suitable for all PV penetration levels.

- 2. None of the papers in the literature considers the effects from the increased current flow when absorbing reactive power in LV networks. Although the absorption of reactive power might help to solve one issue (i.e., voltage rise), it might result to the increment of another (i.e., thermal overloading of assets).
- 3. The majority of papers assume PV inverters with ratings that can be high enough (i.e., oversized inverters) to maintain the full solar output (kWp) and the requested vars. In reality however, commercially available residential-scale PV inverters might have limited capability to inject or absorb the required reactive power (i.e., vars) in periods of high power output (i.e., sunny weather condition). Hence, realistic studies that consider this important limitation are required to understand the extent to which this control function can help to manage voltage issues.
- 4. The majority of centralised control methods proposed in the literature use complex optimisation techniques that might not only require significant investments in network monitoring (not available in LV networks) and communication infrastructure but also lead to significant computational times and increase the risk of not converging to a solution as the optimisation problem significantly increases with network size and number of control variables. Consequently, practical, cost-effective and scalable solutions that use limited and already available information need to be explored.
- 5. Almost all studies are based on managing only voltage issues and, in general, aim to solve issues separately. An integrated, practical and scalable solution that manages both voltage and thermal issues simultaneously is required.

2.4 Residential-scale Battery Energy Storage Systems

Residential-scale BES systems (up to ~20 kWh) have been recently introduced by several manufacturers (e.g., Tesla, LG, ABB, SMA) [92-95] to help householders reducing their electricity bills by storing the excess of PV generation (i.e., generation minus demand) during the day and using it later at night [39, 40]. The aforementioned behaviour is illustrated in Figure 2-9 where the orange shaded area represents the potential energy (i.e., excess of PV generation) that could be stored and then used during the night (purple shaded area). This behaviour can essentially decrease the customer's dependence from the grid as less energy is required to be supplied by the latter, and therefore reduce the corresponding electricity bills.



Figure 2-9 Interaction between residential-scale PV and BES system

As a matter of fact, Moixa Energy, a UK-based company which aims to install 1 million residential-scale BES systems across all UK by 2020, is claiming that each household can actually save up to £350 per year in electricity bills using the technology [96, 97]. The adoption of residential-scale BES systems has already gained significant interest during the last few years given the reduction of feed-in tariff rates for residential-scale PV systems, particularly in countries with rapid PV installation rates (i.e., UK, Germany, Italy) [98, 99]. To give an example, UK customers today receive £0.049 per kWh they export (generated from PV systems) into the grid, and pay £0.14 per kWh imported [100]. This significant difference in price is the main factor that incentivises customers to adopt residential-scale BES systems) and use it at a later time rather than exporting it into the grid. Indeed, a recent report on the energy storage market in Germany highlights that the declining

Active Management of PV-Rich Low Voltage Networks

feed-in-tariffs, along with the high electricity prices (47% increase since 2006), are the major drivers for the energy storage market growth; which is forecasted to exceed the annual value of 1 billion USD by 2021 [101].

In addition, the market growth is also expected to increase even more as several DNOs are now imposing fixed, network-wide limits on PV exports (i.e., generation minus demand) in an effort to reduce technical impacts due to high PV penetrations. In these cases, residential-scale BES systems will allow to store and use the corresponding energy above the imposed limit which would otherwise be curtailed (due to imposed export limits). To give some examples, in Germany, where more than one million PV systems are connected in LV networks, customers with installed capacities smaller than 30 kWp are not allowed to export more than 70% of their installed capacity [98]. Similarly, DNOs in Australia and Hawaii impose export limits to new PV system installations [102, 103].

If the future energy storage market predictions are correct, then large amount of technical innovation on this technology is required to exploit all of its possible applications. This is also evident through recent European [104] and government level [105, 106] innovation funds, aiming to provide innovation on this technology, and also through already launched energy storage projects [107-110]. Moreover, DNOs see this new technology as a potential tool that can help mitigate impacts associated with the excess of PV generation (e.g., voltage and thermal issues), thus providing an alternative solution to reinforcement.

One of the first pilot projects in California, US, which installed 5kW/7.7kWh BES systems in fifteen households with PV systems, investigated how the technology can help reducing the peak demand through load shifting [107, 108]. Although the project faced technical problems with some equipment, forcing a number of BES systems to go offline, it was shown that storage can indeed smooth the peak load while also reducing the electricity bills of the corresponding customers. Another project in Newington, Australia, which was set to investigate the use of residential-scale BES system (i.e., 4kW/10 kWh) in a household with PV system and electric vehicle, showed that even though it is able to reduce the household's net imported energy from the grid, it is unable to fully charge the electric vehicle [109]. In the UK, the low carbon network fund (LCNF) project Sola Bristol ran by Western Power

Distribution (WPD) aimed to investigate the adoption of residential-scale BES systems (i.e., 2kW/4.8kWh) in households with PV systems to reduce peak electricity consumption and manage thermal overloads [110]. Using variable tariffs to incentivise customers using the battery, the project showed that indeed the electricity consumption can be reduced. However, it was highlighted that further work is essential to exploit additional benefits of this technology.

2.4.1 State of the Art

Given that the use of residential-scale BES systems in LV networks is relatively a new concept, literature in this area is currently, as expected, limited [41-46, 111]. The majority of available studies focus on control schemes that provide benefits only to the DNOs (i.e., voltage management, load peak shaving) and use complex optimisation techniques to achieve their goals [41-46]. On the other hand, studies that consider customer benefits (i.e., increased self-consumption, electricity bill reduction) are very limited and more importantly, the provision of control schemes that considers both the customers and DNO benefits is yet to be explored [111].

2.4.1.1 Network-oriented Research

Authors in [41] investigate the potential of installing a BES system to manage technical impacts on LV feeders with a primary goal to increase their ability to host more PV systems. The analysis is performed using a multi-objective optimisation function that aims to manage voltages and reduce the peak load power while considering the cost of the BES system required to be installed. The proposed method, which is assessed on a real feeder with 62 residential customers and 30% of PV penetration, first identifies the optimal location for the BES system to be installed to then solve the multi-objective optimisation problem. Based on different objective values, the authors create pareto-optimal isocost curves that determine the corresponding BES system cost based on the required performance (i.e., peak shaving, voltage regulation). As expected, it is highlighted that the adoption of the BES system is able to manage voltage issues, hence increase the hosting capacity; however a trade-off between the objectives (i.e., peak power reduction and voltage management) is required in order to define the most adequate size for the BES system.

To manage voltage issues in PV-rich LV networks, authors in [42] propose a control scheme that allows DNOs to control the BES systems of residential customers to provide reactive compensation during specific time periods in exchange for a subsidy covering part of the BES system cost. Based on an optimisation function and using a synthetic MV feeder with aggregated LV customers, authors demonstrate that the proposed approach provides adequate voltage management. While this method shows promising results, it requires BES systems to reduce their charging power rate during periods of voltage issues so that the inverter (of the BES system) will be able to absorb reactive power (limitation of commercial inverters discussed in 2.3.1.4). This, unfortunately, might limit the amount of energy that the corresponding battery can store during the day.

The work in [43] proposes three optimisation-based (i.e., centralised, decentralised, distributed) and a rule-based control algorithms for residential-scale BES systems in order to reduce the variability in the power profile in LV networks. The first algorithm, which is based on a centralised controller, considers full network observability and communication to estimate the load and generation. These estimations are then used to optimally solve a model predictive control (MPC) problem to define the charging/discharging behaviour of each BES system in the LV network. The second algorithm, which considers a decentralised controller installed in every BES system, adopts the same approach with the central controller but only using local measurements and no communication. The third algorithm considers a hierarchical distributed control approach where each BES system is communicating with a central hub (i.e., Market Maker) that provides real-time prices of selling and buying power to or from the grid according to a network-wide objective (i.e., reduce the variability demand – flatten the power profile). Lastly, the forth algorithm considers a rule-based controller that manages the charging/discharging of BES systems according to the household's net demand profile. The performance of the proposed controllers was assessed considering 2kWh residential-scale BES systems as well as Australian generation and demand data. As expected, results showed that the centralised controller achieves the best performance in terms of flattening the power profile; however this method becomes immediately infeasible as the size of the optimisation problem increases with network size. While the decentralised and distributed control approaches, that do not require excessive network observability and communication, achieve similar performance (with the centralised), they both rely on predictions of demand and generation to solve the proposed MPC problem. This, however, increases implementation challenges as the proposed methods require historical data and case-specific settings to solve the MPC problem. As for the rulebased controller, it was highlighted that it does achieve the corresponding objective (i.e., reduction of power profile variability). Although the rule-based controller increases the practicality of the proposed method, no quantification was provided in terms of voltage issues and more importantly all proposed methods ignore the customer's primary goal when adopting a residential-scale BES system; to increase self-consumption therefore reduce electricity bill.

To manage voltage issues in LV networks, authors in [46] adopt a slightly different approach. Their proposed method aims to calculate the minimum required BES system capacity able to manage voltages issues by defining a household export limit and forcing the BES system to charge based on the power that exceeds that limit. The proposed method, which is investigated on a Danish LV network considering three penetrations (i.e., 50, 75, 100%) of 3-phase connected residential-scale PV systems, proves to be effective in managing voltage issues. However, the assumption of 3phase connected PV systems does not highlight the issues associated with heavily unbalanced phases which are common in many other European countries (i.e., UK, France). Moreover, while the proposed method efficiently solves the corresponding voltage issues, the energy stored in the BES systems is not quantified and is unclear if this method maximises the usage (i.e., reaching 100% state of charge) of the BES systems for the customer's benefit. Consequently, this reduces its profitability as the stored energy might not be enough to cover the household demand during the night.

Another approach in [45], similar to the previous study, proposes a control method to manage voltages in LV networks by forcing the BES systems to start charging only once a predefined power threshold is reached. To define the corresponding power threshold an optimisation problem using linear programing is solved. The proposed method, which is assessed on a Belgian LV feeder considering 23 and 50% of PV penetration level, highlights that the power threshold is case specific as it depends on factors such as the penetration and location of PV systems and the network topology. This, unfortunately, limits the scalability of the corresponding method as a power

threshold is required to be defined for each LV network. More importantly, similar to [46], the proposed method does not ensure that the BES systems will store the maximum available energy during the day as they charge only to manage voltage issues. This, however, limits the primary goal of the residential-scale BES systems which is to reduce the owner's electricity bill by storing the maximum available excess of PV generation and using it later in the day.



Figure 2-10 Conventional storage strategy and proposed strategy in [45]

Despite the effectiveness of the studies described in [41-43, 45, 46], all of them use optimisation algorithms to adequately control residential-scale BES systems and, therefore, achieve their objective (i.e., voltage management, peak shaving). This, however, as mentioned in section 2.3.2, increases the complexity and poses a significant implementation challenge as it entails extensive investments to provide the required monitoring and communication infrastructure in LV networks.

More recently, and in the context of providing more practical BES control schemes, without the need of advanced communication infrastructures, authors in [44] propose a decentralised control method that makes use of the reactive capability of PV systems combined with a droop-based charging/discharging control of BES systems to manage voltage issues in LV networks. The proposed method assumes that PV systems will continuously provide reactive compensation (based on their reactive capabilities) and that the BES systems will start charging, based on voltage-power droop curve, only once a critical voltage is reached. Considering a case study of a residential feeder with 12 houses all equipped with 6kWp PV systems, the authors investigated the performance of the proposed method adopting two scenarios where: 1) a universal droop curve is used for every BES systems, and, 2) a different droop curve is used for each BES systems (to ensure uniform energy storage). The results show that both cases achieve efficient voltage management; however, the first case

requires customers located at the end of the feeder to install larger BES systems (i.e., up to 5kWh) as they face higher voltages (thus more frequent charging) compared to customers located closer to the substation. On the other hand, the second case shows that if a specific droop curve is used for every BES system, a smaller battery size (i.e., up to 2kWh) is required to be installed, given that everyone contributes to the to the voltage management (uniform charging). Nonetheless, to define the corresponding droop curves an off-line voltage sensitivity analysis considering the corresponding LV network is required which makes the approach case specific. Additionally and similar to the previously discussed studies, the charging of BES systems is only activated once a critical voltage level is reached. This, however, limits the ability of the BES systems to store any excess of PV generation during periods or days where there are no voltage issues (i.e., not very sunny days).

2.4.1.2 Customer-oriented Research

Research considering the control of residential-scale BES systems for the customer's benefit (i.e., reduce electricity bills) is limited. Nonetheless, the study in [111] provides a thorough analysis that evaluates the economic potential of using a residential-scale BES system, given the falling feed-in tariffs for residential-scale PV systems and considering a range of variables: the electricity price forecast, battery costs forecast, load demand response and BES system lifecycle. Using Australian demand and generation data, the optimal size of the BES system (i.e., 5kWh) is defined based on a 2.5kWp PV system which is assumed to be the most common size in Australian residential installations. Once the size is defined, an optimal charging/discharging operation is developed adopting state space dynamic programing. This optimal operation is essentially maximising the profitability of the BES system (i.e., minimisation of the total cost of electricity use) using future predictions of load and generation. Then adopting the corresponding optimal operation, the payback period for the technology is calculated considering different BES sizes and costs per kWh capacity. The analysis highlights that residential-scale BES systems can indeed become economically viable in the foreseeable future (i.e., around 2020), given the continuous rise in electricity prices and the decreasing BES system costs.

Despite the thorough analysis of the economic potential of using residential-scale BES systems in LV networks, the study considers only specific size of residentialscale PV systems and therefore the stochastic variability of PV sizes, which can be found in LV networks, is not accounted. In addition while the analysis considers the cost of the BES system, it does not take into account costs associated with the corresponding installation. Crucially, the proposed optimal BES system operation relies on optimisation and perfect prediction of generation and demand which might increase the implementation complexity as historical data and network information is required.

2.4.2 Summary of Gaps in Literature

In general, the literature shows a strong interest on trying to define the most adequate control algorithm (i.e., charging/discharging operation) for residential-scale BES systems depending on the desired objective. The majority of studies discussed in section 2.4.1, focus on controlling residential-scale BES storage systems to manage technical issues in PV-rich LV networks. In these studies, the primary goal of the residential-scale BES systems (reduce customer electricity bills) is not addressed thoroughly and in some cases is not even considered. This, however, creates an important gap between academic research and real practice as most of the customers might not be willing to use their BES systems only to manage technical issues. Based on the research discussed in section 2.4, the most important gaps found in the current literature and addressed in this Thesis are highlighted and summarised below:

- A deep realistic analysis of the normal (i.e., out of the box) operation of residential-scale BES systems considering different penetrations of residentialscale PV systems in LV networks is required. This will allow identifying potential limitations of this technology and help develop solutions catering for those.
- 2. Almost all of the studies found in the literature apply complex optimisation techniques to define the optimal operation of residential-scale BES systems. These, unfortunately, require significant investments in network monitoring and communication infrastructure. Consequently, practical, cost-effective and scalable solutions that use limited and already available information are needed.

3. Control schemes for residential-scale BES systems that take into account both the benefits of customers (i.e., increase self-consumption, thus reduced electricity bills) and DNO (i.e., voltage and thermal management) are yet to be explored. It is essential to develop control algorithms that aim to maximise the amount of energy stored in BES systems during the day while managing technical issues (e.g., voltage and thermal issues).

2.5 LV OLTC-fitted Transformers

Due to the passive nature of LV networks around the world (i.e., unidirectional power flow, thus no controllable elements), transformers in secondary distribution substations (e.g., 11kV/0.4kV or 6.6kV/0.4kV in the UK, 20kV/0.4kV in France) are designed with transformation ratios high enough to compensate voltage drops according to the worst case scenario (i.e., maximum load conditions without considering PV generation). More importantly these transformers are equipped with off-load tap changers [16-19] which means that the ratio between the primary and the secondary voltage can only be changed after disconnecting the load [112, 113]. Consequently, once a given off-load tap position has been found to be adequate for the corresponding LV transformer loading conditions, it is very unlikely that this setting is changed in the lifetime of the transformer (unless the customer composition of the network itself changes significantly).

Considering the above, the last point of voltage regulation in distribution networks is traditionally performed at the primary distribution substations (e.g., 33/11kV or 33/6.6kV in the UK) which are equipped with OLTC-fitted transformers [16-19]. The principle of voltage regulation on distribution networks, as illustrated in Figure 2-11, is to maintain the voltage at the secondary side (point A) close to a predefined voltage target (commonly above the nominal) so that the voltage of all connected customers in the medium (MV) and low voltage feeders (particularly those connected in the far end, point D) is within the statutory limits during maximum load (dark blue line) [21]. However, given the pre-defined settings (i.e., tap position) of off-load tap changers and the relatively high transformation ratio of transformers in LV networks, the connection of large numbers of residential-scale PV systems may result in voltages significantly higher than the upper statutory voltage limits (i.e., point D in Figure 2-11) during minimum load (red line).

Crucially, the degree to which voltages can be reduced or increased is constrained due to the voltage compliance of medium voltage (MV) customers and the thousands of customers connected in the LV networks [114]. Thus to increase the 'on-load' flexibility in LV networks the use of LV OLTC-fitted transformers can be considered as a potential solution to manage voltages closer to LV customers and therefore increase the corresponding PV hosting capacity. Indeed, German techno-economic studies in [29], highlight that the adoption of OLTC-fitted transformers in LV networks (compared to traditional network reinforcement) can be both a technical and economically viable solution to manage voltages.



Figure 2-11 Voltage variation in radial distribution feeder [21]

2.5.1 Transformer and OLTC Operating Principles

To understand how OLTC-fitted transformers can help in actively managing voltages closer to LV customers, the basic operation of transformers and OLTCs is discussed in the following sections.

2.5.1.1 Transformer Operating Principle

The operation of a transformer is based on the principles of electromagnetism to change an AC voltage level to another. Its basic operation principle, which is detailed in [115], is illustrated in Figure 2-12 which demonstrates a two winding ideal transformer. If voltage is applied (V_p) on a conductor (e.g. transformer primary winding), the resulted alternating current (I_p) flowing in the conductor will create a changing magnetic field, flux Φ , around the conductor which can induce a voltage (V_s) to another conductor ends (e.g. transformer secondary winding) connected also in the same magnetic field. The resulted voltage magnitude (V_s) on the secondary winding depends on the turns ratio (N_p/N_s) of the transformer. For example, if the secondary winding has half the number of turns of the primary winding, then the voltage on the secondary winding will be half the voltage across the primary

winding. Therefore, the secondary winding voltage, step up or down, can be approximated using (2.4).



Figure 2-12 Basic transformer operation (ideal transformer)

OLTC-fitted transformers, using a mechanical device, are able to change the number of turns to be used, in discrete steps. Hence, this ability can offer the advantage to change the transformation ratio according to the available tap positions. This allows voltage adjustments at the substation, under loading conditions, to increase or reduce the voltage level.

2.5.1.2 OLTC Operating Principle

OLTCs are typically installed on the primary side of transformers and they are controlled by the Automatic Voltage Control (AVC) relay which uses voltage measurements from the secondary side of the transformer to define the corresponding control actions (i.e., tap change). This operation is illustrated in Figure 2-13 where a voltage transformer (VT) measures the voltage on the secondary side of the transformer (i.e., V_s) [116]. This measured voltage is always compared with a predefined voltage target and no regulation is performed if it is within the defined bandwidth settings (defined based on acceptable voltage variations). Once the measured voltage deviates from the bandwidth, the AVC sends a signal to the OLTC in order to change the tap position (i.e., up or down). To avoid excessive tap changes due to short period voltage deviations (e.g., a few seconds) the tap change signal is delayed.

(2.5)



Figure 2-13 Voltage control using OLTC

2.5.2 State of the Art

The adoption of LV OLTC-fitted transformers has been recently considered as a potential solution to manage voltage issues in PV-rich LV networks. Although the OLTC is considered to be a well matured technology and widely used at higher voltage levels (e.g., 33/11kV or 33/6.6kV in the UK) [19], its adoption in LV networks is relatively a new concept and requires a thorough investigation given the complexity of LV networks (i.e., unbalanced connection of load and generation, high penetrations of LCTs, bidirectional power flow, rapid voltage deviations due to PV systems). The majority of studies found in the literature investigate rule-based control approaches that combine the use of OLTC-fitted transformers with other controllable devices (i.e., residential-scale PV systems, capacitors, in-line voltage regulators) to manage voltage issues [28, 74, 76, 113, 117-123]. Some others use more complex optimisation techniques to achieve the same goal [124-126].

Germany is perhaps the first country to adopt the use of LV OLTC-fitted transformers. This is mainly because of the large volumes of residential-scale PV systems resulting from their feed-in-tariff incentives. The German DNO EnBW Regional AG trailed the use of LV OLTC-fitted transformers and in-line voltage regulators in two German LV networks (i.e., Freiamt, Sonderbuch) [117, 118]. The control approach considered a load-depended adjustment of the OLTC voltage target (i.e., changed according to the total load variation) and the control of in-line voltage regulators based on the feeders reverse power. This approach however, requires some

understanding of the topology and feeders impedances to approximate the corresponding relationship between power and voltage. It is also important to highlight that the approach requires the installation of additional devices (i.e., in-line voltage regulators) in the feeders; hence increasing cost and potentially limiting its deployment due to practical aspects (e.g., space for in-line voltage regulators).

Another pilot project in [76], "Active intelligent low-voltage network", ran by the German DNO E.ON Mitte AG investigated the use of OLTC-fitted transformer in combination with PV systems reactive power control to manage voltage issues in an LV network in Felsberg-Niedervorschütz. Although the project showed that effective voltage management can be achieved with the proposed approach, emphasis was primarily given on the reactive compensation through PV systems which does not highlight the extent to which OLTC-fitted transformers can help to manage voltage issues in PV-rich LV networks.

The pilot project DG DemoNet-Smart LV Grid investigated the application of different control approaches to manage PV related voltage issues in an Austrian LV network (i.e., Eberstalzell) using a LV OLTC-fitted transformer combined with reactive compensation from residential-scale PV systems [74, 119-121]. The control approaches included 1) OLTC control with local (i.e., substation) monitoring only, 2) OLTC control considering line drop compensation and local monitoring only, 3) OLTC control considering line drop compensation and reactive power control of PV systems, and 4) OLTC control considering real-time smart meter data. The project, as expected, highlighted that the higher visibility provided by smart meters (i.e., voltages) can offer a better voltage regulation through the OLTC. However, no detailed information is provided on the adopted OLTC control algorithm and the analysis considered only one PV penetration level (i.e., 173 customers with 60 residential-scale PV systems). Additionally the analysis did not assess any impacts related to different control cycle lengths and the corresponding quantification of tap changes. The quantification of these parameters is important in order take more informative decisions when adopting such methods.

Similarly, authors in [113] propose a coordinated control of an LV OLTC-fitted transformer with reactive power compensation through PV inverters to manage voltages in a typical German LV network. The proposed OLTC control algorithm

makes use of remote monitoring data (i.e., voltages) at critical points in the feeders to then perform a test in order to check if a tap up or down can bring the maximum and minimum voltages within the statutory limits. Results show that voltages can indeed be managed with the proposed control strategy; however, the work was limited only to some deterministic cases of a specific PV penetration level. This, unfortunately, does not provide the full picture of potential scenarios (i.e., different penetrations and locations of PV systems). More importantly, the effect of the proposed method only quantified the voltage at a specific network point rather than considering the voltage at all customer connection points.

In [122, 123], authors investigate the effect of using a voltage range as a control input for OLTC-fitted transformers aiming to reduce the number of tap changes required to manage voltage issues in distribution networks. However, to define the corresponding voltage range at the busbar, knowledge of important feeder characteristics is essential (i.e., impedance, number of connected customers, and penetration level of PV systems). Consequently, the adoption of such an approach will have to be tuned for each network; hence, reducing its scalability.

A more detailed analysis in terms of the provision of simulation-based results considering the techno-economic assessment of different voltage control strategies including the adoption of OLTC-fitted transformers in LV networks is presented in [28]. In this study, the OLTC controller uses information from the two most critical network points (where the maximum and minimum voltages occur), to then trigger a tap change. For example, a tap change (up or down) is triggered if the voltage at one of the monitoring points is outside a predefined voltage bandwidth and the process is repeated until the corresponding monitored voltage falls back within the bandwidth limits. Nonetheless, as the OLTC control actions are triggered based only on one of the two measurements, a significant challenge might arise when both monitored voltages (max and min) are outside the bandwidth. Although the study, which considers a real German LV network (i.e., 122 households, 12 PV systems), shows that voltage issues can be managed with the proposed method, the analysis was limited only to a few demand and generation snapshots rather than a time-series analysis. More importantly, the performance is quantified considering only the "weakest" network point instead of the all customer connection points.

Several other studies propose the use of more complex techniques to provide coordination between the LV OLTC-fitted transformer and the residential-scale PV systems [124-126]. In [124], for example, authors propose an optimisation method that uses load and irradiance forecast to provide an optimal reactive power control of PV systems while reducing the number of tap changes required by the OLTC-fitted transformer. Similarly, authors in [125] propose an optimisation method to manage voltage issues while minimising the corresponding feeder losses. Considering the study in [126] a model predictive control (MPC) strategy that uses simplified grid models to predict future voltage variations is proposed to control LV OLTC-fitted transformers aiming to manage voltage issues n PV-rich LV networks. Although the application of these techniques (i.e., optimisation, MPC, forecasting) is likely to have a better performance in terms of voltage management and number of tap changes, they require extended information of the LV network topology, the number and type of connected LCTs as well as the deployment of costly communication infrastructures. This, unfortunately, creates significant deployment challenges as this information might not be readily available to DNOs. Furthermore, these methods have to be tailored to each of the thousands of LV networks in a given region.

An interesting and more recent study, based on the UK pilot project "LoVIA" [47], proposes the coordinated control of LV OLTC-fitted transformer and capacitor banks to manage contrasting voltage issues (i.e., one feeder with voltage rise and other with voltage drop) in LV networks [112]. The study proposes three control strategies where in the first two 1) a constant and 2) a time-based controlled voltage target (for the OLTC) is considered. The third control strategy considers 3) a rule-based OLTC controller which automatically updates the voltage target every control cycle, based on remote end monitoring points at each feeder. To manage contrasting voltage issues, the study proposes the use of capacitor banks to boost the voltage in those feeders experiencing significant voltage drop issues. The study, which considers a year-long stochastic analysis on real UK LV network with 351 residential customers and different PV penetration levels, highlighted that the adoption of an OLTC-fitted transformer in the investigated network can help to delay voltage issues to 50% of PV penetration (compared 30% when using off-load tap changer, i.e., no control). Although a comparable performance in terms of voltage management is achieved with all proposed methods, the rule-based controller results to the lowest number of tap changes. Despite the complete analysis and effective performance of the study, the proposed rule-based control method uses fixed compensation factors which have to be defined considering the voltage step of the each OLTC tap change, thus making the adoption of this method case specific. In addition, the method does not take into account the effect (in terms of voltage) of the new voltage target on the other remote monitoring points.

2.5.3 Summary of Gaps in Literature

Based on the studies available in the literature and discussed in this section, it is evident that the adoption of OLTC-fitted transformers can be a potential solution to manage voltage issues in LV networks. The most important gaps found in the current literature and addressed in this Thesis are highlighted and summarised below:

- The majority of studies in the literature consider German and Austrian LV networks (i.e., customers with three-phase connections). Consequently and as previously mentioned in section 2.3.2.1.2, the performance adopting an LV OLTC-fitted transformer must also be investigated in the context of single-phase customer connections to quantify the potential benefits in other LV networks (e.g., UK, France, USA).
- 2. Almost all studies in the current literature focus on control strategies that consider the coordinated control of LV OLTC-fitted transformers with other controllable technologies (e.g., residential-scale PV systems, in-line voltage regulators, capacitors). This, however, does not highlight the extent to which OLTC-fitted transformers alone can help to manage voltage issues in PV-rich LV networks.
- 3. The case studies performed in the literature are mostly limited to deterministic power flow analyses that consider specific PV penetration levels. Some others, even more limited, considered only several snapshots of load and generation profiles. Stochastic time-series analyses considering different PV penetrations levels and high granularity (e.g., scale of seconds and minutes) demand and generation profiles are essential to truly assess the performance of OLTC-fitted transformers in LV networks.
- 4. The literature highlights that in order to efficiently manage the busbar voltage and, consequently, the customer voltages, measurements across the LV network

are required to visualize its state. However, measurements in LV networks are typically non-existent and, therefore, information about the network state is not available. Consequently, remote monitoring at strategic points in the network can provide visibility of voltages close to the customers' connection points. However, considering the large number of LV networks around the world, it is crucial to understand the extent to which monitoring and its location can benefit the performance the OLTC-based voltage control.

- 5. The majority of studies in the literature do not quantify the number of tap changes. High number of tap changes can lead to the OLTC wear and tear [124], therefore the effects of the corresponding OLTC control settings, such as control cycles, is essential to be investigated.
- 6. A scalable and adaptive OLTC control logic that aims in managing contrasting voltages issues, rise (due to the presence of generation) and drop (due to loads) is yet to be developed.
Chapter 3

Control Schemes for PV-Rich LV Networks

3.1 Introduction

This chapter provides the details of the control schemes proposed in this Thesis aiming at managing voltages and/or thermal issues in PV-rich LV networks. Section 3.2, presents the proposed control approaches to actively manage residential-scale PV systems, while sections 3.3 and 3.4 correspond to the management of residential-scale battery energy storage systems and LV OLTC-fitted transformers, respectively.

3.2 Residential-scale PV Systems

This section details the control approaches to actively manage residential-scale PV systems. Sub-section 3.2.1 corresponds to the proposed decentralised control of residential-scale PV systems aiming to manage voltage issues in LV networks while sub-section 3.2.2, provides details of the proposed centralised controller that aims to manage thermal issues. Sub-section, 3.2.3, proposes the combination of the centralised and decentralised controller to manage both thermal and voltage issues in LV networks.

3.2.1 Decentralised Voltage Control

As mentioned in section 2.3, modern inverters installed in residential-scale PV system installations are already embedded with fast (i.e., fraction of second) power control functions that can be used to manage voltage issues in a decentralised way; thus requiring no communication infrastructure. Indeed, the literature shows that the use of already embedded control functions (i.e., Volt-Var, Volt-Watt) in residential-scale PV inverters can offer efficient voltage management. Nonetheless, an in depth analysis (e.g., voltage compliance, utilisation of assets, energy curtailment etc.) investigating the effects of different Volt-Var and Volt-Watt set-points (i.e., curves) might have in an LV network, is not covered in the literature. This analysis can provide important insights to DNOs, allowing them to take more informed decisions when adopting these decentralised power control functions. More importantly, such

investigation will potentially allow identifying a single set of Volt-Var or Volt-Watt curve settings that can effectively manage voltage issues in any LV networks.

To bridge this important gap, this Thesis proposes a Volt-Var and Volt-Watt analysis to investigate the effects of different set-points (i.e., curves) for each power control function. To enable this investigation, this section presents how the corresponding Volt-Var and Volt-Watt set-points (i.e., curves) can be defined (details in section 5.2).





Figure 3-1 Volt-Var curve adjustable settings

The PV inverter Volt-Var set-points are provided in the form of a Volt-Var curve (Figure 3-1). This curve, depending on the DNO requirements is defined by a variable number of points (commercial inverters usually support up to 6 points) as shown in Figure 3-1. In order to design this curve, the p.u. middle point of the statutory voltage limits is used as a reference point (*Vref*). Then based on a p.u. "deadband" value (*db*) *V*2 and *V*3 can be calculated using (3.1) and (3.2), respectively.

$$V2 = Vref - db \tag{3.1}$$

$$V3 = Vref + db \tag{3.2}$$

*V*1 and *V*4 can then be calculated using *V*2, *V*3 and the specified "activeband" (*ab*) as shown in (3.3) and (3.4), respectively.

$$V1 = V2 - ab \tag{3.3}$$

$$V4 = V3 + ab \tag{3.4}$$

For example, the set-points can be defined such that the inverter provides maximum possible reactive power at the full range of allowable voltage (V1 to V4), or possibly a more narrow range of set-points to provide much more conservative voltage regulation. The reactive power output values (i.e., Q1, Q2, Q3, and Q4) are defined as a percentage of available vars given the present active power output and the reactive power capability of the PV inverter.

3.2.1.2 Volt-Watt Set-points



Figure 3-2 Volt-Watt curve adjustable settings

The PV inverter Volt-Watt set-points are also provided in the form of a Volt-Watt curve (Figure 3-2). This curve, depending on the DNO requirements is defined by a variable number of points (commercial inverters usually support up to 6 points) as shown in Figure 3-2. In order to design this curve the p.u. middle point of the statutory voltage limits is also used as a reference point (*Vref*). Then based on a p.u. "idleband" value (*ib*) V1 and V2 can be calculated using (3.5) and (3.6), respectively.

$$V1 = Vref \tag{3.5}$$

$$V2 = V1 + ib \tag{3.6}$$

V3 and V4 can then be calculated using the V2 and the specified "activeband" (ab) and "deadband" (db) as shown in (3.7) and (3.8), respectively.

$$V3 = V2 + ab \tag{3.7}$$

$$V4 = V3 + db \tag{3.8}$$

The active power output values (i.e., *P*1, *P*2, *P*3, and *P*4) are defined as a percentage of the maximum PV power output (i.e., Pmax).

3.2.2 Centralised Thermal Control (CTC)

The vast majority of studies found in the literature focus on voltage management only; ignoring the fact that reverse power flow can result in the thermal overloading of the most expensive network assets (i.e., transformer, feeders/cables). Considering this important gap, along with the ones summarised in section 2.3.3, an innovative and scalable centralised controller is proposed here aiming to manage thermal issues in PV-rich LV networks. Compared to other methods that use complex optimisation techniques and require significant investments in communication infrastructure and network monitoring, the proposed method considers a rule-based controller using limited network observability (i.e., substation measurements only). Consequently, the proposed control logic offers significant benefits to DNOs as it does not require the installation of remote monitoring points; thus, reducing the need of costly communication infrastructure.

3.2.2.1 Control Architecture

Figure 3-3 shows the implementation architecture of the proposed centralised thermal control logic. Sensors are installed at the head of LV feeders (i.e., at the substation) to monitor the 3-phase active power for each feeder and phase.

A pyranometer located at the substation monitors the per unit clear sky irradiance as well as the one that can be seen on the PV panels. With the control logic programmed in a programmable logic controller, PLC, (also located at the substation), the known installed PV capacity per feeder (and per phase) and the monitored data transferred to the PLC, it is possible to compute the active power limit set-points (flat universal signal) for PV systems (per feeder) so that the total PV generation will not lead to transformer or feeders thermal overloads.

These active power limit set-points can then be transmitted (e.g., radio, dedicated lines, WiMAX) through a device and picked up by the corresponding group of receivers installed on PV inverters. Here, a group of receivers is considered to be the



Figure 3-3 Monitoring and Control Architecture

group of PV inverters located in a specific feeder. Thereafter, PV inverters adopt the received active power limit set-points.

Possible Communication Infrastructure

It is important to highlight that the implementation of such control scheme does not require new complex and costly communication infrastructure as active power limit set points can be sent using existing communication infrastructure such as the European radio ripple control [127, 128]. European radio ripple control is a low data rate long wave (VLF) broadcast system presently used in Germany and Hungary for various energy related applications such as [129]:

- Remote control of street lighting (lighting management)
- Power plant load management
- Customer tariff switching and load management

Radio Ripple Control Implementation Details

The transmission coverage radius by a single VLF transmitter is typically 300 to 500km thanks to ground wave propagation along earth curvature. Users (i.e., DNOs) of the European (or any other country) radio ripple control can send their control signals to the host computer via integrated services digital network (ISDN) or via the Internet. The control signals are then transmitted based on communication protocols [128] (e.g., Versacom, Semagyr) and picked up by the corresponding receivers [130, 131] installed at each PV inverter.

Each receiver has an individual address and continuously monitors every broadcast signal. If the signal being sent has an address that matches the address of the receiver, the receiver interprets the message. Otherwise, the message is ignored. In addition, "group level" addressing can be used where groups of receivers respond to the same address.

Considering the extensiveness of LV networks around the world and the corresponding number of PV system groups, it is also important to highlight that this communication infrastructure can be beneficial as it can address millions of individual devices. For example, the Versacom protocol (also known as DIN 43861-401 Type A) which is one of the major transmission protocols used can have up to 9x8 bits of address or 72 bits (i.e., a total of 2^{72} addresses).

3.2.2.2 Control Logic

The proposed control logic algorithm is divided into 2 stages.

Stage 1, considers the feeders, where the aggregated demand is estimated for each feeder ($P_{F_{n_{\varphi}}}^{demand}$, per phase) using local measurements at the substation. Once the aggregated demand is estimated a check is performed to identify the most loaded phase ("worst") of a feeder in order to deal with unbalance issues. Based on aggregated demand, in the corresponding phase, the available PV generation headroom (i.e., the maximum amount of generation that will not lead to overloading) is calculated. Using this headroom ($P_{F_{n_{\varphi}}}^{PV_Headroom}$), along with the known PV installed capacity, the active power limit set-point ($pctP_{F_n}^{s.p.}$ in %) is calculated in order to limit the generation of all PV systems in the corresponding feeder. The new

calculated active power limit set-point ensures that the generation of PV systems will not lead to feeder overloading.

Stage 2, considers the transformer where a check is performed to identify if it will face any thermal issues when settings from Stage 1 are applied. If true, the excess generation ($P_{TX}^{PV_Excess}$) required to be reduced and a new active power limit set-point ($pctP_{F_n}^{s.p.}$ in %) for each feeder is recalculated. The final active power limit set-point ensures that both transformer and feeders will not overload.

Every control cycle *i* (depending on the supported telecommunication infrastructure) the monitored active power per phase ($P_{F_{n_{\varphi}}}^{mon}$ in kW, where $\varphi \in [a, b, c]$) and the monitored irradiance are collected. The installed PV capacity per feeder ($IC_{F_{n}}^{PV}$) and per phase ($IC_{F_{n_{\varphi}}}^{PV}$) is assumed to be known.

The two stages of the control logic algorithm are detailed in the following subsections and the flowchart of each stage is presented in Figure 3-4 and Figure 3-5, respectively.

Stage 1 – Feeder Level

First, using (3.9) the aggregated demand is estimated for each feeder ($P_{F_{n_{\varphi}}}^{demand}$, per phase) using local measurements at the substation. These measurements are the active power ($P_{F_{n_{\varphi}}}^{mon}$ per feeder/phase), the p.u. irradiance ($E_{p.u.}^{PV}$) seen on the PV panels, and the known installed PV capacity ($IC_{F_{n_{\varphi}}}^{PV}$, per feeder/phase).

$$P_{F_{n_{\varphi}}}^{demand} = \begin{cases} P_{F_{n_{\varphi}}}^{mon} + \left(pctP_{F_{n_{i-1}}}^{s.p.} \times IC_{F_{n_{\varphi}}}^{PV} \right) &, pctP_{F_{n_{i-1}}}^{s.p.} < E_{p.u.i}^{PV} \\ P_{F_{n_{\varphi}}}^{mon} + \left(E_{p.u.i}^{PV} \times IC_{F_{n_{\varphi}}}^{PV} \right) &, pctP_{F_{n_{i-1}}}^{s.p.} > E_{p.u.i}^{PV} \end{cases}$$
(3.9)

Thereafter using the monitoring data at the head of each feeder $(P_{F_{n_{\varphi}}}^{mon})$, the most loaded phase (in case of reverse power flow) for each feeder is identified. This, as previously stated, is performed in order to deal with possible unbalance situations where one phase has more (or less) PV systems or loads than the other two. Therefore, any set-points that will eventually solve technical issues for the "worst" phase, they will also solve technical issues for the other two phases. Once this "worst" phase is identified, the algorithm calculates the available generation headroom ($P_{Fn_{\varphi}}^{PV_Headroom}$) using (3.10). This headroom is the maximum amount of generation that the corresponding phase can support without overloading.

$$P_{F_{n_{\varphi}}}^{PV_Headroom} = \left| P_{F_{n_{\varphi}}}^{demand} + \left(a \times P_{F_{n}}^{max} \right) \right|$$
(3.10)

where *a* corresponds to a threshold (e.g., 1.0 which corresponds to a 100% of the cable capacity $P_{F_n}^{max}$).

Using the available generation headroom $(P_{F_{n_{\varphi}}}^{PV_Headroom})$ and the phase installed PV capacity $(IC_{F_{n_{\varphi}}}^{PV})$ the algorithm calculates the active power limit set-point $(pctP_{F_{n}}^{s.p.})$ using (3.11).

$$pctP_{F_{n_{i}}}^{s.p.} = \frac{P_{F_{n_{\varphi}}}^{PV_Headroom}}{IC_{F_{n_{\varphi}}}^{PV}}, \qquad \begin{cases} 0 & , if \ pctP_{F_{n_{i}}}^{s.p.} < 0 \\ 1 & , if \ pctP_{F_{n_{i}}}^{s.p.} > 1 \end{cases}$$
(3.11)

The $pctP_{F_n}^{s.p.}$ is essentially the percentage of the value to which the maximum rated output power of all PV systems in the specific feeder will be limited to in order to keep the feeder's utilisation level at or below 100%.

Lastly, the worst case loading $(P_{F_n}^{Loading})$ is estimated using the calculated $pctP_{F_n}^{s.p.}$, the estimated demand $(P_{F_n\varphi}^{demand})$ and the clear sky p.u. irradiance $(E_{p.u.i}^{clear})$.

$$P_{F_n}^{Loading} = \begin{cases} \left(pctP_{F_{n_i}}^{s.p.} \times IC_{F_n} \right) - P_{F_n}^{demand} & \text{, if } pctP_{F_{n_i}}^{s.p.} < E_{p.u.i}^{clear} \\ \left(E_{p.u.}^{clear} \times IC_{F_n} \right) - P_{F_n}^{demand} & \text{, if } pctP_{F_{n_i}}^{s.p.} \ge E_{p.u.i}^{clear} \end{cases}$$
(3.12)

<u>Stage 2 – Transformer Level</u>

Once the active power limit set-points for each feeder and worst case loading of each feeder are calculated, stage 2 essentially checks if the settings defined by Stage 1 will not lead to transformer overloading. Therefore, if the worst case loading of the transformer exceeds the transformer's capacity (P_{TX}^{max}) then the additional generation to be reduced is calculated using (3.13).

$$P_{TX}^{PV_Excess} = \begin{cases} \sum_{1}^{N} P_{F_n}^{Loading} - (a \times P_{TX}^{max}) & , if \sum_{1}^{N} P_{F_n}^{demand} > (a \times P_{TX}^{max}) \\ 0 & , if \sum_{1}^{N} P_{F_n}^{demand} \le (a \times P_{TX}^{max}) \end{cases}$$
(3.13)

If $P_{TX}^{PV_Excess}$ is higher than zero it means that the generation has to be limited even more. In that case the $pctP_{F_n}^{s.p.}$ is recalculated for each feeder using equation (3.14) to distribute the additional generation curtailment among all PV systems. Otherwise, $pctP_{F_n}^{s.p.}$ remains the same as calculated in Stage 1.

$$pctP_{F_{n_{i}}}^{s.p.} = \begin{cases} \frac{(pctP_{F_{n_{i}}}^{s.p.} \times IC_{F_{n}}) - \frac{P_{TX}^{PV-Excess}}{N feeders}}{IC_{F_{n}}^{PV}} & , if \ pctP_{F_{n_{i}}}^{s.p.} < E_{p.u._{i}}^{clear} \\ \frac{(E_{p.u._{i}}^{clear} \times IC_{F_{n}}) - \frac{P_{TX}^{PV-Excess}}{N feeders}}{IC_{F_{n}}^{PV}} & , if \ pctP_{F_{n_{i}}}^{s.p.} \geq E_{p.u._{i}}^{clear} \end{cases}$$
(3.14)

Finally, the active power limit set-points ($pctP_{F_n}^{s.p.}$) are rounded down to the nearest integer. Once this is done, active power limit set-points are sent to the corresponding group of PV inverters. This process is carried out every control cycle. The control cycle is a parameter to specify how often the logic will provide a control, if necessary. In an *x*-min control cycle, the average monitored data of the last *x* minutes will be used for the algorithm. To give an example, if a 10-min control cycle is considered then, the monitored data of the last 10 minutes will be used for the control data of the last 10 minutes will be used for the control data of the last 10 minutes will be used for the control logic algorithm.



STAGE 1: Feeder Level

Figure 3-4 Centralised Thermal Control Flowchart (Stage 1)



STAGE 2: Transformer Level



3.2.3 Combined Centralised Thermal and Decentralised Voltage Control

Section 2.3, highlighted that almost all studies performed in the literature are based on managing voltage issues only and, in general, aim to solve issues (e.g., voltage, thermal) separately. Even though a limited number of studies [85, 86] propose solutions to simultaneously manage voltage and thermal issues, they lack practicality and scalability as they are based on complex optimisation techniques that require extensive network observability and need to be tuned for each of the thousands of LV networks in a given region.

To provide a complete active network management, aiming to solve both thermal and voltage issues, this section proposes the combination of the centralised thermal control (presented in section 3.2.2) with the decentralised Volt-Watt support. The centralised thermal control essentially aims to manage thermal issues while the decentralised control (i.e., Volt-Watt control function) aims to manage voltage issues. The proposed control method, unlike other not-easy-to-implement solutions (i.e., complex optimisation algorithms, increased network visibility), offers a more cost-effective, scalable and practical alternative as it requires limited network observability (i.e., substation measurements only).

3.2.3.1 Control Logic

The previously described centralised thermal control logic (presented in section 3.2.2) and Volt-Watt control are used simultaneously. For the latter, a curve is adopted that considers the highest voltage "idleband" so curtailment only occurs when strictly needed. This curve can also be selected considering a Volt-Watt analysis (performed in case study section 5.4). All PV inverters (in the network) are set up to follow the same curve, which takes action only if the voltage (at the connection point) is higher than V2 (see Figure 3-2). It is also important to highlight that the centralised thermal controller signals are applied first so that the Volt-Watt control is limited only to the active power limit specified by the centralised thermal controller. This is achieved as the inverter is continuously checking in real-time (i.e., fraction of a second) if there is any signal to change the active power limit. If a signal is detected, the active power limit is updated first on the PV inverter and then the Volt-Watt function adopts the new maximum generation point ($P1 = P2 = P_{max} \times$ $pctP_{F_n}^{s.p.}$, see Figure 3-2) from the corresponding Volt-Watt curve. The change of active power limit is applied within the range of milliseconds from the moment it is received; hence, latency issues in terms of the Volt-Watt function are negligible.

3.3 Residential-scale Battery Energy Storage

This section provides the details of the proposed advanced operation mode for residential-scale BES systems aiming to bring benefits to both customers (i.e., reduced grid dependency, hence reduced bills) and DNOs (i.e., voltage and thermal management). First, sub-section 3.3.1 presents the "normal operation" mode, i.e., the control mode likely to be adopted by manufacturers for householders to make the most of their PV systems. The latter is used as a base case to assess the performance of the proposed "advanced operation" mode which is presented in sub-section 3.3.2.

3.3.1 Normal Operation Mode

Although several residential-scale BES systems are currently available in the market, the exact details of their "normal operation" mode are not available; manufacturers only provide general descriptions related to the basic operating principles of their products [93, 132]. Hence, based on the most common operating principles of commercial BES systems as well as information found in the literature [133-135], the following two basic rules are considered to be adopted by the industry:

- Charging/discharging cycles during the day should be kept low in order to maintain the lifespan of the BES system. Allowing multiple cycles per day significantly reduces the BES system's lifespan [136, 137] as it can support approximately 3,000 to 5,000 full cycles [94, 95, 138, 139].
- 2. The battery should only charge from the excess PV generation and, unless otherwise specified, will only discharge during load-led periods.

Considering the above two rules and according to manufacturer's operation specifications [93, 132], the "normal operation" mode is adopted in this Thesis, and used as the base case to compare and assess the performance of the proposed "advanced operation" mode (detailed in section 3.3.2).

3.3.1.1 Description

The normal operation mode considers only one cycle per day, i.e., one charging and discharging period and the adopted operation is described as follows:

• Charging period. Occurs during daylight as soon as the PV system generates electricity. During this period, the BES system always charges the excess of PV

generation (i.e., negative net demand). The charging period ends once the PV generation is zero.

• **Discharging period.** Occurs only when the PV system is not generating (i.e., night time). During this period, the BES system discharges with a power rate equal to the household's demand (i.e., follows demands). The discharging period ends once the PV system starts generating again the following day (i.e., new charging period).

3.3.1.2 Illustrative Example

In order to demonstrate the normal operation of a BES system, Figure 3-6 is used as an illustrative example showing the demand, PV generation and BES system charge/discharge behaviour of a household with a 3kWp PV system and a 3.3kWp 10kWh BES system. The example considers 2 consecutive sunny days. For simplicity, the same generation and demand profiles are assumed for each day.



Figure 3-6 Illustrative example of normal operation mode

The first day of the example is assumed to be the first day where the BES system is installed at the household and, therefore, its state of charge (SOC, dashed blue line) is considered to be equal to the lower battery reserve limit (i.e., 10%). Upper and lower reserve limits, mean that the BES system is able to charge until the upper limit and discharge until the lower reserve limit of the available capacity. Such limits help maintaining the lifespan of the corresponding battery [136]. Since the SOC is at the lower level, the BES system is not able to supply any energy to the household (black line); hence, its charge/discharge profile (red line) remains zero until 8am. From 8am

Control Schemes for PV-Rich LV Networks

onwards, when the PV generation increases (magenta line), the BES system starts charging (i.e., positive values) with a power rate equal to the excess PV generation (i.e., PV generation minus the demand). While charging, the SOC is increasing as more energy is stored in the battery and, for this case, reaches full SOC (i.e., 90% due to the upper reserve) around 2pm. Once the SOC reaches the maximum, the charge/discharge profile goes to zero and remains there until the time where the PV system is not generating anymore. Once the PV generation is equal to zero (around 8pm), the BES system starts discharging (i.e., positive values) with a power rate equal to the household demand. While discharging, it can be seen that the SOC is decreasing until the next day.

On the second day, the BES system is still supplying the household demand during the early hours. Since the total energy stored in the BES is higher than the energy demand of the corresponding household, the battery starts charging with some energy already stored (i.e., around 25% of SOC). This, however, forces the battery to reach full SOC even earlier in the following day. Indeed, as demonstrated in Figure 3-6, a full SOC is achieved around 12pm on the second day, which is 2 hours earlier than the first day.

3.3.1.3 Limitations

The adoption of residential-scale BES systems using the "normal operation" mode as described above suffers of the following two important limitations:

- **BES systems might not adequately discharge.** Due to the uncertainty of household demand each day, the normal operation modes might not be able to fully discharge the battery overnight. This in turn limits the ability of the battery to fully utilise its storage capacity the following day. This issue becomes even more critical in cases where the household demand is significantly low.
- **BES systems reach full SOC very early.** Due to the fact that the normal operation modes cannot guarantee full discharge of the battery during the night, the battery might become full earlier the next day. From a DNO perspective, this operation might not provide significant benefits in terms of voltage and thermal issues since batteries might not be able to absorb excess PV generation (exports) during critical periods (around midday).

3.3.2 Advanced Operation Mode

This section presents the proposed "advanced operation" mode for residential-scale BES systems which is based on the common operating principles of commercial BES systems previously discussed in section 3.3.1 and, hence, still beneficial to customers. It overcomes the limitations discussed in the previous section by modifying the charging/discharging rates so as to minimise network impacts.

To overcome the limitations discussed in section 3.3.1.2, the proposed method defines specific times and final SOC for the charging and discharging periods in a way that is beneficial to both customers and the LV network (i.e., DNO). By knowing when a charging/discharging period starts and ends, it is possible to calculate the charging/discharging power rate that makes the BES system progressively charge/discharge to achieve the defined final SOC.

Furthermore, to play an active role in the management of voltages during daylight, the proposed advanced operation mode also increases the charging power rate when the voltage at the connection point rises above the statutory limit. In addition, to ensure the demand is always met, the discharging rate is increased when needed. It is important to note that the performance of the proposed advanced operation mode might be influenced by the size of the installed BES systems, as the ability to manage issues depends on their storage capability.

The control logic of the proposed advanced operation mode, which is based only on one cycle per day (one charging and one discharging period), is detailed below and the flowchart of the algorithm is given in Figure 3-8.

1. Defining Charging and Discharging Periods and Final SOCs

The lengths of the charging (T_c , in hours) and discharging ($24 - T_c$) periods need to be defined as well as the corresponding final states of charge, SOC_{final}^{C} and SOC_{final}^{D} , respectively. Although the latter two can adopt diverse values, it is suggested to have a fully charged battery at the end of the charging period and an empty battery at the end of the discharging period (in both cases taking account of the reserve limits). The former (full battery) allows collecting the maximum possible "green" energy from the PV system and, therefore, increasing the customer's self-consumption (i.e., reduced grid dependence) while the latter (empty battery) ensures that adequate room will be available in the battery to charge the following day. In addition, the charging and discharging periods should be defined based on the time at which the peak sun irradiance is occurring, t_{peak} . Figure 3-7 illustrates this considering the local seasonal irradiance profile.

- a) Charging Period: The total charging period (T_c) is defined based on the DNO and customer requirements and should always be equal or smaller than the daylight period (total time between sunrise and sunset). Once T_c is defined, the time where the charging period starts (t_c^{start}) and ends (t_c^{end}) is set to be $t_{peak} - \frac{T_c}{2}$ and $t_{peak} + \frac{T_c}{2}$, respectively. This setting, as shown in Figure 3-7, allows having a charging period which is symmetrical to the peak irradiance time, t_{peak} .
- b) Discharging Period: Once the charging period is defined, the discharging period is defined as the time between the end of the charging period until the start of the next charging period (i.e., next day).



Figure 3-7 Charging and discharging period definition

2. Charging and Discharging Power Rates

The controller progressively charges or discharges the BES system based on a power rate proportional to the amount of energy required to charge or discharge so the desired final SOC (SOC_{final}^{C} or SOC_{final}^{D}) can be achieved.

The following process which is also graphically demonstrated with the flowchart given in Figure 3-8, is performed locally at each BES system for each instant t.

Charging power rate. The BES system is restricted to charge only from the excess of PV generation $(P_t^{PV_excess})$ at the corresponding instant *t*, calculated using (3.15). $P_t^{PV_excess}$ is defined as the difference between PV generation (P_t^{PV}) and the household demand (P_t^{demand}) . The latter two are based on local measurements (i.e., generation and demand of the household) which are taken every instant *t*.

$$P_t^{PV_excess} = P_t^{PV} - P_t^{demand}$$
(3.15)

If the $P_t^{PV_excess}$ is equal or less than zero (i.e., demand is higher than generation) then the charging power rate is set to zero, as there is no excess of PV generation to charge from. If the $P_t^{PV_excess}$ is higher than zero, then the charging power rate P_{t+1}^C is calculated using (3.16) which is subject to the restriction in (3.17), where SOC_t is the SOC at instant *t* and $BES_{KWh}^{capacity}$ is the system's capacity in kWh.

$$P_{t+1}^{C} = \begin{cases} 0 , \text{ if } P_{t}^{PV_{excess}} \leq 0\\ \frac{(SOC_{final}^{c} - SOC_{t}) \times BES_{KWh}^{capacity}}{(t_{peak} + \frac{T_{C}}{2}) - t} , \text{ if } P_{t}^{PV_{excess}} > 0 \end{cases}$$
(3.16)

$$P_{t+1}^{C} \in [0, P_{t}^{PV_{excess}}]$$
(3.17)

Discharging power rate. When in the discharging period, the discharging power rate for the next instant (P_{t+1}^D) is calculated using (3.18).

$$P_{t+1}^{D} = \frac{(SOC_t - SOC_{final}^{D}) \times BES_{KWh}^{capacity}}{\left(t_{peak} - \frac{T_c}{2}\right) - t}$$
(3.18)

It is important to highlight that if the household demand (P_t^{demand}) is larger than the discharging power rate (P_{t+1}^D) , calculated in (3.18), then the P_{t+1}^D is automatically set to be equal to the P_t^{demand} so that the BES system will always follow the local demand (i.e., self-consumption). This is true provided the battery has enough stored energy in the early morning.

3. Voltage Control Feature

In order to directly manage voltage issues, a voltage control feature is considered in the charging period of the proposed advanced operation mode. This feature is triggered once the household voltage, V_t (measured at each instant), reaches the statutory limit (V_{limit}^{upper}). When this happens, the net household active power is recorded and is considered to be the household's active power export limit ($P_t^{exp_lim}$) that results to a voltage close to the statutory limit. This limit is then updated for every subsequent instant ($P_{t+1}^{exp_lim}$) using (3.19); where *a* corresponds to a threshold (e.g., 1.0 corresponds to a 100% of the statutory voltage limit). This equation smoothens the values of the limits (given the changing nature of demand and generation as well as the effects of other customers on voltages) and allows tuning the limit towards more conservative values (lower *a*) if needed.

$$P_{t+1}^{exp_lim} = \frac{(a \times V_{limit}^{upper}) \times P_t^{exp_lim}}{V_t}$$
(3.19)

For as long as the voltage control feature is on, the charging power rate P_{t+1}^{C} is updated accordingly in order to satisfy the household export limit $P_{t+1}^{exp_lim}$ and, therefore, keep the voltage close to the statutory limit. To perform the update, first, the household's export power $(P_{t+1}^{house_exp})$ resulting from the P_{t+1}^{C} in (3.16) is calculated using (3.20). It is important to highlight that, in this case, a persistent forecasting is assumed for the household demand and generation. This means that, for this equation only, the excess of PV generation at instant t $(P_t^{PV_excess})$ is assumed to be the same for the next instant t+1 $(P_{t+1}^{PV_excess})$.

$$P_{t+1}^{house_exp} = P_{t+1}^{PV_excess} - P_{t+1}^{C}$$
(3.20)

If the $P_{t+1}^{house_exp}$ is found to be larger than the $P_{t+1}^{exp_lim}$, it means that the charging power rate, calculated using (3.16) needs to increase (to reduce the household's export power) in order to satisfy the export power limit. The updated P_{t+1}^{C} is then calculated using (3.21).

$$P_{t+1}^{C} = P_{t+1}^{PV_excess} - P_{t+1}^{exp_lim}, \quad if \ P_{t+1}^{house_exp} > P_{t+1}^{exp_lim}$$
(3.21)

If, however, the $P_{t+1}^{house_exp}$ is lower than or equal to the export limit ($P_{t+1}^{exp_lim}$), the charging power rate is not updated, i.e., remains as calculated in (3.16). The voltage control feature is deactivated once the charging period ends.



Figure 3-8 Advanced operation mode flowchart

3.3.2.1 Illustrative Example

Figure 3-9 demonstrates the behaviour of the proposed advanced operation of a BES system using the same example as in section 3.3.1.2 (i.e., household with 3kWp PV system and a 3.3kWp 10kWh BES system).



Figure 3-9 Illustrative example of advanced operation mode

As also described previously (3.3.1.2), the first day of the example is assumed to be the first day where the BES system is installed at the household and, therefore, it is considered to be empty (i.e., SOC at the lowest point, 10%). Considering this, the BES system is not able to supply any energy to the household (black line); hence, its charge/discharge profile (red line) remains zero until 8am when the discharging period finishes. From 8am onwards, when the charging period begins (i.e., here assumed to be 8am to 6pm), the BES system starts to progressively charge with a power rate specified by the proposed advanced mode. The proposed progressive power rate provides a solution to the limitation of the normal operation mode where the BES systems become full very early during the day. As observed in Figure 3-9, the SOC is increasing linearly as more energy is stored in the battery and it can be seen that the proposed advanced mode manages to reach full SOC at the assumed specified end of the charging period (i.e., here assumed to be 6pm).

It is also important to highlight that in the case where voltage (at the household connection point) increases above 1.1p.u., the proposed advanced mode will increase the charging power rate (as detailed in section 3.3.2) in order to bring the voltage below 1.1p.u. In such case, SOC is expected to increase faster as additional energy is

required to be stored. Considering the nature of the progressive charging (linear increment of SOC until the end of the specified charging period), when voltage issues occur during midday (peak sun radiation), SOC will always be close to 50%. Hence almost 50% of the storage headroom will always be available to use in order to manage the corresponding voltage issues.

Once the BES system goes into the discharging mode (i.e., 6pm to 8am for this example) the battery is discharging with a power rate equal to the household demand. As detailed in section 3.3.2, the proposed advanced operation mode (when in discharging period) will progressively discharge with a power rate specified by the controller. However, if the household demand is higher than the specified discharging rate, it automatically follows the demand as indeed happens in this case. For example, at 6pm (beginning of example's discharging period) the discharging power rate, calculated by the controller, is close to 0.7kW (i.e., 10kWh/14hrs) which is lower than the household demand; hence, the BES system is discharging with a power equal to the demand. For this particular case, it is shown that the BES system is becoming empty at an earlier time (i.e., 4am) than the specified one (i.e., 8am – end of discharging period) due to the high discharging power rates caused by following the household demand.

It is also important to mention that in cases where household demand is significantly lower than the stored energy, the progressive discharge of the proposed advanced operation mode guarantees that the BES system will fully discharge by the start of the charging period. This, as a consequence, overcomes the drawbacks of the normal operation mode and allows the full utilisation of the storage capacity in every charging period.

More importantly, Figure 3-9 shows that the same expected performance is also noticed in the following day providing a very good utilisation of the BES system, compared to the performance when using the normal operation.

3.4 LV OLTC-fitted Transformers

This section provides the details of the scalable and adaptive OLTC control logic proposed in this Thesis which aims at managing contrasting voltages issues (rise and drop). Considering the gaps identified in the current literature, the proposed method allows the automatic update of the voltage target (at the busbar) according to network conditions while using limited network observability (only from critical remote monitoring points). Crucially, this provides the significant benefit of being easy to adapt to network changes (i.e., additional PV system installations or loads) without the need of reconfiguring OLTC settings.

3.4.1 Architecture

Figure 3-10 shows a simplified schematic, demonstrating the control architecture of the proposed control scheme which considers metrology and communication units (MCUs) at critical points of LV feeders (e.g., middle or end). The MCUs collect and send the corresponding monitoring data (i.e., voltages) to the remote terminal unit (RTU) located at the secondary (i.e., MV/LV) distribution substation. The RTU is, in this case, the physical device in which any control logic is coded. Based on this logic, the RTU can then send to the OLTC controller a command to produce a busbar voltage (V^{target}) that ultimately alleviates any potential voltage issues.



Figure 3-10 OLTC proposed control architecture

3.4.2 Control Logic

The control logic considers voltages at remote points to calculate the voltage at the busbar (V^{target} , to be sent to the OLTC controller) required to bring voltages along the feeders within the statutory limits.

At every control cycle (*i*), the minimum and maximum values (V_i^{min}, V_i^{max}) of all monitored line-to-neutral phase voltages (V_i^a, V_i^b, V_i^c) at the remote points of the feeders are found. Then, it is checked whether these two values are within the controller's safe zone as shown in Figure 3-11. The latter (i.e., the safe zone), considers a bandwidth (green area in Figure 3-11) which is defined as a percentage value above and below a reference voltage (V_{ref}) . The reference voltage and safe zone are parameters that can be adjusted according to the DNO requirements. Thereafter, and as illustrated in Figure 3-11 (a), if both values V_i^{min} and V_i^{max} are within the safe zone, then no action is taken and the controller proceeds to the next control cycle. On the other hand, and as shown in Figure 3-11 (b), if any of the two values lies outside the safe zone, then their differences from the V_{ref} are calculated $(\Delta V_i^{min}, \Delta V_i^{max})$. The average of these voltage differences (ΔV_i^{mean}) is then used to calculate the new voltage target (V_{i+1}^{target}) as in (3.22).

$$V_{i+1}^{target} = V_i^{target} + \Delta V_i^{mean}$$
(3.22)

If the new voltage target is outside the statutory limits $(V_{limit}^{min}, V_{limit}^{max})$, then the corresponding value is set to its nearest limit. Furthermore, using (3.23), the expected values of the remote monitoring points are estimated with a simple –yet effective–linear expression. If these values are within the statutory limits, the new voltage target (V_{i+1}^{target}) is sent to the OLTC controller. Otherwise, the voltage target is not changed $(V_{i+1}^{target} = V_i^{target})$.

$$V_{estimation}^{monitored} = \frac{V_{i+1}^{target} \cdot V_i^{monitored}}{V_i^{target}}$$
(3.23)

The flowchart for the proposed control logic is given in Figure 3-12. It is important to highlight that due to the nature of the control logic (i.e., considering both highest and lowest voltages), the voltage target calculated will be influenced by the higher

difference $(\Delta V_i^{min}, \Delta V_i^{max})$ and, therefore, the new voltage target is expected to maintain the monitored voltages within the statutory voltage limits. It is important to highlight that in case of simultaneous voltage rise and drop of same magnitude, the average voltage difference will be equal to zero and, therefore, the voltage target will not change.



Figure 3-11 Controller example



Figure 3-12 OLTC control logic

Chapter 4

Case Study Part 1: Modelling

4.1 Introduction

As previously mentioned in section 1.2.4, one of the main challenges associated with the corresponding research and also found to be one of the gaps in the literature is the lack of realistic modelling, including the use of high resolution time-series analyses of LV networks. To bridge this gap, detailed data of a real French LV network, smart meter data from 975 customers, and 3 real monitored PV generation profiles were facilitated by the industrial partner EDF R&D for the purposes of this Thesis. These data, which are detailed in this chapter, allow modelling real French LV networks to then perform realistic, high resolution (i.e., 1-min) time-series analyses.

4.2 Real French LV Network

A real French LV network, located in Paris¹, and operated by ERDF, is adopted as the test LV network to assess the performance of the active network management schemes proposed in this Thesis. According to the industrial partner (EDF R&D), the corresponding network is selected as it is considered to be a typical LV network in France. This allows understanding the extent to which the proposed active network management schemes can provide benefits in terms of voltage and thermal management.

The topology and main characteristics per feeder (i.e., single-phase customers and the total sum of the 3-ph cable length) are shown in Figure 4-1. The LV network consists of four feeders (each represented in a different colour) and a total of 162 domestic single-phase customers (blue dots). These feeders are supplied by a three-phase 400kVA single distribution (i.e., 20/0.41kV) transformer (black triangle).

¹The exact geographical location of the network is not provided due to confidentiality aspects.



Figure 4-1 French LV network

4.2.1 Transformer

The transformer is equipped with an off-load tap changer having a range of +1/-1 (taps) with 2.5% per tap, i.e., 3 off-load tap positions in total. Assuming (and adopted in the analyses) that the voltage at the primary side of the MV/LV transformer is the nominal line-to-line voltage (i.e., 20kV), the busbar voltages corresponding to different off-load tap positions are shown in Table 4-1. According to the data provided, the off-load tap changer of the test LV network is set to tap position 2 which results on the nominal transformation of voltages (i.e., 20kV to 400V).

Off-load	MV	LV		Vbase = 400V
Tap position	L-L (V)	L-L (V)	L-N (V)	V (p.u.)
1 (+2.5%)	20000	400	230.9	1
2	20000	410	236.7	1.025
3 (-2.5%)	20000	420.3	242.6	1.051

Table 4-1 Voltage regulation of the modelled LV off-load tap changer

4.3 Real Smart Meter Data

Data recorded from 975 smart meters, of both residential and commercial customers, located in France were also provided in the form of .csv files. These files include information of the date, time and power of each of the customers with a 30-min resolution for the period of October 2013 to June 2014. The data were collected from different types of French customers, as shown in Table 4-2, with different subscribed

powers ranging from 3 to 36kW. These data are used to realistically model the individual daily behaviour of LV customers.

RES1 (Blue Residential Base)	Residential - Flat tariff - Power subscribed =< 6kVA
RES11 (Blue Residential Base)	Residential - Flat tariff - Power subscribed > 6kVA
RES2 (Blue Residential HPHC ²)	Residential - Time of use tariff
PRO1 (Blue Professional Base)	Professional - Flat tariff
PRO2 (Blue Professional HPHC)	Professional - Time of use tariff

Table 4-2 Types of customers in France

Figure 4-2, shows the composition of the received data by type (Figure 4-2 (a)) and subscribed customer power (Figure 4-2 (b)). As shown in Figure 4-2 (a) the majority of the smart meter data is coming from RES2 (44%) and RES1 (37%) type of customers where the rest are from RES11 (9%), PRO1 (7%) and PRO2 (3%). It is important to highlight that 60% of the total smart meter data are customers with a 6kW subscribed power, 15% with 9kW and 11% with 12kW. The remaining 14% of data is divided to 36, 30, 24, 18, 15 and 3kW subscribed customers.





According to the Regulatory Commission of Energy (CRE - Commission de régulation de l'énergie) [140], the distribution of EDF's customers among different types is as shown in Figure 4-3. Therefore, considering only the Blue Residential and Professional types (93% of the total connections in France) the proportions of RES1+RES11, RES2, PRO1, and PRO2 types are 48, 41.4, 8.4, and 2.1%, respectively. These proportions highlight that the smart data received represent a realistic sample of the real distribution of customers in France.

² FR: Heures Pleines / Heures Creuses – EN: Peak hours/ Off-Peak hours



Туре	Customers		
RES1+RES11	13.7 million		
RES2	11.8 million		
PRO1	2.4 million		
PRO2	0.6 million		

Figure 4-3 Distribution of EDF's customers among the different types [140]

4.3.1 Construction of Domestic Load Profiles

The smart meter data are used to realistically model the load behaviour in the simulations performed in this Thesis. The power information was extracted and daily half-hourly load profiles were created for all types of customers, subscribed power, month and type of day. An example of how load profiles were created is shown in Figure 4-4 for a RES1 - 3kW customer in October. The recorded load profile (3rd column - Watts) of a specific weekday (1st column - 1st October 2013) for hours 00:00 to 23:30 (2nd column - 48 points) constitutes a single load profile to be added in the October's pool of weekday load profiles.



Figure 4-4 Load Profile Creation Example

A total of 179,850 load profiles were created and categorised in folders based on the type of customer. Each type of customer was categorised in sub-folders for each subscribed power including subcategories for the 12 different months. Each month contains 9 categories/folders (one for each available subscribed power) where each subscribed power folder contains a pool of daily load profiles categorised by the type of day (i.e., weekday, weekend). Figure 4-5 shows how the data were categorised in

folders to create pools of load profiles to then use and realistically model the load behaviour in the analyses.



Figure 4-5 Smart meter data categorisation

Since in this Thesis all analyses consider 1-min time-series simulations, it is important to highlight that the 30-min resolution (48 points) load profiles are converted to 1-min resolution (1440 points) as shown in Figure 4-6. Each point within a segment of 30 minutes in Figure 4-6 (b) has the same value that corresponds to the specific half-hourly value in Figure 4-6 (a).



Figure 4-6 Conversion of a load profile from (a) 30-min to (b) 1-min resolution

4.3.2 Allocation of Domestic Load Profiles

The available information regarding the loads (i.e., customers) in the real French LV network is limited only to the subscribed power of customers connected in each feeder – in this case, 3, 6, 9 and 12kW. Therefore, real French domestic load profiles, created from smart meter data as described in the previous section 4.3.1, are used for the analyses.

As previously mentioned, according to the French Regulatory Commission of Energy [140], the number of RES1+RES11, RES2, PRO1, and PRO2 connections in France are 13.7, 11.8, 2.4, and 0.6 million, respectively. To make the analyses more realistic, the random distribution of domestic loads in the network considers the percentages of different types of customers using the information of the number of connections for each type as shown in Table 4-3. It is important to highlight that the share of customers between RES1 and RES11 was assumed to be the same share as in the smart meter dataset.

Туре	Customer Connections
RES1	11.1 million
RES11	2.6 million
RES2	11.8 million
PRO1	2.4 million
PRO2	0.6 million

 Table 4-3 Number of connections for each type of customer [140]

Therefore, the type of each load in the test LV network is assigned based on the subscribed power and the total number of connections per type. Once the type of customer is assigned, a load profile is also assigned to that customer model using a random profile from the pool of load profiles (created using the smart meter data). The procedure can be described with the following example.

Assuming a customer of a 6kW subscribed power; its type could be RES1, RES2, PRO1 or PRO2. Considering the information shown in Table 4-3 a customer type is randomly selected considering the statistical probability of being a specific type as defined by (4.1), (4.2), (4.3) and (4.4).

$$P_{RES1} = \frac{11.1}{(11.1 + 11.8 + 2.4 + 0.6)} = 42.9\%$$
(4.1)

$$P_{RES2} = \frac{11.8}{(11.1 + 11.8 + 2.4 + 0.6)} = 45.6\%$$
(4.2)

$$P_{PRO1} = \frac{2.4}{(11.1 + 11.8 + 2.4 + 0.6)} = 9.3\%$$
(4.3)

$$P_{PRO2} = \frac{0.6}{(11.1 + 11.8 + 2.4 + 0.6)} = 2.3\%$$
(4.4)

Once the customer's type is selected (based on the statistical probabilities above), the procedure to select a load profile for the corresponding customer model is demonstrated in Figure 4-5 and described below:

- First the folder containing all the load profiles for the assigned type of customer is selected. For demonstration purposes it is assumed that a RES2 type of customer is assigned (using on the statistical probabilities above).
- 2. Then, within the customer type folder, the folder containing all the created load profiles for the month required to perform the analysis is selected.
- 3. Within the selected month folder, the folder containing all the load profiles for the specific subscribed customer power is selected. For demonstration purposes the subscribed customer power is assumed to be 6kW.
- 4. Within the subscribed customer power, the folder containing the load profiles for the type of day that the analysis needs to be performed for, is selected.
- 5. Finally, a random daily load profile is selected from the pool of profiles contained in the last selected folder. This profile is used to model the corresponding customer's 24-hr power demand.

4.4 Realistic PV Generation Profiles

PV generation profiles are modelled using an available tool developed by the University of Loughborough, Centre for Renewable Energy Systems Technology (CREST) [141]. In the CREST tool, once the day and the configuration information of the installed PV panel (e.g., location, efficiency, area, etc.) are determined, 1-min

resolution generation and net radiation profiles can be produced. Both of the profiles correspond to the outdoor irradiance on the chosen day of the year.

Using the aforementioned tool [141], considering PV panels with tilt angle of 35° , azimuth of 0° (south facing), and Paris as the geographical location, it was possible to extract the daily radiation profile of the middle day of each month of the year and generate 100 daily sky brightness/cloudiness indexes (B_{index}) for each month. Using these data, another tool is created to randomly generate sunny PV profiles for each month. The new tool, using equation (4.5), essentially replicates the CREST tool procedure to generate the sunniest PV profiles for each month. Figure 4-7 (a) and (b), shows 4 random daily PV generation profiles (3 and 6kWp systems) for January (winter) and June (summer), respectively.

$$PV_{profile} = \eta_{panel} \times \eta_{inverter} \times A_{panel} \times H_{month} \times B_{index}$$
(4.5)

where η_{panel} and $\eta_{inverter}$ are the panel and inverter efficiencies, respectively. The A_{panel} is the area of the panel and the H_{month} is the irradiance profile of the corresponding month.





4.4.1 Comparison with Real French PV Generation Profiles

Three real monitored PV generation profiles, from the region of Paris, were also provided by the industrial partner (EDF R&D) for the purposes of this Thesis. These profiles have a 10-sec resolution and were monitored during the days 9th, 11th and 14th of April 2014. In order to understand how realistic are the PV profiles created with the CREST tool (1-min resolution) the real PV profiles were converted to 1-min

resolution, as shown in Figure 4-8. The assessment of two basic metrics (capacity factor and %difference to minute by minute power amplitude) showed that the modelled PV profiles, using the CREST tool, can be considered realistic. The comparison is described in the following sections.

4.4.1.1 Capacity Factor

Three random sunny PV generation profiles in April (Figure 4-8 - blue profiles) were created using the CREST tool to be compared with the real PV profiles (Figure 4-8 - red profiles). The daily capacity factor of the real PV profiles found to be 27%, 23% and 27%, respectively and 26%, 28%, and 27% for the modelled PV profiles, respectively. As shown in Figure 4-8, the results highlight that both profiles real and modelled have almost the same capacity factor.



Figure 4-8 Daily Capacity Factor for Real and Modelled PV Profiles

4.4.1.2 Power Amplitude

The same profiles were analysed to calculate the minute by minute percentage difference of power amplitude during the hours of sunshine (08:30 to 20:00). Figure 4-9, shows the percentage difference (%difference) of power amplitude for each of the PV profiles (real and modelled). As it can be seen, during real conditions, power output can vary up to 125% from minute to minute. These high variations usually appear during the midday where the solar irradiance is high. Therefore, the amplitude can change rapidly from minute to minute due to the sky clearness effect (clouds

passing through). Almost the same variations are noticed with the modelled PV profiles, however, the maximum variation of power amplitude is lower than the real PV profiles by 25%. Therefore, the modelled profiles can be considered as realistic and are used for the analyses.



Figure 4-9 Minute to Minute %Difference of PV Power Amplitude

4.4.2 Allocation of Residential-scale PV Systems and Generation Profiles

A PV system allocation is adopted considering uneven penetrations per feeder and multiple installed capacities. This allows having feeders with higher PV penetration than others, resulting in more realistic and challenging scenarios (i.e., one feeder with higher voltages than other). This proposed PV allocation approach, which was endorsed by the industrial partner EDF R&D, is considered to realistically represent the current and future trend of residential-scale PV system installations in France.

Selected individual customers in the network are allocated PV systems with capacities based on their subscribed power as described below:

 Customers with subscribed power of 3-36kW are allocated up to a maximum of 2 basic PV systems (a basic PV system is considered to have an installed capacity of 3kW).
Customers with subscribed power of >36kW are allocated with 1 basic PV system per 12kW of subscribed power.

Considering the above rules, the following definitions is considered to understand the concept of PV penetration (PV%):

<u>Maximum Potential PV Hosting Capacity (MPHC)</u>: The maximum potential PV hosting capacity is the one resulting from having all customers with the maximum possible PV installed capacities (worst case scenario). In this case, all customers with a subscribed power 3-36kWp are allocated 2 basic PV systems and customers with a >36kWp subscribed power are allocated with 1 basic PV system per 12kW of subscribed power. Hence, the PV penetration (PV%) corresponding to a given installed capacity is calculated as the percentage of the MPHC.

Depending on the investigated PV penetration (PV%), PV systems are randomly allocated in the network until the corresponding fraction of MPHC in the network is reached. This is done having as a constraint the maximum PV capacity per customer as specified above.

4.4.2.1 Specifications of PV Inverters

The studies performed in this report consider the installation of PV inverters which are commonly used in residential-scale PV installations in the UK, France, and Europe [73]. The basic characteristics and specifications of the modelled inverters are given in Table 4-4:

OUTPUT	
Apparent Power (S)	3/6/12 kVA
Active Power (P)	3/6/12 kWp
Reactive Power (Q)	0.9(leading) to 0.9(lagging)
CONTROL FUNCTIONS	
Volt-Var Control	Q(V) curve, up to 6 point curve
Volt-Watt Control	P(V) curve, up to 6 point curve
Active power limit	Limit of generation up to a % of the Pmax
COMMUNICATION INTERFACES	
WiFi	2-way communication
Radio Ripple Control Receiver (RRCR)	2-way communication
Local Area Network (LAN) via Ethernet	2-way communication

Table 4-4 Modelled PV	Inverter -	Basic s	specifications
------------------------------	------------	---------	----------------

4.5 Monte Carlo Methodology

To cater for the uncertainties related to household demand as well as PV generation and location, the Monte Carlo methodology developed in [15] is adapted and used in this Thesis to assess the performance of the proposed control schemes. The main steps for a single simulation, given a specific network and PV penetration level, are described below and summarised in Figure 4-10.

- 1. The type of customers is selected based on their subscribed power and the calculated probability of each customer type as described in section 4.3.2.
- 2. For each customer, a random daily load profile is selected from the pool of load profiles corresponding to the customer's selected day/season, type, subscribed power and type of day.
- 3. Depending on the PV penetration level (PV%) PV systems are randomly allocated in the network based on the procedure described in section 4.4.2.
- 4. PV generation profiles are generated considering the season and the PV system capacity.
 - a. Unbalanced, high resolution, time-series, three-phase four-wire power flows over a 24-hour period are carried out considering business as usual (BAU) scenario (without active management control). Simulation results are collected (voltage, current, power, etc.).
 - b. Unbalanced, high resolution, time-series, three-phase four-wire power flows over a 24-hour period are carried out considering a given active network management scheme. Simulation results are collected (voltage, current, power, etc.).

The process described above is repeated one hundred times for each investigated PV penetration level. This number, as suggested and proved in [15], provides a trade-off between computational time and accuracy of the results. This allows to capture the stochastic nature of the LV network as the power flow analysis (voltage, current, power, etc.) from each daily simulation will be influenced differently; according to the load profiles, the location and size of the PV systems. The corresponding daily

results (voltage, current, power, etc.) are stored for every single simulation. After a complete Monte Carlo analysis (i.e., 100 simulations of each investigated PV penetration level), the results from all the daily time-series simulations are analysed to assess the performance of the proposed control schemes using specific metrics (detailed in the following sections).



Figure 4-10 Monte Carlo Methodology

4.6 Performance Metrics

To quantify the technical impacts caused by different penetrations of residentialscale PV systems and also assess the performance of the proposed control schemes presented in Chapter 3, several performance metrics are adopted for the simulation analyses. This section, presents the metrics adopted in all simulation analyses regardless the case study (following chapters). Different case-specific metrics that might be required to certain specific control schemes are detailed in the corresponding chapters.

4.6.1 Voltage Issues

To understand the impacts of the residential-scale PV systems in terms of the voltage performance, the number of customers with voltage issues is calculated in each simulation.

<u>Number of customers with voltage issues</u>: This metric takes the voltage profile calculated for each customer connection point from the power flow simulation to then check if the European Standard EN 50160 [142], adopting the French statutory voltage limits (i.e., +/- 10% of nominal), is satisfied. If the customer's voltage does not comply with the standard, then this costumer is considered to have a voltage issue. Thus, the total number of EN 50160 non-compliant customers in the network is calculated.

The EN 50160 [142] indicates that the nominal voltage (U_n) in LV networks is 230 V (between phases and neutral) and:

"Under normal operating conditions, excluding situations arising from faults or voltages interruptions,

- During each period of one week 95% of the 10 min mean rms values of the supply voltage shall be within the range U_n +/- 10%.
- All 10-minute mean rms values of the supply voltage shall be within the range of U_n+10%/-15%." [142].

Since the time-series load and PV generation profiles have a resolution of 1 minute, the daily voltage profiles for each customer in the feeder are averaged in every 10 minutes to make the calculation according to EN 50160. Once the number of EN 50160 non-compliant customers is calculated for each simulation, the average and standard deviation are determined for each penetration level.

It is important to clarify that the compliance of customer connection points with the EN 50160 standard is used here for quantification purposes. Furthermore, the standard considers a week-long analysis instead of a single day. Consequently, the quantification of non-compliant customers as adopted in this work is a good metric

but does not necessarily mean that the corresponding customers will actually experience voltage issues.

4.6.2 Thermal Issues

To understand the impacts of the residential-scale PV systems in the adequacy (capacity to supply demand) of LV networks, the utilization factor at the head of the feeders and transformer is calculated in each simulation.

<u>Utilisation level of feeder</u>: This metric assesses the utilisation level in the main segment of the feeder. This index is calculated as the 10 minute maximum current divided by the ampacity (cable rating) of the main segment of the feeder. To calculate the 10 minute maximum current, the current in the main feeder calculated from the simulation (1-min resolution) is averaged in every 10 minutes.

<u>Utilisation level of transformer</u>: This metric assesses the utilisation level of the substation transformer. This index is calculated as the 10 minute maximum power divided by the transformer capacity (i.e., 400 kVA). To calculate the 10 minute maximum power, the power at the transformer secondary side is calculated from the simulation (1-min resolution) is averaged in every 10 minutes.

The idea of these metrics is to show how the utilisation of the most important and expensive assets (i.e., feeder cables, transformer) of the network behaves with different PV penetration levels. These metrics allow visualising the assets' utilisation levels and therefore identifying if they increase above their maximum specified limits (i.e., thermal limits). It is important to highlight that increasing the utilisation level of the assets above their limits might lead to the increment of their insulation temperature above their operational limit which may result in damaging or accelerating the ageing of the corresponding assets. Crucially, these metrics help understanding how adopting the proposed control schemes can help keep the utilisation level of the assets always at or below their maximum limit.

4.6.3 Energy Metrics

<u>Total energy produced by residential-scale PV systems</u>: This metric allows evaluating how the proposed control approaches affect the PV generation (i.e., level of curtailment).

<u>Network energy losses</u>: This metric allows understanding how the proposed approaches affect the network's energy losses.

4.6.4 Control Actions

The total number of daily control actions required from all the proposed centralised control approaches is used as a metric to assess their performance. This metric is important as it allows understanding the extent to which the corresponding proposed control approaches are considered to be practical. Considering the decentralised control approaches, this metric is not quantified as control actions are performed locally and do not depend on any communication infrastructure.

In a centralised control approach, where communication infrastructure is required to perform the corresponding control actions, the fewer the daily control actions are, the better. A high number of control actions entails frequent use of the communication infrastructure which can highly influence its performance, leading to low and fluctuating bandwidths which in turn increases the time required to transfer data (i.e., latency) [143]. In addition, a high number of control actions can lead to the wear and tear of the corresponding control devices (e.g., tap changer, PV inverter).

Chapter 5

Case Study Part 2: Simulation Analyses for Residential-scale PV systems

5.1 Introduction

The control schemes for residential-scale PV systems, detailed in Chapter 3, are adopted on the real French LV network to assess their performance under different PV penetration levels during summer (i.e., worst case scenario; low demand with high generation). First, a stochastic analysis is performed in section 5.2 to assess the performance of the most common decentralised voltage control approaches (i.e., Volt-Var and Volt-Watt). Thereafter, the performance of the proposed centralised thermal control is investigated in section 5.3. The performance of the proposed combined centralised thermal and decentralised voltage control is presented in section 5.4.

All the minute-by-minute time-series power flow simulations are performed in OpenDSS [144] and driven (using the COM Server) by MATLAB where all the proposed control algorithms are implemented.

5.2 Decentralised Voltage Control

This section investigates the benefits of adopting decentralised voltage control using the Volt-Var and Volt-Watt control functions on the test LV network considering different penetration levels (i.e., 0 to 100%) of residential-scale PV systems for different days in summer (i.e., lowest demand days). The performance of the two control functions (i.e., Volt-Var and Volt-Watt) is investigated for 15 different control set-points (i.e., 15 different Volt-Var and Volt-Watt curves shown in Appendix A and B) adopting the Monte Carlo methodology described in section 4.5.

It is important to highlight that the following analyses assume that all PV inverters in the LV network follow the same curve as the aim is to find a single setting (practical to implement) that brings the most benefits.

5.2.1 Volt-Var Analysis

The 15 investigated Volt-Var curves were defined considering all possible combinations of "deadbands" and "activebands" in steps of 0.02 p.u. To explain the different combinations, db can range from 0.00 to 0.08 in steps of 0.02. For each one of these db values, the available combinations in ab values (also in steps of 0.02 p.u.) are investigated until the point where V1 and V4 are equal to the French statutory voltage limits. Here, db and ab are considered in steps of 0.02 in order to reduce the number of investigated scenarios and, therefore, reduce the total computational time of the analyses. Q1 and Q4 are set to 100% (injecting) and -100% (absorbing), respectively, while Q2 and Q3 are both set to 0%.

Considering the limited reactive capability of normal residential-scale PV inverters (discussed in section 2.3.1.4), the analysis performed in the following studies investigates both the use of normal and 10% over-rated PV inverters in an effort to understand to which extent the absorption/injection of reactive power can help to keep voltages within the statutory limits without curtailing any PV generation.



Figure 5-1 Reactive capability of 10% over-rated residential-scale PV inverter Reactive power capability of over-rated residential-scale PV inverters

To overcome the limitation of normal residential-scale PV inverters, the adoption of over-rated PV inverters can be considered as a potential solution. Adopting a 10% over-rated PV inverter (e.g., S=1.1p.u.) allows to absorb/inject up to 0.46 p.u of reactive power (Var) at peak generation (e.g., $\hat{P}_t = 1 p.u$.). The reactive capability

of a 10% over-rated PV inverter is demonstrated in Figure 5-1 considering 0.9 (leading or lagging) power factor. As the figure shows and compared to the case of a normal inverter (Figure 2-5), during periods of high generation (full irradiance/clear sky) the inverter is still able to absorb or inject reactive power.

5.2.1.1 Voltage Issues

In general, and as expected, without any control, customers experience voltage issues. The majority of voltage issues are caused due to voltage rise as a result of the reverse power flow caused by the PV generation. To demonstrate this and the corresponding performance when adopting a decentralised voltage control, Figure 5-2 shows the average (of 100 Monte Carlo simulations) number of EN 50160 non-compliant customers for each PV penetration level and for each investigated Volt-Var curve. It is demonstrated that without control (blue line with + markers), ~2 customers (in average) might start facing voltage problems at 30% of PV penetration level. As the PV penetration level increases, the number of EN 50160 non-compliant customers is also increasing, reaching the average number of 120 non-compliant customers at 100% of PV penetration.

In general, when the Volt-Var control is adopted, with either normal or 10% overrated inverters, the number of non-compliant customers decreases for each PV penetration level. Considering the adoption of normal inverters, Figure 5-2 (a), shows that a similar performance is achieved for all the investigated Volt-Var curves where the best performance (compared to the rest) is noticed when using curve 1 (red line with o markers) and 2 (pink line with * markers). This is explained as these curves do not have any "deadband" and they are more sensitive to voltage changes. Hence, all PV systems absorb more reactive power which results in the reduction of voltage level. These results highlight that the adoption of Volt-Var control with normal PV inverters is not very effective in managing voltages as the benefits are minimal due to the limited reactive capability of PV inverters at peak generations periods, where voltages are higher.



Figure 5-2 Voltage issues with Volt-Var control

On the other hand, adopting 10% over-rated PV inverters, a much better performance can be achieved (compared to the case with normal inverters) as the Volt-Var control is able to reduce significantly the number of non-compliant customers. More specific, all results in Figure 5-2 (b) show that the best performance is achieved with the most "sensitive" curves; the shorter the "deadband" and "activeband", the higher the absorption/injection of reactive power. For example, the application of any curve between 1 and 7 is able to keep all customers compliant with the EN 50160 up to 100% of PV penetration level.

5.2.1.2 Thermal Issues

Although the adoption of Volt-Var control allows reducing the number of EN 50160 non-compliant customers, Figure 5-3 and Figure 5-4 show that the utilisation level of the transformer and feeder two (i.e., longest with the highest number of connected customers) increases for all penetration levels above 20% for both normal and overrated inverters, compared with the case without control. In general, it can be stated that the shorter the *'deadband''* and *'activeband''*, the higher the utilisation level of the transformer. This is due to the fact that the transformer has to inject the amount of reactive power that is absorbed by the PV systems which in turn increases the current flowing into the network.

Active Management of PV-Rich Low Voltage Networks



Figure 5-3 Transformer utilisation level with Volt-Var control

Considering the adoption of normal inverters and taking the most '*sensitive*' curve (i.e., curve 1) as an example, Figure 5-3 (a) shows that the utilisation level of the transformer increases by ~10 and ~20% (compared to the no control case) at 70 and 100% of penetration levels, respectively. However, adopting 10% over-rated PV inverters, Figure 5-3 (b) shows that the utilisation level of the transformer is increasing by 30 and 40% more (compared to the no control case) at the same penetration levels (i.e., 70 and 100%, respectively). This is due to the fact that the over-rated PV inverters can absorb reactive power (i.e., higher current flowing in the network as the substation has to deliver reactive power) even at high generation periods where the utilisation level is already high.



Figure 5-4 Feeder utilisation level with Volt-Var control

5.2.1.3 Energy Metrics

Since the adoption of the Volt-Var control does not affect the active power of PV systems (i.e., no active power curtailment is performed), the total energy produced by the PV systems is provided only in the following sections (see Figure 5-8).

However, considering the increased current flow due to the reactive power absorption from the PV systems, the energy losses (in kWh) are expected to increase. Indeed, as Figure 5-5 (a) and (b) show, the energy losses resulting from the adoption of either normal or over-rated inverters increase for all PV penetration levels compared to the case without control. Taking, for example, the case of normal PV inverters with curve 1 (highest losses), losses increase almost by ~30% at each penetration level compared to the case without control. This goes to ~43% when adopting 10% over-rated PV inverters. This is because all PV systems absorb reactive power and, therefore, the transformer has to deliver more reactive power to the LV network. As a result, more current is flowing into the network and the $I^2 \times R$ losses increase. While similar performance is noticed for all the investigated curves, the effect is lower when the "deadband" of the curves increases.



Figure 5-5 Total energy losses with Volt-Var control

5.2.1.4 Summary

Normal rated inverters. The analysis demonstrates that the adoption of any of the 15 Volt-Var curves results in a similar, slight reduction of EN50160 non-compliant customers. Overall, voltage rise issues cannot be solved because the ability of the inverters to absorb reactive power is significantly limited during periods of high

generation (where voltages are high), i.e., the rated kVA power is almost reached. The utilisation level of the transformer and feeders, however, increases for all penetration levels as this control results in the absorption of reactive power which increases the current flowing in the network. In addition, this control approach results in slightly higher energy losses.

10% over-rated inverters. The analysis demonstrates that, in general, the application of Volt-Var control with over-rated inverters (10% more in this case) can provide a better voltage management (i.e., significant reduction of EN50160 non-compliant customers and, in some cases, shifting PV penetration level) without the need of curtailing any generation. This is due to the fact that an over-rated inverter is able to absorb reactive power even at peak generation periods. The best performance is achieved with the most "sensitive" curves; the shorter the "deadband" and "activeband", the higher the absorption/injection of reactive power. This control function, however, significantly increases the energy losses and the utilisation level of transformer and feeders.

5.2.2 Volt-Watt Analysis

Similarly to the previous section, the 15 investigated Volt-Watt curves were defined considering all possible combinations of "idlebands" and "activeband" in steps of 0.02 p.u. To explain the different combinations, V1 equals to Vref (1.0 p.u.) and *ib* can range from 0.00 to 0.08 in steps of 0.02. For each one of these *ib* values, the available combinations in *ab* values (also in steps of 0.02 p.u.) are investigated until the point where V3 is equal to the maximum French statutory voltage limit (1.1p.u.). Similar to the previous section, *ib* and *ab* are considered in steps of 0.02 in order to reduce the number of investigated scenarios and, therefore, reduce the total computational time of analyses. *P*1 and *P*2 are set to 100% and *P*3 and *P*4 to 0%.

5.2.2.1 Voltage Issues

Figure 5-6 shows the average (of 100 Monte Carlo simulations) number of EN 50160 non-compliant customers for each PV penetration level based on the applied Volt-Watt curve. The results show that the adoption of the PV inverter Volt-Watt control can eliminate all voltage issues in the LV network with almost any of the

investigated Volt-Watt curves. "Curve 15" results in ~5 non-compliant customers at 100% of PV penetration.



Figure 5-6 Number of EN 50160 non-compliant customers

Therefore, the adoption of the Volt-Watt control can effectively manage voltage issues in PV-rich LV networks. This control, however, comes at the expense of curtailing generation, which is discussed in the following sections.

5.2.2.2 Thermal Issues

As the Volt-Watt control is directly influencing the generation capability (active power -kW) of the PV systems, it is expected that the transformer and feeder utilisation levels will change upon the adoption of Volt-Watt control. Indeed, as Figure 5-7 (a) demonstrates, the transformer's utilisation level significantly reduces with any of the investigated curves. However, although curve 14 (first curve that solves all voltage problems) reduces the transformer's utilisation level below 100% up to 80% of penetration, it is not successful at 90 and 100% of penetration levels. If a curve with shorter "activeband" is used, say curve 13, then the transformer's utilisation level is always kept below 100%.

In terms of feeder's utilisation level, Figure 5-7 (b) which demonstrates the utilisation level of Feeder 2 (i.e., the longest with the highest number of connected customers) shows that all curves result in utilisation levels below 100% in comparison with the case where no control is adopted.



Figure 5-7 Transformer and feeder utilisation level with Volt-Watt

5.2.2.3 Energy Metrics

Energy Production. Figure 5-8 shows the daily average total energy (kWh) produced by the PV systems for each PV penetration level based on the applied Volt-Watt curves. The case where no control (blue line with + markers) is applied corresponds to the daily average maximum energy that the PV systems can produce at each penetration level. As expected, the higher the penetration, the higher the produced energy.



Figure 5-8 Total energy production from PV systems

When the Volt-Watt control is applied (with any of the Volt-Watt curves), generation curtailment occurs. Taking as an example the first curve that solve all voltage issues (curve 14), it can be seen that no energy is curtailed up to 30% of penetration level. However, with a penetration level of 40%, 3% of the energy needs to be curtailed. This increases to 5 and 23% at 50 and 100% of PV penetration levels, respectively. It

is important to mention that the shorter the "idleband" and "activeband", the higher the curtailment. This can be noticed with the adoption of curve 1 as most of the energy is curtailed due to the high voltage sensitivity of the controller.

Energy Losses (kWh). As the generation is curtailed due to the Volt-Watt control, losses are expected to be lower since less current is flowing in the network. Indeed, Figure 5-9 proves this concept showing that the energy losses are reduced significantly depending on the applied Volt-Watt curve. Of course, the higher the curtailment, the lower the losses, as shown in the results using curve 1. Considering the curve that solves all voltage problems, curve 14, the losses start reducing by 14% at 40% of penetration level. This figure goes up to 23 and 56% at 50 and 100% of penetration levels, respectively.



Figure 5-9 Total energy losses

5.2.2.4 Summary

The analysis demonstrates that this approach is more effective in managing voltages. Indeed, most of the curves show that the penetration level of PV systems can be potentially shifted to 100% without facing any voltage problems. The analysis also shows that a single Volt-Watt curve (i.e., curve 15) can be found that provides the best trade-off between voltage issues and the required volume of curtailment. In terms of the transformer and feeder utilisation levels, the analysis shows that the utilisation levels are always kept below 100%, except for two curves (i.e., 14 and 15) where assets are still overloaded at high penetration levels.

5.3 Centralised Thermal Control (CTC)

This section investigates the benefits of adopting the proposed centralised thermal control to manage thermal issues in PV-rich LV networks by limiting the generation capability of PV systems. First the time-series performance of the proposed method on the test LV network is presented. Its performance is then stochastically investigated considering different penetration levels of residential-scale PV systems for different days in summer (i.e., lowest demand days).

5.3.1 Time-Series Performance

In order to describe the time-series performance of the proposed centralised thermal control, a deterministic single case scenario is illustrated in this section adopting 1min control cycles. The performance is compared with the case where no control is applied. The test LV network considers 70% of PV penetration (i.e., 777kWp PV capacity) and the load profiles for each customer are randomly selected and allocated based on the procedure described in section 4.3.2 for the day 15 June 2014 (considered to have the lowest demand during summer). The PV systems are randomly allocated in the network based on the selected penetration level and assumed to be either 3kWp or 6kWp systems, sharing the same generation profile in June (summer).



Figure 5-10 Monitored power at transformer

Figure 5-10 (a) shows the daily 1 minute active power monitored at the transformer for the case without control. As it can be seen, the transformer is overloaded during 10am and 4pm. This is because the sun irradiance during this period is high and PV systems are generating at maximum. The same behaviour can be noticed at the head of each feeder, as shown in Figure 5-11 (active power monitored per phase at the

head of each feeder). More specifically, Feeder 2, which has the highest number of PV systems installed, faces thermal issues as the reverse power flow is higher than the capacity of the cable. It is also important to mention that the total produced energy from PV systems was 5.31MWh.



Figure 5-11 Monitored 3-phase power at the head of feeders - No control

In terms of voltage issues, the number of EN 50160 non-compliant customers is 112 and the 1-minute daily voltage profiles of all 162 customers in the network is shown in Figure 5-12 (a). Voltage rise issues can clearly be seen during the peak generation hours.

However, when the centralised thermal control is adopted, thermal issues are eliminated (see power monitored at the transformer and feeders in Figure 5-10 (b) and Figure 5-13, respectively); both transformer and feeders are always below their thermal limits. This is the result of using an active power limit that ensures the maximum possible PV generation without violating the thermal limits. The active power limit $pctP_{F_n}^{s.p.}$ for PV systems in Feeder 2, presented in Figure 5-14, shows that the generation limit is reduced to keep the transformer and feeder utilisation limits within the limits as opposed to the case without control.

Moreover, considering that the reduction of generation affects the voltage, it is also important to highlight that the number of EN 50160 non-compliant customers is reduced to 64. The 1-minute daily voltage profiles of all customers are shown in Figure 5-12 (b). As expected, the voltage level is reduced significantly for each customer.



Figure 5-12 Daily 1-minute voltage profiles of every customer



Figure 5-13 Monitored 3-phase power at the head of feeders - CTC, 1-min

In terms of energy produced in this specific case (one single day), the proposed solution resulted in the production of 4.54MWh of generation where the case without control resulted in 5.31MWh. Thus, the adoption of the proposed centralised thermal control was able to solve all thermal issues with the expense of curtailing 14.5% of

the total generation. Although curtailment of generation might be considered as reduced profit to the PV system owner, it is important as it allows accommodating even more LCTs in LV networks without the need of costly network reinforcements, thus accelerating the transition towards an eco-friendly low carbon environment.



Figure 5-14 Active power limits for PV systems – CTC, 1-min

5.3.2 Stochastic Analysis

In this section the proposed centralised thermal control is applied on the real French LV network considering different penetration levels of PV systems during summer (i.e., lowest demand days, highest generation). The performance of the proposed control logic is investigated by applying the Monte Carlo methodology as described in section 4.5 and adopting four different control cycles (i.e., 1, 10, 20 and 30-min). The simulation results shown in this section present and assess the benefits and disadvantages of adopting the proposed control method.

5.3.2.1 Voltage Issues

Figure 5-15 shows the performance of the proposed centralised thermal control in terms of the average number of customers (considering all Monte Carlo simulations) which are not compliant with the EN 50160 standard. The influence of different control cycles (i.e., 1, 10, 20 and 30-min) and the average standard deviation for each penetration level is presented with a different line colour.

Although the primary goal of the proposed control approach is to manage thermal issues, results show that, overall and considering all control cycles, the number of non-compliant customers reduces significantly from 50% of penetration level and onwards. This outcome occurs as a natural effect of the proposed control approach which instructs the PV systems to reduce their maximum generation point (i.e.,

curtailment) in order to manage thermal issues; thus, the customer's voltage level is reduced with the reduction of active power.



Figure 5-15 CTC - voltage issues

The best performance is achieved when adopting 1-min control cycles (red line) as the active power limit for the PV systems is calculated more frequently using the monitoring data of the last minute (higher accuracy). For example, taking into account all penetration levels above 60%, the number of customers facing voltage issues is on average 10 less than the number when adopting 10, 20 and 30-min control cycles. To provide more details, adopting 1-min control cycles and considering 70 and 100% of penetration levels, the number of customers facing voltage issues reduces from 90 and 120 (without control) to 50 and 60, respectively.

In general, it is highlighted that although the proposed centralised thermal control cannot shift the occurrence of voltage issues to a higher penetration level, it reduces the magnitude of voltage issues across all penetration levels above 50%.

5.3.2.2 Thermal Issues

In terms of thermal issues, Figure 5-16 (a) shows the transformer's average maximum utilisation level for the cases without (blue line) and with control (centralised thermal control) adopting four control cycles (i.e., 1, 10, 20 and 30-min).

Without control the transformer's maximum utilisation remains stable around 45% until 20% of PV penetration level and increases thereafter. The aforementioned behaviour is noticed when the local production is higher than the local consumption and the MV/LV transformer starts to export energy to the MV network. Thus, higher penetration levels produce more power to be exported and, therefore, result in a

higher utilisation of the electric infrastructure. The maximum utilisation level (100%) is reached in average after 50% of PV penetration, where the transformer is significantly overloaded.

To add more value to these results, Figure 5-16 (b) presents the percentage of time during a day that the transformer's utilisation level is above its kVA limit. For example, considering the case without control (blue bars), the transformer is overloaded for 6, 17 and 24% of the day at 60, 80 and 100% of PV penetration, respectively.

The feeders' utilisation presents a similar behaviour to that of the transformer without control. This is noticable in Figure 5-17 (especially for Feeders 2 and 3), which presents the average maximum utilisation level of the first segment of each feeder. The utilisation level of all feeders, except Feeders 2 and 3, remains below 100% for all penetration levels. For example, the utilisation level of Feeder 2 remains stable until 20% of PV penetration and increases rapidly from 30% onwards. The maximum utilisation level (100%) of Feeder 2 is reached in average after 40% of PV penetration. This shows that this Feeder will be overloaded earlier than the transformer (which might occur after 50% of PV penetration). The maximum utilisation level for Feeder 3 is reached in average after 90% of PV penetration.



Figure 5-16 CTC - (a) transformer utilisation and (b) % of day overloading

However, the adoption of the proposed centralised thermal control shows that with either control cycle is possible to significantly reduce the transformer's and feeders' maximum utilisation levels as illustrated in Figure 5-16 (a) and Figure 5-17, respectively. Considering the transformer and the penetration levels where the

transformer is overloaded without control (i.e., above 50%), the average maximum utilisation level (considering all Monte Carlo simulations) is always below 100% when adopting 1-min control cycles and below 108% for the other control cycles (i.e., 10, 20 and 30-min). The same effect is also noticed considering feeder utilisation levels and, particularly, in Feeder 2 which is the most loaded. It is important to mention that these figures represent the average maximum utilisation level recorded throughout the day and it does not mean that the assets are always operating at these levels.



Figure 5-17 CTC - Feeders' utilisation levels

The percentage of the time during the day that the transformer is overloaded (i.e., utilisation level above the kVA limit) is significantly decreased when adopting the centralised thermal control. For example, as observed in Figure 5-16 (b) and

Case Study Part 2: Simulation Analyses for Residential-scale PV Systems

considering 100% of PV penetration, the period within the day where the transformer is above its kVA limit decreased by almost 24, 20, 17 and 15% for the 1, 10, 20 and 30-min control cycles, respectively. Results also show that the longer is the control cycle, the more will be the period (within a day) where the utilisation of the assets might be above their kVA limit. This can be explained as specific active power limit set-points ($pctP_{F_{n_i}}^{s.p.}$), provided at the beginning of a control cycle, might not cater for sudden irradiance and demand changes until the next control cycle.

A possible solution to cater for this issue is to use a more conservative (lower) thermal threshold (i.e., a) for the centralised thermal controller, so as to curtail generation at lower utilisation levels. Similarly, another solution can be considered to reduce the duration of the control cycle when asset utilisation exceeds a second (higher) threshold.

In general, considering these results (i.e., maximum utilisation level of transformer and feeders) the proposed centralised thermal control proves to be effective in managing thermal issues in LV networks.

5.3.2.3 Energy Metrics

Figure 5-18 (a) shows the daily average total energy (kWh) produced by the all PV systems for each penetration level with and without the centralised thermal control. The case without control (blue line) shows that the higher the penetration, the higher the produced energy. A total of 3.8, 4.9 and 6.1 MWh of energy is produced at 60, 80 and 100% of PV penetration levels, respectively.

On the other hand, the adoption of the centralised thermal control, as expected, leads to the curtailment of energy generated from the PV systems and the same performance is achieved with any of the investigated control cycles. Considering the results, no energy is curtailed up to 50% of penetration level where at 60% of PV penetration, 2% of the energy is curtailed. This increases to 12 and 22% at 80 and 100% of PV penetration level, respectively.

In addition to the aforementioned, when adopting the centralised thermal control, energy losses are expected to reduce as less current is flowing in the network (due to generation curtailment). Indeed, this can be seen in Figure 5-18 (b) where energy



Figure 5-18 CTC - (a) energy produced and (b) network energy losses

losses are reduced significantly when energy (same performance in all investigated control cycles). To give an example, losses start reducing by 9.5% at 60% of penetration level. This figure goes up to 31.5 and 48.2% at 80 and 100% of penetration levels, respectively. However, it is important to highlight that the losses presented here correspond to the theoretical losses (e.g., assuming no network reinforcement, no limitation of PV system installations) where the power flow can be higher than the rated power of the feeder.

5.3.2.4 Control Actions

Figure 5-19 shows the average and standard deviation of the number of signals sent to PV systems within the day for each penetration level and control cycle, respectively.

As expected, results demonstrate that the length of control cycles is directly influencing the number of signals sent to the PV systems throughout the day. In general, the higher the penetration, the more the number of PV signals. Conversely, the longer the control cycle, the fewer the number of signals.

Considering the number of PV signals, the best performance is achieved with 30-min control cycles where the maximum number of PV signals sent throughout the day is around 40 (average) and significantly lower than the other control cycles (i.e., 160, 78 and 60 for 1, 10 and 20-min control cycles).



Figure 5-19 CTC - number of PV signals sent

5.3.3 Summary

The analysis demonstrates that the proposed centralised thermal controller is an effective solution to manage and solve thermal issues and slightly improve, indirectly, customer voltages. The maximum utilisation level of both transformer and feeders is significantly reduced to values close to the capacity limits; asset utilisation is always below 108% for all control cycles except 1-min where the utilisation never goes above 100%. The number of customers which experience voltage issues is significantly reduced as a natural result of reducing the maximum active power limit of PV systems. Although the adopted (and investigated) control cycles (i.e., 1, 10, 20 and 30-min) do not significantly influence the performance in terms of thermal and voltage issues, it is highlighted that the shorter the control cycle, the better the performance. On the contrary, shorter control cycles lead to higher number of control actions (i.e., PV signals in this case) which might cause operational challenges (i.e., communication). Hence, for this case study, the 30-min control cycle can be considered as a trade-off control setting between the number of control actions and operation performance (in terms of thermal and voltage issues).

5.4 Combined Centralised Thermal and Decentralised Voltage Control

As the last two sections (i.e., 5.2, 5.3) demonstrate, the adoption of the corresponding proposed control schemes, although being effective, they primarily address only one technical issue: either voltage or thermal. Here, the performance of the proposed combined centralised thermal and decentralised voltage control is assessed on the

real French LV network to provide a more complete active network management solution (i.e., management of both thermal and voltage issues).

The analysis is performed adopting the most adequate operational settings (i.e., Volt-Watt curve, control cycle) found in the previous analyses. Therefore, 30-min control cycles are adopted for the centralised thermal control and for the decentralised voltage control all PV inverters adopt the Volt-Watt curve 15 (Figure 5-20), which is identified as the most suitable to manage voltage issues while requiring the least generation curtailment (takes action only if the voltage, at the connection point, is higher than 1.08 p.u.).



Figure 5-20 Volt-Watt curve 15

The coordination between the centralised and decentralised controllers will naturally be achieved with the adopted control cycle (i.e., 30-min) where the decentralised voltage controller will have priority (i.e., acting first). More specifically, as voltage issues in PV-rich LV networks are usually occurring at an earlier penetration than thermal issues, the decentralised voltage controller which works in real time (e.g., fraction of a second) will be acting first while the centralised thermal controller will be acting (if needed) every 30 minutes. This provides coordination between the two controllers while giving priority to the decentralised one.

5.4.1 Time-Series Performance

In this section, the time-series performance of the proposed combined centralised thermal and decentralised voltage control is illustrated considering the same deterministic single case scenario used in section 5.3.1.

In general the time-series performance highlights that the proposed combined control approach is effectively managing both thermal and voltage issues. The power

Case Study Part 2: Simulation Analyses for Residential-scale PV Systems

monitored at the transformer (Figure 5-21) and feeders (Figure 5-22) is always kept below the corresponding capacity limits of the assets. In addition, it can be observed that the monitored power is slightly lower at peak generation periods compared to the case where only the centralised thermal control is applied (see Figure 5-10 (b) and Figure 5-13 in section 5.3.1). This is, in fact, an effect of the decentralised voltage control (i.e., Volt-Watt) which forces an additional reduction of the PV generation in order to keep voltages within the statutory limits.



Figure 5-21 Monitored power at transformer, CTC + Volt-Watt



Figure 5-22 Monitored 3-phase power at the head feeders – CTC + Volt-Watt

Indeed as Figure 5-23 illustrates, the voltage profiles of all customers in the test LV network, compared to the one where only the centralised thermal controller is applied (see Figure 5-12, section5.3.1), are always below the statutory voltage limits and, hence, all customers are compliant with the standard EN 50160.



Figure 5-23 Daily 1-minute voltage profiles of every customer, CTC + Volt-Watt

However, and as expected, to achieve this performance, an additional 4% of the total energy produced by the PV systems when the centralised thermal control is applied alone had to be curtailed. In total, considering the case without control, the proposed combined control method is able to solve all voltage and thermal issues by curtailing 17.87% of the total potential PV generation. As previously mentioned, although curtailment of generation might be considered as reduced profit to the PV system owner, it is important to highlight that it allows accommodating higher number of LCTs in LV networks without the need of costly network reinforcements, thus accelerating the transition towards an eco-friendly low carbon environment. As an alternative to this and considering the adoption of residential-scale BES systems, this amount of energy, otherwise curtailed, could be stored and used by the customer to support household demand when needed (more details in the following sections).

5.4.2 Stochastic Analysis

Same as in section 5.3.2, the performance of the proposed combined centralised thermal and decentralised voltage control is stochastically assessed (i.e., Monte Carlo methodology) on the test LV network considering different penetration levels of PV systems during summer (i.e., lowest demand days, highest generation).

5.4.2.1 Voltage Issues

As demonstrated in Figure 5-24, the proposed combined method found to be effective in eliminating all voltage issues, regardless the penetration level of PV systems. In contrast to the case where only the centralised thermal control is applied, the decentralised voltage control, as expected, provides additional flexibility in terms of voltage management. Overall, the network's hosting capacity (in terms of voltage issues) is shifted to 100% without facing any voltage issues when the combined centralised and decentralised control is adopted.



Figure 5-24 CTC + Volt-Watt - voltage issues

5.4.2.2 Thermal Issues

Furthermore, as demonstrated in Figure 5-25 (red line) and Figure 5-26 (right), the adoption of the combined centralised thermal and decentralised voltage control is also able to keep the utilisation level of both transformer and feeders below 100% for all penetration levels.

Nonetheless, it is important to highlight that with this approach, the utilisation levels of the assets are slightly lower compared to the centralised thermal control (see Figure 5-16). As previously discussed, this is due to the fact that the decentralised voltage control (i.e., Volt-Watt) results in additional PV generation reduction to bring voltages within the statutory limits. This effect can be noticed at lower penetrations (i.e., 40 and 50%) where the transformer's utilisation level, adopting the combined control method, is slightly lower than the one resulting from the case where only the centralised method is considered. Despite the fact that no asset is facing thermal issues at these penetration levels (i.e., 40 and 50%), customers are

experiencing voltage issues. Hence, the decentralised voltage control reduces the PV generation to manage voltages.

In general, these results (i.e., utilisation level of transformer and feeder) show that the combined centralised thermal and decentralised voltage control is able to provide an efficient thermal management in PV-rich LV networks.







Figure 5-26 CTC + Volt-Watt - Feeders' utilisation levels

5.4.2.3 Energy Metrics

Aligned with the previously discussed results, Figure 5-27 (a), which presents the total energy produced from PV systems, shows, as expected, that the combined control method results in slightly more energy curtailment in order to manage both voltage and thermal issues. It is important to mention that the additional energy curtailment is happening at the penetration levels where the centralised thermal controller is not acting or its actions are not enough to solve those issues (i.e., penetrations levels between 40-80%, see Figure 5-18). To be more specific, energy is not curtailed for up to 40% of penetration level where at 50%, 3.27% of the total

Case Study Part 2: Simulation Analyses for Residential-scale PV Systems

energy is curtailed. This figure increases to 16.32 and 25.47% at 80 and 100% of PV penetration level, respectively. Overall, the combined control method, compared to the case where only the centralised thermal control is adopted, results in an average of 3% more curtailed energy for each penetration level (above 40%).

The same performance is also observed considering the energy losses. As shown in Figure 5-27 (b), losses start reducing by 12.85% at 50% of penetration level. This goes up to 41.48 and 54.5% at 80 and 100% of penetration levels, respectively. Yet again, it is important to highlight that the losses presented here correspond to the theoretical ones (e.g., assuming no network reinforcement, no limitation of PV system installations) where the power flow is allowed to be higher than the rated power of the feeder.



Figure 5-27 CTC+Volt-Watt-(a) energy produced and (b) network energy losses

5.4.2.4 Control Actions

Finally, in terms of control actions per day, Figure 5-28 shows that the proposed combined control method achieves the same performance as the one when the centralised thermal control is applied alone (see Figure 5-19, CTC, 30-min). In general, no PV signals are sent up to 30% of penetration due to the fact that the utilisation of the assets (i.e., transformer and feeders) is always below their capacity limits. Thereafter, the number of signals increases with penetration; reaching the maximum of 40 (average) per day.



Figure 5-28 CTC+Volt-Watt – Number of PV signals

5.4.3 Summary

Compared to the control schemes investigated in sections 5.2 and 5.3, the adoption of the proposed combined thermal and decentralised voltage control provides a more robust active network management in PV-rich LV networks, as it effectively eliminates both voltage and thermal issues. However, to achieve this excellent performance, which allows shifting the network's hosting capacity to 100%, the analysis demonstrates that slightly more energy needs to be curtailed compared to the previously investigated control methods (i.e., average 3% more at each penetration level compared to the centralised thermal control alone).

These findings clearly indicate that the proposed combined centralised thermal and decentralised voltage control can be a potential solution to efficiently manage both thermal and voltage issues.

Chapter 6

Case Study Part 3: Simulation Analyses for Residential-scale BES Systems

6.1 Introduction

The proposed "advanced operation" mode for residential-scale BES systems, detailed in Chapter 3, is adopted here on the real French LV network to assess its performance under different PV penetration levels during summer (i.e., worst case scenario; low demand, high generation). For demonstration purposes its performance is compared against the "normal operation" mode adopted based on available specifications from BES system manufacturer (as detailed in section 3.3.1). A timeseries analysis is first performed on the two BES operation modes. Then, their performances are assessed adopting a stochastic analysis considering different PV penetration levels. This quantitative assessment allows understanding the benefits of adopting residential-scale BES systems in combination with PV systems in order to maximise the storage utilisation while managing voltage and thermal issues.

All the minute-by-minute time-series power flow simulations are performed in OpenDSS [144] and driven (using the COM Server) by MATLAB where all the proposed control algorithms are implemented.

6.2 Technical Specifications

The studies performed here consider the installation of a widely popular residentialscale BES system, currently available in the market. While there is a wide selection of lithium-ion batteries that could have been selected, the Tesla Powerwall [92] is becoming immensely popular, and paired with its low price (per kWh), it was considered as the most appropriate one to be modelled.

The basic technical specifications of the modelled BES provided by Tesla Motors are given in Table 6-1.

	Product Name	Tesla Powerwall
	Technology	Rechargeable lithium-ion battery
TESLA	Capacity	10kWh
	Power	3.3kW
	Efficiency	92.5% DC round-trip efficiency (includes the inverter)
	Depth of discharge	100%
	Compatibility	Single-phase and three-phase

 Table 6-1 Modelled BES System - Basic Specifications [92]

It is important to highlight that although Tesla Powerwall allows a full range of discharge depth (i.e., 100%) the studies in this Thesis adopt a 10% upper and lower battery reserve. These limits, as shown in Figure 6-1, mean that the BES system is able to charge up to 90% and discharge down to 10% of the available capacity. Adopting such limits helps maintaining the lifespan of the battery [136].



Figure 6-1 Modelled Charging and Discharging Depth

6.2.1 BES Sizing

After several discussions with the industrial partner, EDF R&D, the following approach to size BES systems was agreed as it creates realistic and challenging scenarios (important to assess the extent to which the proposed method is able to provide benefits). It is assumed that every household with a PV system installation is equipped with a BES system. The size of the BES systems allocated to a specific household is defined by the PV system's installed capacity. This is described below:

A basic³ PV system installation gets a basic BES system which is considered to have a 10kWh storage capacity with 3.3kWp power. For instance:

- For a household with one basic PV system (i.e., 1×3kWp capacity), one basic BES system (i.e., 1×10kWh with 1×3.3kWp) is allocated.
- For a household with two basic PV systems (i.e., 2×3kWp capacity), two basic BES systems (i.e., 2×10kWh with 2×3.3kWp) are allocated.

6.3 Additional Performance Metrics

This section contains the definition of additional metices that help assessing the performance of residential-scale BES systems from the perspective of both the grid and the customers. All metrices are assessed independently for each customer, i.e., the interactions between customers do not matter. For example, if "Customer A" exports energy into the LV network which is then used by "Customer B", the corresponding interaction is not quantified.

6.3.1 Grid Dependency Index (GDI)

The GDI quantifies how much of the total customer's energy consumption is provided by the LV network. As shown in (6.1), the GDI is defined as the ratio of daily energy imported (E_{import}^{house}) from the grid to the house over the daily household energy consumption ($E_{consumption}^{house}$). A value of 100% indicates that all the energy consumed by the house is provided by the grid, while a value of 0% indicates that all the energy consumed is self-supplied (excess PV generation stored by the BES system). This index helps understanding the impact on the electricity bill of a household. The lower the GDI, the less dependent is the household from the grid, thus resulting in lower bills.

$$GDI = \frac{E_{import}^{house}}{E_{consumption}^{house}}$$
(6.1)

³ As defined in section 4.4.2, a basic PV system is considered to have an installed capacity of 3kWp
6.3.2 LV Network Energy Imports

This metric quantifies the total daily energy imported from grid at the busbar of the secondary substation. This allows understanding how BES systems reduce the grid dependency of the network.

6.3.3 State of Charge (SOC)

The daily SOC profile is used as a metric to understand how the BES system is using its storage capacity throughout the day.

6.3.4 Adopted Advanced Operation Mode Settings

The analyses performed for the proposed advanced operation mode, time-series and stochastic, consider the following control settings (see section 3.3.2):

- $t_{peak} = 1 p. m.$; aligned with the sun irradiance used in this study
- $T_C = 10$ hours ; less than the average daylight hours (15 hours)
- $SOC_{final}^{C} = 0.9 (90\%)$; due to the upper reserve
- $SOC_{final}^{D} = 0.1 (10\%)$; due to the lower reserve
- $V_{limit}^{upper} = 1.1 p. u.$; French statutory limit
- a = 0.98 (98%); slightly conservative to cater for cloud effects

6.4 Time-Series Performance

This section presents the time-series performance and the effects of residential-scale BES systems using the "normal" and proposed "advanced" operation mode. The analysis is performed for three consecutive sunny days on the real French LV network considering 70% of PV penetration. Load profiles for each customer are randomly selected and allocated based on the procedure described in section 4.3.2 for the days 15, 16 and 17 June 2014 (considered to have the lowest demand during summer). The PV systems are randomly allocated in the network based on the selected penetration level and assumed to be either 3kWp or 6kWp systems, sharing the same generation profile which considers June. For each 3kWp or 6kWp PV

system a 10kWh or 20kWh BES system is allocated (as detailed in 6.2.1), respectively.

6.4.1 Normal Operation Mode

Figure 6-2 shows the transformer's utilisation level for the case (a) without and (b) with BES systems. Results clearly show that the adoption of BES systems using the normal operation mode does not bring any significant benefit in terms of reducing the transformer's utilisation level. Although the utilisation reduces significantly during early hours of the first day, BES systems fail to do that on the following days.



(b) with BES systems (normal operation)

Figure 6-2 Monitored power at the transformer

This is caused due to the two limitations stated previously (section 3.3.1.3) as most of the BES systems do not fully discharge overnight. As a result, the BES systems achieve full SOC very early in the day and, hence, are unable to reduce the reverse power flowing back to the substation.

Indeed, Figure 6-3, which shows the individual SOC of each household, validates the aforementioned statement as almost all BES systems achieve full SOC (90%) around midday (i.e., 12pm). Once the BES systems start discharging, the SOC reduces

Case Study Part 3: Simulation Analyses for Residential-scale BES Systems

differently for each household as it depends on the corresponding household demand. The majority of the BES systems never completely discharge and, therefore, a significant volume of energy is still available in the batteries by the time the new charging period (following day) begins. This, forces the BES systems to reach full SOC even earlier the day after. The mean behaviour of the SOC, considering all BES systems and represented with a thick dashed black line in Figure 6-3, demonstrates that the SOC is varying only from 90% to 60%. This highlights the inability of this operation mode to fully utilise the BES system capacity.



Figure 6-3 Individual SOC Profiles for all BES systems (normal operation)

More importantly, the performance described above also limits the ability of the BES systems to reduce potential voltage issues during midday where generation is at peak. To demonstrate this, Figure 6-4 presents the daily voltage profiles of all customers in the network for the case (a) without and (b) with residential-scale BES systems. Figure 6-4 (a), as expected, shows that most of the voltage issues occur during the peak generation period (i.e., 10am to 5pm). When BES systems are in place, Figure 6-4 (b) shows that during the first day, where BES systems are empty, it is possible to flatten the profiles (BES systems store all the excess of PV generation) until ~2pm where all BES systems reach full SOC. Crucially, due to the fact that full SOC is achieved very early, voltage profiles cannot be reduced during the peak generation period. This issue becomes even more critical the following days as the majority of BES systems start the next charging cycle with an average SOC of 60%. Hence, as Figure 6-4 (b) shows, the reduction of voltage profiles is insignificant (see 2nd and 3rd day around 9am to 12pm).





(b) with BES systems (normal operation)

Figure 6-4 Daily 1-minute voltage profiles of every customer

Although the aforementioned results highlight the drawbacks of the normal operation mode to assist in voltage and thermal management (i.e., DNO benefits), the results show that the customers (i.e., individual households) can reduce their grid dependency and, hence, their corresponding electricity bill. To visualise this, Figure 6-5 shows the GDI of all customers (boxplot representation) with PV system installations for the case (a) without and (b) with residential-scale BES systems. Considering the case where BES systems are not available, Figure 6-5 (a) shows that the median number of customers GDI is around 50% which means that almost 50% of their energy consumption is supplied by the substation and the other 50% by the local PV generation. However, considering the case with BES systems, Figure 6-5 (b) demonstrates that the GDI can reduce even more as additional energy can be supplied locally by the BES systems. For this particular analysis, it was found that the customers' GDI is reduced down to $\sim 10\%$ except the first day which is reduced slightly (~30%) as BES systems are empty during the early hours; hence, cannot supply energy. In summary, this specific example shows that during summer, the use of BES systems with normal operation can potentially help reducing customer electricity bills.



Figure 6-5 GDI

Lastly, in terms of energy, Table 6-2 shows, for each day, the total energy imported in the LV network for the case without and with residential-scale BES systems, respectively. As expected, when BES systems are available in the network (excluding the first day were BES systems start empty), the energy imports are significantly reduced compared to the case without BES systems; most energy is now supplied locally.

 Table 6-2 Network Energy Imports – BES Normal Operation

	Day 1	Day 2	Day 3	
Without BES systems	864 kWh	799 kWh	871 kWh	
With BES systems (Normal Operation)	721 kWh	229 kWh	302 kWh	

6.4.2 Advanced Operation Mode

Using the same scenario as in the previous section, the time-series performance of the proposed advanced operation mode for residential-scale BES systems is investigated considering the real French LV network for the same three consecutive summer days with 70% of PV penetration level.





Figure 6-6 Transformer utilisation with BES systems (advanced operation)

Figure 6-6 shows that the proposed advanced operation mode significantly reduces the utilisation level of the transformer compared to the case with normal operation (see Figure 6-2). Although the advanced operation mode is not directly managing thermal issues, the nature of the progressive charging overcomes the limitation of reaching full SOC very early in the day and then allows charging throughout the whole period. Consequently, this method is able to significantly reduce the reverse power flows (due to the PV generation) and, therefore, keep the transformer's utilisation well below its thermal limit (i.e., 400kVA).



Figure 6-7 Individual SOC Profiles for all BES systems (advanced method)

Figure 6-7, which shows the individual daily SOC for each BES system in the investigated network, demonstrates the effectiveness of the proposed advanced operation mode to fully charge and discharge all BES systems by the end of each period (i.e., charging and discharging). In average, and compared to the results in section 6.4.1, the SOC profile (black dashed line) shows that the proposed advanced operation mode is able to fully utilise the whole storage capacity (from 10 to 90%) while having the same performance every day.

It is important to mention that some BES systems (e.g., cyan colour - indicated with arrow) become full earlier than the end of the charging period (i.e., 6pm). This is, in

Case Study Part 3: Simulation Analyses for Residential-scale BES Systems

fact, the result of the voltage control, which is forcing the specific BES systems to charge with higher power rate in order to reduce the voltage at the corresponding connection point.

Similar behaviour can also be noticed during the discharging period where several BES systems become empty before the end of the period (i.e., 8am). This is because these systems are supplying customers with very large demand profiles and, therefore, the discharging power is very high, causing a fast decline of the SOC.



Figure 6-8 Customer Voltages with BES systems (advanced method)

In terms of voltage issues, results shown in Figure 6-8 demonstrate the effectiveness of the proposed advanced operation mode to keep all customer voltages below 1.1p.u (i.e., upper French statutory limit - black dashed line). The ability of the advanced operation to progressively charge throughout the whole charging period allows having enough storage capability by midday where most of the voltage issues occur (high irradiance) and, therefore, BES systems are able to increase their charging power rate to reduce voltage profiles. Additionally, it can also be noticed that the voltage profiles also become more flat even during the evening as power is locally supplied.

More importantly, from the customer side, Figure 6-9 demonstrates that the proposed advanced operation mode is significantly reducing customer's GDI which means that customers are becoming less dependent to the LV network as most of their demand is now supplied locally by the BES systems. Considering the median values (excluding the first day), the GDI is reducing down to ~2% (median) showing that, in general, most of the customers have a very low dependency to the grid which means significantly reduced bills.

Active Management of PV-Rich Low Voltage Networks



Figure 6-9 GDI with BES systems (advanced operation) Table 6-3 Network Energy Imports and Exports – BES Advanced Operation

-	Day 1	Day 2	Day 3
With BES Systems (Advanced Operation)	486 kWh	26 kWh	6 kWh

As a result of the above (low GDI), less energy is now required to be imported in the LV network by the secondary substation. Indeed, Table 6-3 shows that the energy imported to the LV network is significantly lower compared to the cases investigated in section 6.4.1. For example, only 26 and 6 kWh of energy is imported during the second and third day, respectively.

6.5 Stochastic Analysis

In this section, the proposed advanced operation mode for residential-scale BES systems is stochastically investigated on the real French LV network considering different penetration levels of PV systems during summer (i.e., lowest demand days, highest generation). The performance of the proposed mode is compared with the normal operation mode and with the case without residential-scale BES systems (only PV systems) adopting the Monte Carlo methodology described in section 4.5. This analysis presents and assesses the benefits obtained for both customers (i.e., increasing self-consumption, GDI) and DNOs (i.e., voltage and thermal management) when adopting the proposed method.

6.5.1 Voltage Issues

Similarly to the previous analyses, Figure 6-10 shows that without residential-scale BES systems (i.e., no control case in previous analyses), an average of ~10 customers might start facing voltage problems (i.e., EN 50160 non-compliant

customers) at 40% of PV penetration, where this number increases to 120 at 100% of penetration.



Figure 6-10 BES Systems - Voltage issues

When residential-scale BES systems are adopted with normal operation mode, is observed that, although slightly less customers experience voltage issues at each penetration level, the performance remains almost the same as the case without BES systems. This, which is also highlighted in the time-series performance (section 6.4.1) and discussed in 3.3.1.3, is due to the limitations of the normal operation mode where the BES systems reach full SOC very early (at or before midday); hence, voltage profiles cannot be reduced during the peak generation period.

However, a significant performance improvement is, however, achieved when residential-scale BES systems are adopted with the proposed advanced operation mode as almost no customer is affected by voltage issues until very high PV penetration levels (i.e., 60% onwards). Even at these high penetration levels, the number of EN 50160 non-compliant customers is significantly lower compared to the case without residential-scale BES systems. For example, in average only 10 and 35 customers are non-compliant at 70 and 80% of PV penetration levels, whereas at 90 and 100% penetrations the number is reducing from 110 and 120 down to 60 and 75, respectively.

Although these results demonstrate that a significant voltage management improvement is achieved with the proposed method, PV generation curtailment is still required at high PV penetration levels. This is due to the fact that at high penetration levels of PV voltage issues occur earlier during the morning (simultaneous generation of more PV systems) and BES systems are forced to charge

at higher power rates; hence, become full before the end of the charging period and not able to reduce voltages after that.

Although the adoption of such high PV penetrations (i.e., above 50%) is not expected in the short term, these results highlight that residential-scale BES systems along with the proposed advanced operation mode can be an effective solution to manage voltage issues in LV networks while significantly increasing their hosting capacity.

6.5.2 Thermal Issues

Figure 6-11 and in Figure 6-12 show the transformer's and feeders' maximum utilisation level without (blue line) and with residential-scale BES systems adopting the normal (red line) and proposed advanced operation mode (magenta line).

As expected and previously demonstrated (time-series analysis, section 6.4.1), the stochastic analysis of the normal operation mode of residential-scale BES systems proves and highlights that the corresponding mode does not provide significant benefit in terms of thermal management. Although Figure 6-11 and Figure 6-12 show that the maximum utilisation of the assets slightly reduces with the corresponding operation mode, the reduction is not enough (about 5% at each penetration) to increase the network's hosting capacity.

On the other hand, the adoption of the residential-scale BES systems with the proposed advanced operation mode is able to significantly reduce the transformer's and feeders' maximum utilisation levels. As illustrated in Figure 6-11 the transformer's average maximum utilisation level (considering all Monte Carlo simulations) is always below 100% (except at 100% of PV penetration level where the transformer is slightly overloaded). The same effect is also noticed considering feeder utilisation levels (Figure 6-12) and, in particular, Feeder 2 which is the most loaded feeder. It is important to mention that these figures represent the average maximum utilisation level recorded throughout the day and it does not mean that the assets are always operating at these levels.

Although the proposed advanced operation mode is not directly managing thermal issues, results show that the reduction of reverse power flows, because of the BES charging period, is able to provide a satisfactory solution in terms of reduced thermal

issues in LV networks and, hence, significantly increase the hosting capacity of LV networks.



Figure 6-11 BES Systems - Transformer maximum utilisation level



Figure 6-12 BES Systems - Feeders' maximum utilisation levels

6.5.3 Energy Metrics

In terms of the total energy produced by the residential-scale PV systems, Figure 6-13, which presents the average daily total active energy (kWh) produced by the residential-scale PV systems for each penetration level, highlights, as expected, that the adoption to BES systems (with either operation mode) does not result in any curtailment. The same performance (overlapped lines) is observed in all cases (i.e., without and with BES).

Clearly, these results, compared to the previous analyses, highlight that the adoption of residential-scale BES systems along with the proposed advanced operation mode can help increase the energy harvested by the PV generation while reducing the voltage and thermal issues; hence, increasing the hosting capacity of LV networks.

Regarding the energy losses, which can be seen in Figure 6-14, it is highlighted that the adoption of BES systems with either operation mode can help in reducing the

network's energy losses. This can be explained as the result of the reduce power flows in the LV network due to the fact that BES systems are charging some of the excess of the PV generation. In addition, demand is now supplied locally instead of energy flowing from the substation to the customers.



In general, it is observed that the proposed advanced operation mode results in significantly lower energy losses compared to the normal operation mode. For example, the energy losses at each penetration level reduce by an average of ~20% when adopting the normal operation mode, while the adoption of the advanced operation mode results in a reduction of an average of ~65%. This is because the former (normal operation) fails to reduce the reverse power flow throughout the whole PV generation period (i.e., limitation of which BES systems reach full SOC before midday). On the other hand, the proposed advanced operation mode allows charging throughout the whole generation period and, hence, significantly reduces the amount of current flowing back to the substation.

6.5.4 Energy Imported

Figure 6-15 shows the average daily total energy imported to the LV network (at the MV/LV substation) considering the cases without and with residential-scale BES systems adopting the normal and advanced operation mode, respectively.

Interestingly, results show that without BES systems the amount of energy imported in the LV network is reducing almost with PV penetration level as a significant amount of energy is supplied by the PV generation locally. For example, the imported energy from 50% of PV penetration onwards almost reduces by 50% Case Study Part 3: Simulation Analyses for Residential-scale BES Systems (compared to 0% of penetration). However, the energy supplied, in such case, is limited only to the period when PV systems are generating and, hence, are able to supply the corresponding demand only during that period. Given that an important incentive to install PV systems is to increase the production and use of "green energy", these results highlight that still almost 50% of the LV network demand has to be supplied by the substation. No matter the level of PV penetration as the "green energy" from PV systems can supply the corresponding demand only during the generation hours.



Figure 6-15 Energy Imported

However, the adoption of residential-scale BES systems, with either operation mode, significantly reduces the amount of energy imported by the LV network as they are able to store a significant amount of unused "*green energy*" and supply the household demand during the night where PV systems are not generating. It can also be highlighted that the imported energy significantly decreases with penetration levels. With higher PV penetrations the capacity of BES systems also increases (more 20kWh BES systems) and, hence, are able to supply a larger (or all) amount of the corresponding household demand.

Considering the normal operation mode, the imported energy reduces (compared to 0% of PV penetration level) by almost 14% at 20% of PV penetration and by 27 and 38% at 30 and 40% of penetration, respectively. This reduction continues to increase with penetration levels where at 100% the total energy imported reduces to almost 84% (compared to the case without BES systems).

Regarding the adoption of the proposed advanced operation mode, the total energy imported reduces even further as the BES systems always discharge based on the discharging power rate and, therefore, nearby households can also be supplied locally; hence, less energy is required from the substation. Indeed, as observed in Figure 6-15, by adopting the advanced operation mode, the imported energy at each penetration level reduces even more compared to the normal operation mode. For example, the imported energy reduces by almost 37% at 20% of PV penetration and by 62 and 80% at 30 and 40% of penetration, respectively. From 50% onwards, the energy imported almost reduces by 97% which means that almost all the demand in LV network is supplied locally by both the residential-scale PV and BES systems.

These results also highlight that there is a need to revise the way customers are charged in the future as the adoption of residential-scale BES systems will significantly reduce the energy imported and, therefore, reduce the income of DNOs (largely based on consumption).

6.5.4.1 Grid Dependence Index (GDI)

Considering the results shown in the last section, 6.5.4, where the imported energy is reducing for both cases without and with residential-scale BES systems (with either operation mode), it is expected that the households' GDI will decrease significantly as a significant volume of energy is supplied locally (i.e., PV generation, BES systems or both); hence, households can be less dependent from the substation. To explore this, Figure 6-16 statistically represents (boxplots) the average GDI of customers with PV system (with and without BES system) considering all Monte Carlo simulations and all investigated PV penetration levels.



Figure 6-16 GDI

Indeed, considering the case without residential-scale BES systems, results are aligned with the findings in Figure 6-15, showing that almost half of the household

Case Study Part 3: Simulation Analyses for Residential-scale BES Systems

demand is supplied locally by the PV generation. For example, the GDI is in average close to 50% for all penetration levels meaning that households with PV systems are only 50% dependent by the substation (i.e., their electricity bill can be assumed 50% less compared to the case without PV systems).

More importantly, as results demonstrate, households' GDI can be reduced even more when adopting residential-scale BES systems with either operation mode. For example, adopting the normal operation mode, the median GDI drops down to 15% at 10% of PV penetration level and this continues to reduce as penetration increases reaching 7% at 100% of penetration.

Nonetheless, the adoption of the proposed advanced operation mode shows that the GDI can reduce by 5% more at each penetration level. Considering this, the median value of the GDI drops down to 10% when 10% of PV penetration is assumed and this number reduces even more with higher penetration levels as BES systems with higher capacities (i.e., 20kWh) are considered. For example, the median GDI drops to 9% with 50% of PV penetration and continues to reduce to 5% and 2% at 70 and 100% of penetration, respectively.

Given the analysis above, it is clear that the adoption of residential-scale BES systems with the proposed advanced operation can significantly reduce customers' dependence on the grid and, therefore, their corresponding electricity bills.

6.5.5 State of Charge (SOC)

Figure 6-17 shows the average daily SOC profile of the residential-scale BES systems while adopting the (a) normal and (b) proposed advanced operation mode. The different line colours represent the daily average profile of each penetration level (from 10 to 100%).

As expected and previously highlighted (see section 6.4), results shown in Figure 6-17 (a) demonstrate the inability of the normal operation to fully utilise the whole storage capacity of the BES system as its average SOC is varying only from 50 to 90% (i.e., storage capability). This effect can be explained as the normal operation mode fails to fully discharge the BES systems until the next charging period. Consequently, the BES systems start to charge with a significant amount of energy already stored in their batteries. It is also important to mention that the storage

capability reduces with higher penetration levels as bigger BES systems (20kWh) are installed, which might be significantly higher than the household's energy needs. Thus, a higher amount of energy remains in the battery at the end of the discharging cycle which forces the system to reach full SOC even earlier in the following day.



Figure 6-17 SOC

On the other hand, the adoption of the proposed advanced operation mode allows utilising the whole capacity (from 10 to 90% of SOC) of the residential-scale BES systems achieving almost the same excellent performance in all penetration levels. At the beginning of the charging period (i.e., 8am), the SOC is as required, equal to 10%, which allows for a charging headroom up to 90%. This provides the benefit to always start charging from the lowest available SOC point; hence reducing the risk of achieving full SOC very early during in day (one of the limitations identified in section 3.3.1.3). During the charging period, the SOC linearly increases and the BES system becomes full (90% of SOC) at the specified end of the charging period (i.e., 6pm). This proves that the proposed advanced operation mode can make use of the full BES system capacity and always store the maximum possible energy of the otherwise unused PV generation.

Once the discharging period begins (6pm), the opposite behaviour is noticed. The SOC linearly decreases until the end of the specified discharging period (i.e., 8am). Results show that the proposed method is able to fully discharge the BES systems (one of the drawbacks identified in section 3.3.1.3 allowing for a full charging headroom (i.e., 10 to 90%) in the next day.

Case Study Part 3: Simulation Analyses for Residential-scale BES Systems Overall, results show that the proposed advanced operation mode can achieve an excellent performance in terms of the storage utilisation for all penetration levels.

6.6 Summary

The analysis performed in this chapter provides a quantitative assessment of the benefits when adopting residential-scale BES systems in combination with residential-scale PV systems in order to maximise customer benefits while managing voltage and thermal issues.

It is demonstrated that the adoption of residential-scale BES systems with a normal operation mode (control mode adopted by manufacturers) although provides significant benefits to the customers (i.e., reduced dependence to the grid) it does not provide any benefit to the DNO (i.e., voltage or thermal management). In addition the analysis highlighted the already mentioned important limitations (see section 3.3.1.3) of the normal operation mode where the:

- BES systems achieve full SOC before peak generation (i.e., PV impacts still occur around noon); and,
- BES systems start charging the excess of PV generation (early daylight) with a SOC already above the lower reserve (i.e., not adequately discharged overnight)

On the other hand, the proposed advanced operation mode was found to overcome the aforementioned limitations and, while still beneficial to the customers (i.e., reduced dependence to the grid, hence reduced electricity bills), it also allows managing voltage and thermal issues. The analysis demonstrates that the proposed operation mode can effectively eliminate all voltage issues up to high PV penetration levels (i.e., 60%) beyond which it can significantly reduce them. Although the proposed advanced operation mode does not directly manage thermal issues, it is proved that the reduction of reverse power flows, due to the charging period, is able to provide a satisfactory solution in terms of thermal issues in the corresponding test LV network.

Considering the customer benefits, the proposed advanced operation mode significantly reduces their dependence to the grid while still benefiting from the quality of service brought by the grid (e.g., supply always available). In general,

Active Management of PV-Rich Low Voltage Networks

without considering any capital and operational expenditures required for the uptake of residential-scale PV and BES systems, results mean that customers will potentially pay less electricity bills compared to the case where BES systems are not adopted.

In general, this analysis clearly demonstrates that the adoption of residential-scale BES systems with the proposed advanced operation mode can significantly help increasing the harvesting of PV generation (otherwise curtailed) while providing benefits to both customers and the DNO.

Chapter 7

Case Study Part 4: Simulation Analyses for LV OLTC-fitted Transformers

7.1 Introduction

This chapter investigates and presents a quantitative assessment of the benefits when adopting OLTC-fitted transformers with the proposed control logic in order to manage voltage issues in PV-rich LV networks. Its performance is assessed adopting two different levels of monitoring considering location (i.e., middle and end point of feeders) and four different control cycles (i.e., 1, 10, 20, 30-min). First the OLTC-fitted transformer adopted for the corresponding analyses is presented. Then, the time-series performance of the proposed OLTC control logic is assessed using the real French LV network. Lastly, its performance is stochastically investigated considering different penetration levels of residential-scale PV systems for different days in summer (i.e., lowest demand days), levels of monitoring and control cycles. Considering that the adoption of LV OLTC-fitted transformers only affects voltages (i.e., power flows remain the same), the following analyses discuss the performance of the proposed OLTC control logic is assessed adopting the proposed OLTC control logic only in terms of voltage issues and control actions (i.e., tap changes).

All the minute-by-minute time-series power flow simulations are performed in OpenDSS [144] and driven (using the COM Server) by MATLAB where all the proposed control algorithms are implemented.

7.2 Modelled LV OLTC-fitted Transformer

An LV OLTC-fitted transformer, is adopted to perform the corresponding analyses in order to assess the performance of the proposed OLTC control logic that aims in managing voltages in PV-rich LV networks. The OLTC considers a range of +/-8% of voltage with 2% per tap, i.e., 9 tap positions in total, according to available manufacturer specifications [145]. Assuming that the voltage at the primary of the

MV/LV transformer is the nominal line-to-line voltage (i.e., 20,000V), the busbar voltages corresponding to different tap positions are shown in Table 7-1.

It should be noted that the specific transformation ratio is selected in order to provide the flexibility to have tap positions that can provide the maximum possible range of voltage level within the voltage statutory limits (+/-10% of nominal voltage, in France).

Transformer Tap position	MV	LV		Vbase = 400V
	L-L (V)	L-L (V)	L-N (V)	V (p.u)
9 (+8%)	20000	368	212.5	0.92
8 (+6%)	20000	376	217.1	0.94
7 (+4%)	20000	384	221.7	0.96
6 (+2%)	20000	392	226.3	0.98
5	20000	400	230.9	1
4 (-2%)	20000	408	235.6	1.02
3 (-4%)	20000	416	240.2	1.04
2 (-6%)	20000	424	244.8	1.06
1 (-8%)	20000	432	249.4	1.08

 Table 7-1 Voltage regulation of the modelled LV OLTC-fitted transformer

For example, with the corresponding OLTC setup and assuming that the voltage at the primary of the transformer is at nominal value (20kV) and there is no loading, the lowest line-to-neutral voltage at busbar would be around 212.5V (tap position 9) and the highest line-to-neutral voltage at busbar would be 249.4V (tap position 1).

The embedded OLTC model in OpenDSS allows the control of the device by providing a new voltage target at every control cycle. This voltage target, either to manage the busbar or a remote point, is then automatically translated into an internal tap control command to achieve the desired value. In the proposed control logic (section 3.4.2), the required tap changes are carried out in the next time step (e.g., the next minute, assuming 1-min control cycle).

It is also important to highlight that the corresponding OLTC bandwidth is set to 2.5% (+/-1.25%), which means that a new tap position can be triggered if the monitored phase voltage at the busbar is higher or lower than 1.25% of the set-point voltage (voltage target). As a default setting, the OLTC model in OpenDSS monitors the busbar voltage in phase A.

7.3 Time-series Performance

Considering the same deterministic single case scenario used in section 5.3.1, the time-series performance of the proposed OLTC control logic is illustrated in this section adopting remote monitoring at the end of each feeder (considered to be the location of the farthest customer) and two different control cycles (i.e., 1 and 10-min).

Figure 7-1, which presents all customer voltage profiles for the case (a) without and with (b) 1 and (c) 10-min OLTC control, illustrates the performance of the proposed control logic. Considering either control cycles, it can be observed that during the period where voltages rise (around 8am), the tap position of the OLTC, which is shown in Figure 7-2, increases to reduce the busbar voltage and, hence, bring the monitored voltages within the statutory limits. Indeed, the analysis shows that the proposed OLTC control logic with either control cycles is able to bring and keep all customer voltages within the French statutory voltage limits $(0.9 p. u. \le V \le$ 1.1 p. u.). Similarly, when the PV generation starts to reduce, and hence the voltages (around 5pm), the OLTC control logic is increasing the voltage target which so that the remote end voltage will be will not violate the lower limits. Indeed as observed in Figure 7-2, the tap position starts to reduce from around 5pm in order to increase the voltages. When the sun settles down (no PV generation) tap position four is eventually reached (see Figure 7-2, b), which corresponds to 1.02 p.u. of voltage target; the middle point of voltage statutory limits and also the V_{ref} of the OLTC controller.

Results also point out that the overall performance is influenced by the control cycle length. For example, the adoption of 1-min control cycle, although achieves the desired goal (i.e., voltage management), it results in a significantly high number of tap changes (i.e., 204 for this case) as an action is triggered almost every minute due to the intermittent behaviour of the demand and PV generation (e.g., cloud effects). This however can lead to the OLTC wear and tear [124] and, as a consequence, increase maintenance costs. Nonetheless, increasing the control cycle to 10 minutes, the performance in terms of voltages remains the same (as the 1-min control cycle) while significantly reducing the number of tap changes (i.e., 20 for this case, Figure 7-2, b). In general, as the control cycle increases, control actions (i.e., tap changes)

are expected to reduce due to the fact that the controller uses the average monitored voltages of the last x minutes (10 minutes in this case) which in turn smooths the intermittent behaviour of the PV generation (e.g., sudden voltage spikes or drops due to cloud effects).



Figure 7-1 Daily 1-min voltage profiles of customers, OLTC control logic



Figure 7-2 Tap positions

7.4 Stochastic Analysis

Following the time-series performance, the proposed OLTC control logic is stochastically investigated on the real French LV network considering different penetration levels of residential-scale PV systems during summer. Its performance is assessed adopting two different levels of remote monitoring (i.e., middle and end point of each feeder) and four different control cycles (i.e., 1, 10, 20, 30-min) by applying the Monte Carlo methodology described in section 4.5.

7.4.1 Voltage Issues

The analysis highlights that the proposed OLTC control logic can achieve an excellent performance in terms of voltage management. In general adopting the proposed OLTC control logic with any of the adopted remote monitoring points (i.e., middle or end) and control cycles (i.e., 1, 10, 20, and 30 minutes), allows shifting the network's hosting capacity to 50% of penetration without experiencing any issues.



From that point (i.e., 50% of penetration) forward, the performance is slightly affected by the corresponding adopted control cycle length (i.e., 1, 10, 20 and 30min) and the level of remote monitoring (i.e., middle and end point of each feeder). For example, considering the end monitoring, it is observed that the shorter the control cycle length, the better the voltage performance. More specifically, Figure 7-3 (a) shows that at 100% of PV penetration, adopting 1-min control cycles with remote monitoring at the end of each feeder, the number of EN 50160 non-compliant customers reduces from an average of almost 120 to 10. This number, however, *Case Study Part 4: Simulation Analyses for LV OLTC-fitted Transformers* increases to an average of 25 customers when adopting 10-min control cycles and goes to 30 customers with 20 and 30-min control cycles.

Taking into account the level of remote monitoring, Figure 7-3 demonstrates that the control logic performs better with remote monitoring at the end of each feeder. This is due to the increased visibility (closer to the most critical customers - those connected at the far ends) which in turn allows a more accurate calculation of the required voltage target. For instance, at 100% of PV penetration the case with monitoring only at the middle of each feeder resulted in an average of 15 non-compliant customers more than the case with end monitoring. Hence, the end monitoring can be considered as the most effective approach to support the operation of the OLTC.

Overall, the analysis highlights that the proposed OLTC control logic allows increasing the hosting capacity of the LV network whilst also reducing the magnitude of voltage problems across all penetration levels.

7.4.2 Control Actions

Figure 7-4 (a) and (b) show the average number of daily tap changes triggered for each control cycle (i.e., 1, 10, 20, 30-min) and PV penetration level considering remote monitoring at the end and middle of each feeder, respectively. In general, the level of monitoring does not influence the performance as, in average, the same number of tap changes is required for both cases; hence, the discussion here considers only the results obtained and shown in Figure 7-4 (a).

Control cycles, on the other hand, can clearly be seen as significantly influencing the number of tap changes. As also highlighted in the time-series performance (see section 7.3), the number of tap changes is indeed decreasing with longer control cycles. For example, 1-min control cycles lead to an average of 120 and 164 tap changes for 40% and 100% of PV penetrations, respectively. These figures are significantly reduced to an average of 10 and 20 when 20 and 30-min control cycles are adopted, respectively. More importantly, assuming an average of 20 tap changes per day, the resulting annual number of tap changes, in this case, represents only around 1% of the maximum number that, according to manufactures, the LV OLTC-fitted transformer can withstand [145]. Moreover, the ability of the proposed OLTC

control logic to keep a low number of tap changes while effectively managing voltage issues can potentially keep maintenance costs low.



Figure 7-4 OLTC Control – Number of daily tap changes

Finally, considering that the number of non-compliant customers slightly increases with longer control cycles, the 20 and 30-min control cycles can be considered as a trade-off between the number of tap changes and EN 50160 non-compliant customers. Nonetheless, this setting should ultimately be based on the requirements of the DNO.

7.5 Summary

The analysis performed in this chapter provides a quantitative assessment of the benefits when adopting the proposed OLTC control logic LV OLTC-fitted transformers to manage voltage issues in PV-rich LV networks. It is shown that the adoption of the proposed OLTC control logic with limited monitoring (i.e., end point of each feeder) can efficiently shift the occurrence of voltage issues at a higher penetration level, beyond which the number of customers facing voltage issues can significantly reduce (compared to the case without control).

Furthermore, it is highlighted that the location of remote monitoring points in the feeders and the control cycle length play an important role in the performance of the control logic. The analysis demonstrates that having monitors only at the end of each feeder and adopting 20 or 30-min control cycles can be the most effective approach to limit tap changes (i.e., less impact on the OLTC utilisation) while minimising the number of customers that could be affected by voltage issues.

7.6 Comparison of Investigated Control Schemes

Table 7-2 summarises and compares the technical benefits (i.e., voltage and thermal management) of the proposed control schemes investigated in this Thesis, along with different implementation aspects (i.e., potential required investments, effects on customers).

	Technical Benefits		Potential Required Investments		
Control Method	Voltage	Thermal	Monitoring	New Assets	Effects on Customer
Volt-Var	Yes (Limited)	No (Increases)	No	No*	No
Volt-Var (over-rated)	Yes	No (Increases)	No	Yes (Over-rated Inverters)	Yes (Higher Inverter Cost)
Volt-Watt	Yes	Yes (Indirect)	No	No*	Yes (Curtailment)
СТС	Yes (Indirect)	Yes	Yes (Substation)	Yes (Comms)	Yes (Curtailment)
CTC + Volt- Watt	Yes	Yes	Yes (Substation)	Yes (Comms)	Yes (Curtailment)
BES (Normal)	No	No	No	No*	Yes (Reduced Bills)
BES (Advanced)	Yes	Yes (Indirect)	No	No*	Yes (Reduced Bills)
OLTC	Yes	No	Yes (End Points)	Yes (OLTC, Comms)	No

 Table 7-2 Control Schemes Summary

*Assumed that the corresponding PV inverters/BES systems are already installed by the customers.

In terms of technical benefits, Table 7-2 shows that the least beneficial control schemes are the Volt-Var with normal PV inverters and the adoption of BES systems with normal operation. These control schemes provide limited (or no) benefits to the voltage and thermal management. Although, the adoption of the Volt-Var control with over-rated inverters can help managing voltage issues, it entails higher inverter costs to the end customers.

On the other hand, adopting the Volt-Watt control, voltage issues can directly be managed while indirectly reducing potential thermal issues; however, this requires some of the PV generation to be curtailed, leading to reduced profits for the PV owners. Assuming that residential-scale BES systems are in place, the adoption of the proposed advanced operation for BES systems can also bring similar benefits (i.e., direct voltage and indirect thermal management), yet, without curtailing energy.

Furthermore, it provides benefits not only to the DNO but also to the customers (i.e., reduced electricity bills). Nevertheless, the corresponding control performance might be influenced by the corresponding battery size. Similarly, the adoption of LV OLTC-fitted transformers with the proposed control logic can also provide significant benefits in voltage management without affecting customers (curtailment of PV generation). However, some investment is required for the installation of the OLTC-fitted transformer, monitoring devices and communication infrastructure.

Ultimately, the combined centralised thermal and decentralised voltage control (i.e., CTC+Volt-Watt) brings the most benefits in terms of technical issues as it directly manages both voltage and thermal issues. Despite the excellent performance achieved with the corresponding control scheme, which also entails PV generation curtailment, some investment in monitoring (limited) and communication infrastructure is required.

Although an economic assessment (future work) is essential to have a well–rounded comparison, this table allows comparing and identifying the most adequate control scheme based on the technical requirements (e.g., voltage and/or thermal management) and potential investment. Moreover, it is highlighted, that, a trade-off should always be made between the required solution (e.g., voltage and/or thermal management), the corresponding investment (e.g., monitoring, communications) and the effects on customers.

Conclusions and Future Work

8.1 Introduction

This chapter provides first an overview of the research and main literature gaps. Then, the contributions of this Thesis are presented along with a summary of conclusions. Lastly, potential improvements and future work are discussed.

8.2 Research Area Overview and Main Gaps

The cost reduction in residential-scale PV systems along with the incentives rolled out in several countries around the world and in particular the EU, are the two main factors for the rapid adoption of residential-scale PV systems in LV distribution networks. However, the increased penetration of residential-scale PV systems is likely to lead to technical issues such as voltage rise and thermal overloads of the most important and expensive network assets (i.e., transformer, feeder cables).

These issues, in turn, limit the ability of LV networks to accommodate larger volumes of low carbon technologies, in particular, residential-scale PV systems. Consequently, to facilitate the connection of larger volumes without the need for expensive and time-consuming network reinforcements, DNOs should move from passive to active LV networks, where controllable devices and LCTs are actively managed to solve technical issues. Emerging LCTs (i.e., residential-scale PV and BES systems) offer numerous controllability options (i.e., generation curtailment, reactive compensation, power reduction by storing energy) which can be used in either a "centralised" (i.e., transformer as the 'hub') or "decentralised" (local) way to manage voltage and thermal issues. Even more mature technologies, commonly used in higher voltage levels, can be adopted in LV networks. Technologies like OLTCs can be adopted at secondary distribution substations (i.e., MV/LV) and actively managed to alleviate voltage issues.

Despite the fact of active LV networks being a relatively new concept, the literature review carried out in this Thesis highlights that there is an increasing research interest in this area and, in particular, the control of residential-scale PV and BES systems. Although several active network management schemes have been proposed in the literature aiming to increase the hosting capacity of PV-rich LV networks, the majority are:

- Based on complex optimisation techniques which require expensive communication infrastructure and full network observability which are currently not available in LV networks. Considering the extensiveness of LV networks around the world, the installation of monitoring devices (to provide network visibility) might not be economically feasible. In addition, the adoption of optimisation methods might lead to significant computational times and increase the risk of not converging to a solution as the size of the optimisation problem significantly increases with network size and number of control variables. Consequently, it is essential to explore the extent to which already available devices, information and limited monitoring (i.e., secondary distribution substation) could be used to augment the network visibility.
- Limited to deterministic load flow analyses that consider specific PV penetration levels. Some others, even more limited, consider only several snapshots of load and generation profiles. These analyses are not suitable considering the intermittent behaviour of load and PV generation. Hence, to cope with the corresponding uncertainties and to truly understand the impacts on the electricity infrastructure to which the PV systems will connect to, as well as potential solutions, stochastic approaches must be adopted. More importantly high granularity (e.g., scale of seconds and minutes) time-series demand and generation profiles are essential to allow realistic modelling and time-series analyses.
- Focused on managing voltage issues only, ignoring the fact that thermal issues might also occur with high PV penetration levels. Therefore, integrated solutions that consider and manage simultaneously both voltage and thermal issues need to be developed.
- Assuming PV inverters with ratings that can be high enough (i.e., oversized inverters) to maintain the full solar output (kWp) and the requested vars.

However, commercially available residential-scale PV inverters might have limited capability to inject or absorb the required reactive power (i.e., vars) in periods of high power output (i.e., sunny weather condition). Hence, realistic studies that consider this important limitation are required to understand the extent to which reactive compensation through residential-scale PV systems can help managing voltage issues.

8.3 Thesis Main Contributions

Considering the gaps in literature, the main contribution of this Thesis is the proposal and development of innovative, practical and scalable active network management schemes that use limited network monitoring and communication infrastructure to actively manage:

- Residential-scale PV systems;
- Residential-scale BES systems, and;
- LV OLTC-fitted transformers.

The adoption of the proposed active network management schemes which make use of already available devices (i.e., PV inverters, BES systems), information and require limited network monitoring (i.e., at the secondary distribution substation) allows making the adoption of the proposed schemes, in LV networks, more practical and cost-effective. In addition, their generic nature allows being easily adapted to control any other low carbon technology that can support active or reactive power control.

In addition, to tackle the challenges related to this research (i.e., lack of realistic LV network modelling with high resolution time-series analyses), this Thesis contributes by considering a fully modelled real residential French LV network (three-phase four-wire) with different characteristics and number of customers. Moreover, realistic (1-min resolution) daily time-series household (from smart meters) and PV generation profiles are considered. A stochastic approach (i.e., Monte Carlo) is also adopted to cater for the uncertainties related to household demand as well as PV generation and location.

The following subsections present the details of the main contributions.

8.3.1 Management of Residential-scale PV Systems

<u>Decentralised Voltage Control (Volt-Watt)</u>. It is demonstrated that a universal, single set of Volt-Watt settings is possible to be identified with the proposed method as the most adequate to effectively manage voltage issues, regardless the penetration, in a given LV network. This brings benefits to the DNO as it increases the scalability of the corresponding control method due to the fact that these settings are universal for all PV inverters, require a one-off setup (e.g., on the installation day) and are valid for any PV penetration.

<u>Decentralised Voltage Control (Volt-Var</u>). The limitations and drawbacks adopting such a control method, with commercially available residential-scale PV inverters, are demonstrated for the first time. Findings show that the benefits of this control approach are minimal due to the limited reactive capability of PV inverters at peak generation periods, where voltages are higher. This allows DNOs to take more informed decisions when adopting decentralised voltage control with reactive compensation (i.e., Volt-Var).

<u>Centralised Thermal Controller</u>. An innovative and scalable centralised control logic which bridges the gaps in the literature (e.g., adoption of complex optimisation techniques, requirement of extensive network monitoring, not scalable, etc.) is developed and proposed in this Thesis. The proposed control logic aims to manage thermal overloads in PV-rich LV networks (for any penetration) while using limited network observability (i.e., substation measurements only). The proposed control logic, which its performance is successfully assessed considering different PV penetration levels (i.e., 0-100%), offers significant benefits to DNOs as it does not require the installation of remote monitoring points and reduces the need of costly communication infrastructure. More importantly, its generic nature, allows to be deployed and work on any PV-rich LV network while the performance is expected to be the same as the proposed method will always curtail the amount of energy required to keep the utilisation of assets within limits.

<u>Combined Centralised Thermal and Decentralised Voltage Control</u>. To provide a complete active network management aiming to solve both thermal and voltage

issues at any PV penetration level, this Thesis proposes the combination of the centralised thermal controller with the decentralised voltage control (i.e., Volt-Watt). Unlike other solutions that are not easily implementable (due to complex optimisation algorithms, increased network visibility, expensive communication infrastructure), the proposed control method offers a more cost-effective solution requiring limited network observability. Again, as previously mentioned the generic nature of the proposed method, allows to be deployed and work on any PV-rich LV network.

8.3.2 Management of Residential-scale Battery Energy Storage Systems

For the first time, an advanced operation mode that provides benefits both to the customer (i.e., increased self-consumption) and the DNO (i.e., voltage and thermal management) is proposed for residential-scale BES systems. The proposed operation mode, which can be programmed on residential-scale BES systems, offers the advantages of the decentralised control approach thus requiring no communication infrastructure. Based on local measurements (i.e., inverter connection point) the advanced operation mode can progressively charge/discharge the BES system in a way that increases the customer's self-consumption while effectively eliminating all voltage issues up to high PV penetration levels beyond which it can significantly reduce them. The decentralised and generic nature of the proposed operation mode makes it replicable on any PV-rich LV network; however, it should be highlighted that the overall performance might be influenced by the size of the installed BES systems. For instance, although smaller sizes of BES systems will still bring benefit to the customers (i.e., reduced energy bills) they might not bring enough benefits to the DNO (i.e., voltage and thermal management) due to their limited storage capability.

8.3.3 Management of OLTC-fitted LV Transformers

A scalable and adaptive OLTC control logic that aims in managing contrasting voltages issues, rise (due to the presence of generation) and drop (due to loads) is proposed in this research. Compared to traditional OLTC control approaches (i.e., fixed voltage target) that do not consider the diversified voltage levels among different feeders (due to uneven penetration of LCTs, load unbalance, etc.), the proposed method allows the automatic update of the voltage target (at the busbar)

according to network conditions while using limited network observability coming from critical remote monitoring points (i.e., feeders' end points). Crucially, this provides the significant benefit of easily adapting to network changes (i.e., additional installation of PV systems or loads) without the need of reconfiguring OLTC settings. The proposed OLTC control logic found to be effective in eliminating all voltage issues up to high PV penetration levels beyond which it can significantly reduce them.

8.4 Conclusions

The following subsections present the specific findings and knowledge gained from the analyses performed in this Thesis, considering the proposed active network management schemes.

8.4.1 Residential-Scale PV systems

A summary of the main findings and knowledge gained from the analyses performed considering the proposed active management schemes for residential-scale PV systems is presented below.

8.4.1.1 Decentralised Voltage Control

The benefits of adopting a decentralised voltage control is investigated in section 5.2 using the Volt-Var and Volt-Watt control functions and considering different control set-points for each control function. The following findings are obtained:

- 1. Volt-Var Analysis
 - Normal rated inverters
 - Voltage rise problems cannot be solved due to the limited reactive power capability of the normal PV inverter during high generation periods i.e., the rated kVA power is almost reached.
 - The absorption of reactive power increases the current flowing in the network. As a result the utilisation level of the assets increases which might lead to the overloading of the transformer and/or feeder cables.

• 10% over-rated inverters

- The application of Volt-Var control with over-rated inverters can provide a significantly better voltage management without the need of curtailing any generation. However, oversizing the inverters entails higher costs to the end customers which might not be willing to cover.
- Although the adoption of over-rated inverters achieves better performance in terms of voltage management, it leads to even higher levels of asset utilisation (compared to normal inverters). This finding, which is not covered in the literature, allows DNOs to take more informed decisions when adopting decentralised voltage control with reactive compensation.

2. Volt-Watt Analysis

- This method found to be more effective in managing voltages (compared to the Volt-Var). Although this allows increasing the network's hosting capacity, it comes with the expense of curtailing some PV generation; thus potentially reducing the profits of the PV owners.
- It was demonstrated that a single Volt-Watt curve can be identified for a given network as to provide a trade-off between voltage issues and the required volume of curtailment. This finding allows increasing the scalability of the corresponding method as these settings are universal for all PV inverters and require a one-off setup (e.g., on the installation day).
- Although this control function is not directly managing thermal issues, it can also help reducing the utilisation of the assets, thus dealing with another important technical issue that might occur in PV-rich LV networks.

8.4.1.2 Centralised Thermal Controller (CTC)

The performance of the proposed centralised thermal controller is investigated in section 5.3. The following findings are obtained:

- The proposed centralised thermal controller is an effective solution to manage and solve thermal issues while indirectly reducing the number of customers experiencing voltage issues in PV-rich LV networks.
- Although control cycles do not significantly influence the overall performance of thermal management, it is highlighted that the shorter the control cycle, the better the performance. This, however, could potentially lead to higher number of control actions (i.e., PV signals) which might cause operational challenges (e.g., in communication).
- Findings highlight that a specific control cycle can be identified as a trade-off control setting between the number of control actions (i.e., PV signals) and operation performance (in terms of thermal and voltage issues).

8.4.1.3 Combined Centralised Thermal and Decentralised Voltage Control

The performance of the proposed combined centralised thermal and decentralised voltage control is assessed in section 7.3.3. The following findings are obtained:

- The adoption of the proposed combined thermal and decentralised voltage control provides a robust active network management in PV-rich LV networks, as it effectively eliminates all technical issues (both voltage and thermal) regardless the PV penetration level. This can be a potential solution for DNOs to efficiently manage both thermal and voltage issues and hence accelerate the transition towards active LV networks.
- As expected, in order to solve both voltage and thermal issues, slightly more energy is curtailed compared to the other investigated control methods. This however, allows accommodating higher number of LCTs in LV networks without the need of costly network reinforcements, thus accelerating the transition towards an eco-friendly low carbon environment.

8.4.2 Management of Residential-scale BES systems

The analysis performed in section Chapter 6 provides a quantitative assessment of the benefits obtained when adopting residential-scale BES systems in combination with residential-scale PV systems in order to maximise customer benefits while managing voltage and thermal issues. The following findings are obtained:

- The adoption of residential-scale BES systems with a normal operation mode although provides significant benefits to the customers (i.e., reduced dependence to the grid) it does not provide any benefit to the DNO (i.e., voltage or thermal management).
- The following limitations of the normal operation mode were highlighted:
 - BES systems achieve full SOC before peak generation (i.e., PV impacts still occur around noon); and,
 - BES systems start charging the excess of PV generation (early daylight) with a SOC already above the lower reserve (i.e., not adequately discharged overnight).
- The proposed advanced operation mode overcomes the aforementioned limitations and while still beneficial to the customers (i.e., reduced dependence to the grid, hence reduced electricity bills) it also allows managing voltage (directly) and thermal (indirectly) issues.
- The proposed operation mode can effectively eliminate all voltage issues up to high PV penetration levels beyond which it can significantly reduce them.
- The proposed advanced operation mode significantly reduces customers' dependence to the grid (hence reduced electricity bills) while they still benefit from the quality of service brought by the grid (e.g., supply always available).
- The adoption of residential-scale BES systems with the proposed advanced operation mode can significantly increase the harvesting of "green" energy (otherwise curtailed), coming from PV generation, while providing benefits to both customers and DNO. This in turn allows accommodating even higher number of LCTs in LV networks, accelerating that way the transition towards an eco-friendly and low carbon environment.
8.4.3 Management of LV OLTC-fitted LV Transformers

Chapter 7 presents a quantitative assessment of the benefits when adopting OLTCfitted transformers with the proposed control logic in order to manage voltage issues in PV-rich LV networks. The following findings were obtained:

- The adoption of LV OLTC-fitted transformer with the proposed control logic regardless the adopted monitoring point and control cycle can effectively shift the occurrence of voltage issues at a higher penetration level; beyond which the number of customers facing voltage issues can significantly reduce.
- Findings highlight that the location of remote monitoring and the control cycle length play an important role in the performance of the control logic.
 - The shorter the control cycle, the better the voltage performance and the higher the number of tap changes.
 - The closer the remote monitoring to the last customer the better the voltage performance.
- The analysis demonstrates that having monitors only at the end of each feeder and adopting longer control cycles can be the most cost-effective approach (i.e., less monitoring devices) to limit tap changes (i.e., less impact on the OLTC utilisation; reduced wear and tear hence less maintenance costs) while minimising the number of customers that could be affected by voltage issues.

8.5 Future Work

Although research objectives, defined at the beginning of this research, have been successfully achieved in this Thesis there are still areas in which this work can be enhanced and further developed.

8.5.1 Seasonal Analyses and different representative LV networks

This work focused on multiple summer daily scenarios (different demand and PV profiles) given that they could result in higher technical impacts. However, to better assess the performance of the proposed active network management schemes, multiple seasons could also be investigated so as to produce a more meaningful year-

long performance analysis. This will allow DNOs identifying the most adequate and cost effective control scheme able to provide voltage and thermal management throughout the year. In addition, considering the adoption of BES systems, seasonal analysis will also enable customers understanding the overall year-round benefits might gain with the adoption of residential-scale BES systems.

Similarly, to determine the most effective control scheme, that brings most benefits, analyses could also be carried out on different representative LV networks, as a future work.

8.5.2 Different BES systems and controls

This work considers only one type (i.e., lithium), make (i.e., Tesla PowerWall) and two sizes (i.e., 10 and 20kWh) of residential-scale BES systems. However, considering the wide range of BES systems available in the market and the continuous development of this technology, the analysis could potentially consider different battery types, makes and sizes of residential-scale BES systems to assess the performance of the proposed advanced operation mode in a more diverse scenario.

The adoption of reactive compensation through BES systems could also be investigated as a potential solution to manage voltage issues in PV-rich LV networks. This, however, would require the adoption of inverters that are able to support such operation; hence increasing the corresponding costs, that customers might not be willing to pay.

8.5.3 Economic Assessment

An economic assessment considering both the investment and operational costs of the proposed active network management schemes could be considered as a future work. An economic assessment compared with the traditional network reinforcement costs will allow understanding how economically feasible (or not) the proposed schemes are. Potential savings that might be achieved (or not) adopting the proposed schemes (instead of reinforcement) will enable DNOs take more informed decisions when adopting active network management schemes. Moreover, the economic assessment of the proposed schemes can identify the most cost-effective scheme that allows managing technical issues (in PV-rich LV networks), and potentially help in accelerating the transition towards active LV networks.

8.5.4 Consideration and combination of other LCTs

This Thesis has considered only two residential-scale LCTs (PV and BES systems) whereas other emerging technologies such as electric vehicles (EVs) could also be considered. In addition, the potential advantages or disadvantages resulting from the mix of these technologies could be investigated. Such scenarios (mix of technologies) will allow to truly understanding the benefits of adopting BES systems in combination with other technologies. For instance, the excess of PV generation stored in the BES systems during the day can potentially support not only the household demand during the night but also, if required, charge the connected EV.

8.5.5 Comparison with Optimal Solution

To understand the extent to which the proposed schemes provide an effective active network management and achieve adequate performance, the current research could be enhanced, for instance, by performing a comparison with an ideal optimization framework driven by an AC Optimal Power Flow (OPF). Such analysis can provide important insights on how far the outcomes of the proposed schemes are from the optimal solution. These insights could potentially highlight (or not) the practicality and scalability of the proposed schemes. For instance, a comparison of the proposed combined thermal and decentralised voltage control with the optimal solution will highlight the optimal amount of energy that needs to be curtailed as well as the number of control actions required in order to eliminate voltage and thermal issues.

8.5.6 Combination of Proposed Active Network Management Schemes

With respect to the active network management schemes proposed in this Thesis, a further development to allow additional penetrations of low carbon technologies in LV networks is the combination of some of the proposed schemes. For example, the adoption of the centralised thermal controller in coordination with the LV OLTC-fitted transformer could potentially enhance the flexibility of LV networks and, therefore, allow higher penetrations of low carbon technologies. In terms of energy curtailment, such combination could potentially achieve less generation curtailment (compared to the proposed centralised thermal and decentralised voltage control) as

the LV OLTC-fitted transformer manages voltages without affecting the PV generation (as opposed to the case with Volt-Watt).

The proposed OLTC control logic could also be investigated in combination with the residential-scale BES systems adopting the proposed advanced operation mode. Considering the fact that the residential-scale BES systems can effectively eliminate voltage issues up to a specific PV penetration level, the LV OLTC-fitted transformer can potentially alleviate voltage issues beyond that.

A more sophisticated decentralised controller that could control both residential-scale PV and BES systems could be developed based on the proposed advanced operation mode for residential-scale BES systems and the reactive capability (i.e., Volt-Var) of PV systems. The decentralised controller could essentially instruct the PV systems to absorb reactive power in moments where the advanced operation mode of BES systems alone is not enough to keep the voltage within limits. The adoption of such scheme, will allow catering for potential voltage and thermal issues without the need of curtailing "green" energy (produced by PV systems). More importantly, the decentralised nature of this approach will not require any communication infrastructure.

In line with the above, a fully decentralised control scheme could also be investigated combining the decentralised Volt-Var and Volt-Watt control to address voltage issues while giving priority to the Volt-Var control that does not lead to generation curtailment. Coordination (i.e., ensuring Volt-Var priority) could be achieved by adequately specifying Volt-Var and Volt-Watt curves such that only the former acts up to specific voltages where the latter acts once the voltage has reached a specific value. In addition, such combination might increase the ability of residential-scale PV systems to absorb reactive power during high sun radiations as the parallel operation of the Volt-Watt control will potentially reduce the inverter's active power output, hence allowing it to absorb more reactive power.

References

- [1] "Renewable energy in Europe 2016 Recent growth and knock-on effects," Energy, Ed.: European Commission, 2016. [Online]. Available: http://www.eea.europa.eu/publications/renewable-energy-in-europe-2016/download.
- [2] "Renewables 2016: Global Status Report," Paris, France: REN21: Renewable Energy Policy Network for the 21st Century, 2016. [Online]. Available: http://www.ren21.net/wpcontent/uploads/2016/10/REN21_GSR2016_FullReport_en_11.pdf.
- [3] "The German solar rooftop experience," Goverment of India Ministry of New and Renewable Energy, 2016. [Online]. Available: http://mnre.gov.in/file-manager/UserFiles/workshop-gcrt-0870616/german.pdf.
- [4] "Net installed electricity generation capacity in Germany," Germany: Fraunhofer Institute for Solar Energy Systems ISE, 2016. [Online]. Available: https://www.energy-charts.de/power_inst.htm.
- [5] H. Wirth, "Recent Facts about Photovoltaics in Germany," Germany: Fraunhofer ISE, 2016. [Online]. Available: https://www.ise.fraunhofer.de/en/publications/veroeffentlichungen-pdfdateien-en/studien-und-konzeptpapiere/recent-facts-about-photovoltaics-ingermany.pdf.
- [6] "Snapshot of Global Photovoltaic Markets 2015," International Energy Agency (IEA), 2016. [Online]. Available: http://www.ieapvps.org/fileadmin/dam/public/report/PICS/IEA-PVPS_ ____A_Snapshot_of_Global_PV_-_1992-2015_-_Final_2_02.pdf.
- [7] "Tableau de bord : Solaire Photovoltaïque Deuxième Trimestre 2016 (Solar Photovoltaic Second Quarter 2016)," France: Ministère De L'écologie, Du Développement Durable Et De L'énergie, 2016. [Online]. Available: http://www.developpement-durable.gouv.fr/IMG/pdf/Tableau_de_bord_-_Solaire_photovoltaique.pdf.
- [8] "France rolls out solar tenders for 20 GW by 2023," PV Magazine, 2016. [Online]. Available: http://www.pvmagazine.com/news/details/beitrag/france-rolls-out-solar-tenders-for-20-gwby-2023_100025234/#ixzz4OIvNC055.
- [9] "Solar photovoltaics deployment (September 2016)," UK: Department for Business, Energy & Industrial Strategy, March 2014. [Online]. Available: https://www.gov.uk/government/statistics/solar-photovoltaics-deployment.
- [10] "German and UK PV markets slow down," Renewables International Magazine, 2016. [Online]. Available: http://www.renewablesinternational.net/german-and-uk-pv-markets-slowdown/150/452/94549/.
- [11] H. Kuang, S. Li, and Z. Wu, "Discussion on advantages and disadvantages of distributed generation connected to the grid," in *Electrical and Control Engineering (ICECE)*, 2011 International Conference on, 2011, pp. 170-173.

- [12] "Global Market Outlook for Photovoltaics 2014-2018," European Photovoltaic Industry Association (EPIA). [Online]. Available: http://www.cleanenergybusinesscouncil.com/site/resources/files/reports/EPIA _Global_Market_Outlook_for_Photovoltaics_2014-2018_-_Medium_Res.pdf.
- [13] K. Fekete, Z. Klaic, and L. Majdandzic, "Expansion of the residential photovoltaic systems and its harmonic impact on the distribution grid," *Renewable Energy*, vol. 43, pp. 140-148, 7// 2012.
- [14] M. E. Baran, H. Hooshyar, Z. Shen, and A. Huang, "Accommodating High PV Penetration on Distribution Feeders," *IEEE Transactions on Smart Grid*, vol. 3, no. 2, pp. 1039-1046, 2012.
- [15] A. Navarro-Espinosa and L. F. Ochoa, "Probabilistic Impact Assessment of Low Carbon Technologies in LV Distribution Systems," *IEEE Transactions* on Power Systems, vol. 31, no. 3, pp. 2192-2203, 2016.
- [16] T. Haggis, Network design manual, UK: E.ON, 2006. [Online]. Available: www.eon.uk.com/downloads/network_design_manual.pdf. Accessed on January 2015.
- [17] Technical guide The MV/LV transformer substations Italy: ABB, 2015. [Online]. Available: http://new.abb.com/docs/librariesprovider27/defaultdocument-library/abb-transformerstations_ebook.pdf?sfvrsn=2. Accessed on February 2017.
- [18] S. Volut and J. Schonek, Electrical installation guide According to IEC International Standards, France: Schneider Electric, 2008. [Online]. Available: https://www.schneiderelectric.pl/documents/Polish_Catalogue/Technical/eig.pdf. Accessed on February 2017.
- [19] C. Puret, MV public distribution networks throughout the world: Merlin Gerin, 1992. [Online]. Available: http://www.studiecd.dk/cahiers_techniques/MV_public_distribution_network s.pdf. Accessed on February 2017.
- [20] A. Canova, L. Giaccone, F. Spertino, and M. Tartaglia, "Electrical Impact of Photovoltaic Plant in Distributed Network," *Industry Applications, IEEE Transactions on*, vol. 45, no. 1, pp. 341-347, 2009.
- [21] N. Jenkins, R. Allan, P. Crossley, D. Kirschen, and G. Strbac, *Embedded Generation*. London: IET, 2000, p. 292.
- [22] A. Manzoni and R. Castro, "Microgeneration Impact on LV Distribution Grids: A Review of Recent Research on Overvoltage Mitigation Techniques," vol. 6, no. 1, pp. 117-131, 2016.
- [23] J. D. Watson, N. R. Watson, D. Santos-Martin, A. R. Wood, S. Lemon, and A. J. V. Miller, "Impact of solar photovoltaics on the low-voltage distribution network in New Zealand," *IET Generation, Transmission & Distribution*, vol. 10, no. 1, pp. 1-9, 2016.
- [24] J. Descheemaeker, M. Van Lumig, and J. Desmet, "Influence of the supply voltage on the performance of household appliances," in *23rd International Conference on Electricity Distribution (CIRED)*, Lyon, France, 2015, p. 5.

- [25] G. W. Ault, R. A. F. Currie, and J. R. McDonald, "Active power flow management solutions for maximising DG connection capacity," in 2006 IEEE Power Engineering Society General Meeting, 2006, p. 5 pp.
- [26] A. Collinson, F. Dai, A. Beddoes, and J. Crabtree, "Solutions for the Connection and Operation of Distributed Generation," UK Departement of Trade and Industry (DTI), 2003. [Online]. Available: http://webarchive.nationalarchives.gov.uk/20100919181607/http://www.ensg. gov.uk/assets/solutions.pdf.
- [27] L. Ochoa, A. Keane, and G. Harrison, "Minimizing the reactive support for distributed generation: Enhanced passive operation and smart distribution networks," in 2012 IEEE Power and Energy Society General Meeting, 2012, pp. 1-1.
- [28] T. Stetz, F. Marten, and M. Braun, "Improved Low Voltage Grid-Integration of Photovoltaic Systems in Germany," *Sustainable Energy, IEEE Transactions on*, vol. 4, no. 2, pp. 534-542, 2013.
- [29] T. Stetz, K. Diwold, M. Kraiczy, D. Geibel, S. Schmidt, and M. Braun, "Techno-Economic Assessment of Voltage Control Strategies in Low Voltage Grids," *Smart Grid, IEEE Transactions on*, vol. 5, no. 4, pp. 2125-2132, 2014.
- [30] J. W. Smith, W. Sunderman, R. Dugan, and B. Seal, "Smart inverter volt/var control functions for high penetration of PV on distribution systems," in *Power Systems Conference and Exposition (PSCE)*, 2011 IEEE/PES, 2011, pp. 1-6.
- [31] A. O'Connell and A. Keane, "Volt–var curves for photovoltaic inverters in distribution systems," *IET Generation, Transmission & Distribution,* 2016.
- [32] M. Juamperez, G. Yang, and S. KjÆR, "Voltage regulation in LV grids by coordinated volt-var control strategies," (in English), *Journal of Modern Power Systems and Clean Energy*, vol. 2, no. 4, pp. 319-328, 2014/12/01 2014.
- [33] A. Cagnano, E. D. Tuglie, M. Liserre, and R. A. Mastromauro, "Online Optimal Reactive Power Control Strategy of PV Inverters," *IEEE Transactions on Industrial Electronics*, vol. 58, no. 10, pp. 4549-4558, 2011.
- [34] B. I. Craciun, D. Sera, E. A. Man, T. Kerekes, V. A. Muresan, and R. Teodorescu, "Improved voltage regulation strategies by PV inverters in LV rural networks," in 2012 3rd IEEE International Symposium on Power Electronics for Distributed Generation Systems (PEDG), 2012, pp. 775-781.
- [35] S. Weckx and J. Driesen, "Optimal Local Reactive Power Control by PV Inverters," *IEEE Transactions on Sustainable Energy*, vol. 7, no. 4, pp. 1624-1633, 2016.
- [36] M. Rylander, L. Huijuan, J. Smith, and W. Sunderman, "Default volt-var inverter settings to improve distribution system performance," in 2016 IEEE Power and Energy Society General Meeting (PESGM), 2016, pp. 1-5.
- [37] I. Kim, R. G. Harley, R. Regassa, and Y. D. Valle, "The effect of the volt/var control of photovoltaic systems on the time-series steady-state analysis of a

distribution network," in 2015 Clemson University Power Systems Conference (PSC), 2015, pp. 1-6.

- [38] B. Bletterie, S. Kadam, R. Bolgaryn, and A. Zegers, "Voltage Control with PV Inverters in Low Voltage Networks In Depth Analysis of Different Concepts and Parameterization Criteria," *IEEE Transactions on Power Systems*, vol. 32, no. 1, pp. 177-185, 2017.
- [39] C. Restrepo, A. Salazar, H. Schweizer, and A. Ginart, "Residential Battery Storage: Is the Timing Right?," *IEEE Electrification Magazine*, vol. 3, no. 3, pp. 14-21, 2015.
- [40] J. P. Barton and D. G. Infield, "Energy storage and its use with intermittent renewable energy," *IEEE Transactions on Energy Conversion*, vol. 19, no. 2, pp. 441-448, 2004.
- [41] J. Tant, F. Geth, D. Six, P. Tant, and J. Driesen, "Multiobjective Battery Storage to Improve PV Integration in Residential Distribution Grids," *IEEE Transactions on Sustainable Energy*, vol. 4, no. 1, pp. 182-191, 2013.
- [42] H. Sugihara, K. Yokoyama, O. Saeki, K. Tsuji, and T. Funaki, "Economic and Efficient Voltage Management Using Customer-Owned Energy Storage Systems in a Distribution Network With High Penetration of Photovoltaic Systems," *IEEE Transactions on Power Systems*, vol. 28, no. 1, pp. 102-111, 2013.
- [43] K. Worthmann, C. M. Kellett, P. Braun, L. Grüne, and S. R. Weller, "Distributed and Decentralized Control of Residential Energy Systems Incorporating Battery Storage," *IEEE Transactions on Smart Grid*, vol. 6, no. 4, pp. 1914-1923, 2015.
- [44] M. N. Kabir, Y. Mishra, G. Ledwich, Z. Y. Dong, and K. P. Wong, "Coordinated Control of Grid-Connected Photovoltaic Reactive Power and Battery Energy Storage Systems to Improve the Voltage Profile of a Residential Distribution Feeder," *IEEE Transactions on Industrial Informatics*, vol. 10, no. 2, pp. 967-977, 2014.
- [45] F. Marra, G. Yang, C. Træholt, J. Østergaard, and E. Larsen, "A Decentralized Storage Strategy for Residential Feeders With Photovoltaics," *IEEE Transactions on Smart Grid*, vol. 5, no. 2, pp. 974-981, 2014.
- [46] S. Hashemi, J. Østergaard, and G. Yang, "A Scenario-Based Approach for Energy Storage Capacity Determination in LV Grids With High PV Penetration," *IEEE Transactions on Smart Grid*, vol. 5, no. 3, pp. 1514-1522, 2014.
- [47] G. Byson. (2015). *Low Voltage Integrated Automation (LoVIA)*. Available: http://www.enwl.co.uk/about-us/the-future/lcnf-tier-1-nia/lovia
- [48] "Smart Metering deployment in the European Union," ed: European Commission, 2016.
- [49] "Low Voltage Network Solutions," Electricity North West Limited, 2014. [Online]. Available: http://www.enwl.co.uk/docs/default-source/future-low-voltage/lv-network-solutions-close-down-june-14-v4-1-final.pdf?sfvrsn=2.
- [50] M. Sebastian and J. Chaniolleau, "Barriers to the massive connection of DRES in distribution grids," European Regional Development Fund, 2015.

[Online]. Available: http://grid4eu.blob.core.windows.net/media-igreengrid-prod/43968/20150703_igreengrid_identification-of-the-barriers-.pdf.

- [51] A. T. Procopiou and L. F. Ochoa, "Voltage Control in PV-Rich LV Networks without Remote Monitoring," *IEEE Transactions on Power Systems*, vol. PP, no. 99, pp. 1-1, 2016.
- [52] A. T. Procopiou, J. Quirós-Tortós, and L. F. Ochoa, "HPC-Based Probabilistic Analysis of LV Networks With EVs: Impacts and Control," *IEEE Transactions on Smart Grid*, vol. 8, no. 3, pp. 1479-1487, 2017.
- [53] A. T. Procopiou, C. Long, and L. F. Ochoa, "On the Effects of Monitoring and Control Settings on Voltage Control in PV-Rich LV Networks," in *IEEE PES General Meeting*, Denver (CO), 2015.
- [54] L. Chao, A. T. Procopiou, L. F. Ochoa, G. Bryson, and D. Randles, "Performance of OLTC-based control strategies for LV networks with photovoltaics," in *Power & Energy Society General Meeting*, 2015 IEEE, 2015, pp. 1-5.
- [55] A. T. Procopiou and L. F. Ochoa, "Voltage Control in PV-Rich LV Networks Without Remote Monitoring," *IEEE Transactions on Power Systems*, vol. 32, no. 2, pp. 1224-1236, 2017.
- [56] A. T. Procopiou and L. F. Ochoa, "Deliverable 2.4 "Benefits of adopting storage devices in LV networks"," Active Management of LV Networks, France: Electricite de France (EDF) R&D, 2016. [Online]. Available: https://www.researchgate.net/publication/309259223_Deliverable_24_Benefit ts_of_adopting_storage_devices_in_LV_networks.
- [57] A. T. Procopiou and L. F. Ochoa, "Deliverable 2.3 "Benefits of controlling EVs and PV in combination with other technologies"," Active Management of LV Networks, France: Electricite de France (EDF) R&D, 2016. [Online]. Available: https://www.researchgate.net/publication/303686962_Deliverable_23_Benefits_of_controlling_EVs_and_PV_in_combination_with_other_technologies.
- [58] A. T. Procopiou and L. F. Ochoa, "Deliverable 2.2 "Benefits of controlling EVs and PV"," Active Management of LV Networks, France: Electricite de France (EDF) R&D, 2015. [Online]. Available: https://www.researchgate.net/publication/283462139_Deliverable_22_Benefit ts_of_controlling_EVs_and_PV.
- [59] A. T. Procopiou and L. F. Ochoa, "Deliverable 1.3 "Monitoring Aspects and Benefits" and Deliverable 2.1 "EVs and PV: Literature review and initial modelling"," Active Management of LV Networks, France: Electricite de France (EDF) R&D, 2015. [Online]. Available: https://www.researchgate.net/publication/283569406_Deliverable_13_Monit oring_Aspects_and_Benefits_and_Deliverable_21_EVs_and_PV_Literature_ review_and_initial_modelling.
- [60] A. T. Procopiou and L. F. Ochoa, "Deliverable 1.2 "Benefits of OLTC transformers in LV"," Active Management of LV Networks, France: Electricite de France (EDF) R&D, 2014. [Online]. Available: https://www.researchgate.net/publication/283569568_Deliverable_12_Benefits_of_OLTC_transformers_in_LV.

- [61] A. T. Procopiou and L. F. Ochoa, "Deliverable 1.1 "Literature review and initial modelling"," Active Management of LV Networks Electricite de France (EDF) R&D, 2014. [Online]. Available: https://www.researchgate.net/publication/283566646_Deliverable_11_Literat ure_review_and_initial_modelling.
- [62] M. M. Begovic, K. Insu, D. Novosel, J. R. Aguero, and A. Rohatgi, "Integration of Photovoltaic Distributed Generation in the Power Distribution Grid," in System Science (HICSS), 2012 45th Hawaii International Conference on, 2012, pp. 1977-1986.
- [63] A. Navarro, L. F. Ochoa, and D. Randles, "Monte Carlo-based assessment of PV impacts on real UK low voltage networks," in *Power and Energy Society General Meeting (PES), 2013 IEEE*, 2013, pp. 1-5.
- [64] A. Navarro, P. Mancarella, L. F. Ochoa, and D. Randles, "Impacts of photovoltaics on low voltage networks: A case study for the North West of England," in 22nd International Conference and Exhibition on Electricity Distribution (CIRED 2013), 2013, pp. 1-4.
- [65] P. B. Kitworawut, D. T. Azuatalam, and A. J. Collin, "An investigation into the technical impacts of microgeneration on UK-type LV distribution networks," in 2016 Australasian Universities Power Engineering Conference (AUPEC), 2016, pp. 1-5.
- [66] M. A. Mahmud, M. J. Hossain, and H. R. Pota, "Worst case scenario for large distribution networks with distributed generation," in 2011 IEEE Power and Energy Society General Meeting, 2011, pp. 1-7.
- [67] S. N. Liew and G. Strbac, "Maximising penetration of wind generation in existing distribution networks," *IEE Proceedings Generation, Transmission and Distribution*, vol. 149, no. 3, pp. 256-262, 2002.
- [68] R. A. Shayani and M. A. G. d. Oliveira, "Photovoltaic Generation Penetration Limits in Radial Distribution Systems," *IEEE Transactions on Power Systems*, vol. 26, no. 3, pp. 1625-1631, 2011.
- [69] Y. Zhang, F. Li, Z. Hu, and G. Shaddick, "Quantification of low voltage network reinforcement costs: A statistical approach," *IEEE Transactions on Power Systems*, vol. 28, no. 2, pp. 810-818, 2013.
- [70] SolarEdge. (2014). *SolarEdge Inverters, Power Control Options*. Available: http://www.solaredge.com/files/pdfs/application_note_power_control_config uration.pdf
- [71] SMA. (2014). Sunny Boy PV Inverters. Available: http://www.sma.de/en/products/solarinverters/sunny-boy-3000tl-3600tl-4000tl-5000tl-with-reactive-power-control.html
- [72] "Requirements for micro-generating plants to be connected in parallel with public low-voltagedistribution networks (EN 50 438)," European Committee for Standardisation, 2016.
- [73] C. Gilligan. (2014). *IHS PV Inverter Rankings*. Available: https://technology.ihs.com/530072/ihs-pv-inverter-rankings-asian-supplierstake-another-big-leap

- [74] R. Schwalbe *et al.*, "DG-demonet smart LV grid increasing hosting capacity of LV grids by extended planning and voltage control," in 2015 International Symposium on Smart Electric Distribution Systems and Technologies (EDST), 2015, pp. 63-69.
- [75] A. Einfalt, F. Zeilinger, R. Schwalbe, B. Bletterie, and S. Kadam, "Controlling active low voltage distribution grids with minimum efforts on costs and engineering," in *Industrial Electronics Society, IECON 2013 - 39th Annual Conference of the IEEE*, 2013, pp. 7456-7461.
- [76] D. Geibel *et al.*, "Active, intelligent low voltage networks. Concept, realisation and field test results," in *Electricity Distribution (CIRED 2013)*, 22nd International Conference and Exhibition on, 2013, pp. 1-4.
- [77] P. Jahangiri and D. C. Aliprantis, "Distributed Volt/VAr Control by PV Inverters," *IEEE Transactions on Power Systems*, vol. 28, no. 3, pp. 3429-3439, 2013.
- [78] M. J. E. Alam, K. M. Muttaqi, and D. Sutanto, "A Multi-Mode Control Strategy for VAr Support by Solar PV Inverters in Distribution Networks," *IEEE Transactions on Power Systems*, vol. 30, no. 3, pp. 1316-1326, 2015.
- [79] K. Lemkens, F. Geth, P. Vingerhoets, and G. Deconinck, "Reducing overvoltage problems with active power curtailment-Simulation results," in *IEEE PES ISGT Europe 2013*, 2013, pp. 1-5.
- [80] S. Ghosh, S. Rahman, and M. Pipattanasomporn, "Distribution Voltage Regulation Through Active Power Curtailment With PV Inverters and Solar Generation Forecasts," *IEEE Transactions on Sustainable Energy*, vol. 8, no. 1, pp. 13-22, 2017.
- [81] F. Olivier, P. Aristidou, D. Ernst, and T. V. Cutsem, "Active Management of Low-Voltage Networks for Mitigating Overvoltages Due to Photovoltaic Units," *IEEE Transactions on Smart Grid*, vol. 7, no. 2, pp. 926-936, 2016.
- [82] S. Alyami, Y. Wang, C. Wang, J. Zhao, and B. Zhao, "Adaptive Real Power Capping Method for Fair Overvoltage Regulation of Distribution Networks With High Penetration of PV Systems," *IEEE Transactions on Smart Grid*, vol. 5, no. 6, pp. 2729-2738, 2014.
- [83] X. Su, M. A. S. Masoum, and P. J. Wolfs, "Optimal PV Inverter Reactive Power Control and Real Power Curtailment to Improve Performance of Unbalanced Four-Wire LV Distribution Networks," *IEEE Transactions on Sustainable Energy*, vol. 5, no. 3, pp. 967-977, 2014.
- [84] A. N. M. M. Haque, M. T. Rahman, P. H. Nguyen, and F. W. Bliek, "Smart curtailment for congestion management in LV distribution network," in 2016 *IEEE Power and Energy Society General Meeting (PESGM)*, 2016, pp. 1-5.
- [85] A. N. M. M. Haque, P. H. Nguyen, T. H. Vo, and F. W. Bliek, "Agent-based unified approach for thermal and voltage constraint management in LV distribution network," *Electric Power Systems Research*, vol. 143, pp. 462-473, 2// 2017.
- [86] G. D. Carne, M. Liserre, K. Christakou, and M. Paolone, "Integrated voltage control and line congestion management in Active Distribution Networks by

means of smart transformers," in 2014 IEEE 23rd International Symposium on Industrial Electronics (ISIE), 2014, pp. 2613-2619.

- [87] N. Etherden and M. H. J. Bollen, "Increasing the hosting capacity of distribution networks by curtailment of renewable energy resources," in 2011 *IEEE Trondheim PowerTech*, 2011, pp. 1-7.
- [88] "Validation of Photovoltaic (PV) Connection Assessment Tool (Closedown Report)," vol. 1.1: UK Power Networks, 2015. [Online]. Available: https://goo.gl/IPtEiW. Accessed on 26 January 2017.
- [89] B. Bletterie, S. Kadam, R. Pitz, and A. Abart, "Optimisation of LV networks with high photovoltaic penetration-Balancing the grid with smart meters," in *2013 IEEE Grenoble Conference*, 2013, pp. 1-6.
- [90] W. Peng, Y. Baghzouz, and S. Haddad, "Local load power factor correction by grid-interactive PV inverters," in *2013 IEEE Grenoble Conference*, 2013, pp. 1-6.
- [91] H. Pezeshki and P. Wolfs, "Impact of high PV penetration on distribution transformer life time," in 2013 IEEE Power & Energy Society General Meeting, 2013, pp. 1-5.
- [92] "Tesla Powerwall," Tesla, 2016. [Online]. Available: https://www.tesla.com/en_GB/powerwall.
- [93] "ABB REACT-3.6/4.6-TL," ABB, 2016. [Online]. Available: http://new.abb.com/power-converters-inverters/solar/photovoltaic-energystorage/react-3-6kw-4-6kw.
- [94] "LG Chem Residential Energy Storage RESU6.4EX," LG, 2016. [Online]. Available: http://www.lgchem.com/global/ess/ess/product-detail-PDEC0001.
- [95] "Sunny Boy 3600/5000 Smart Energy The perfect combination of PV inverter and battery," SMA, 2016. [Online]. Available: http://www.sma.de/en/products/solarinverters/sunny-boy-3600-5000-smartenergy.
- [96] UK: Moixa launches affordable home solar+storage kit: PV Magazine, 2016. [Online]. Available: https://www.pv-magazine.com/2016/10/05/uk-moixalaunches-affordable-home-solarstorage-kit_100026366/. Accessed on February 2017.
- [97] J. Deign, Moixa wants to install a million batteries by 2020: Energy Storage Report, 2016. [Online]. Available: http://energystoragereport.info/moixa-wants-install-million-batteries-2020/. Accessed on February 2017.
- [98] H. Wirth, "Recent Facts about Photovoltaics in Germany," Germany: Fraunhofer ISE, 2017. [Online]. Available: https://www.ise.fraunhofer.de/content/dam/ise/en/documents/publications/stu dies/recent-facts-about-photovoltaics-in-germany.pdf.
- [99] Feed-in tariff rates are declining but the overall benefits of solar panels are still very attractive: EDF Energy, 2014. [Online]. Available: https://www.edfenergy.com/for-home/energy-efficiency-news/feed-in-tariffs-rates-declining-decc. Accessed on February 2017.

- [100] Feed-in Tariff (FIT): Tariff Table 1 April 2016 PV Only, United Kingdom: Office of Gas and Electricity Markets (Ofgem), 2016. [Online]. Available: https://www.ofgem.gov.uk/system/files/docs/2016/04/01_april_2016_tariff_t able.pdf. Accessed on February 2017.
- [101] S. Brett, The German Energy Storage Market 2016-2021: The Next Energy Transition: GTM Research, 2016. [Online]. Available: https://www.greentechmedia.com/research/report/the-german-energy-storagemarket-2016-2021.
- [102] J. Rotstein, Technical Requirements for Renewable Energy Systems Connected to the Low Voltage (LV) Network via Inverters, Australia: Horizon Power, 2015. [Online]. Available: https://horizonpower.com.au/media/1121/technical-requirements-forrenewable-energy-systems-connected-to-the-low-voltage-lv-network-viainverters.pdf. Accessed on February 2017.
- [103] Distributed Energy Resources (Docket No. 2014-0192) PUC Reforms Energy Programs to Support Future Sustainable Growth in Hawaii Rooftop Solar Market, Hawaii: Hawaii Public Utilities Commission (PUC). [Online]. Available: http://puc.hawaii.gov/wp-content/uploads/2015/10/DER-Phase-1-DO-Summary.pdf. Accessed on February 2017.
- [104] EUROPA Strategic Energy Technologies Information System (SETIS), 2017. [Online]. Available: https://setis.ec.europa.eu/energy-research/projectslisting. Accessed on February 2017.
- [105] Low Carbon Networks Fund: Ofgem, 2015. [Online]. Available: https://www.ofgem.gov.uk/electricity/distribution-networks/networkinnovation/low-carbon-networks-fund. Accessed on February 2017.
- [106] £8 million boost for energy storage innovation: Department of Energy & Climate Change (DECC), 2014. [Online]. Available: https://www.gov.uk/government/news/8-million-boost-for-energy-storageinnovation. Accessed on February 2017.
- [107] M. Rawson and E. Sanchez, Sacramento Municipal Utility District -Photovoltaic and Smart Grid Pilot at Anatolia: California Energy Commission, 2015. [Online]. Available: http://www.energy.ca.gov/2015publications/CEC-500-2015-047/CEC-500-2015-047.pdf. Accessed on February 2017.
- [108] M. Rawson, SMUD PV and Smart Grid Pilot at Anatolia: Sacramento Municipal Utility District, 2011. [Online]. Available: https://www1.eere.energy.gov/solar/pdfs/highpenforum1-14_rawson_smud.pdf. Accessed on February 2017.
- [109] Case Study R510 Residential Energy Storage Newington Smart Home Trial: Redflow - Advanced Energy Storage, 2012. [Online]. Available: http://redflow.com/wp-content/uploads/2012/10/Newington-Smart-Home-R510-Case-Study.pdf. Accessed on February 2017.
- [110] Sola Bristol Project Closedown Report: Western Power Distribution (WPD), 2016. [Online]. Available: https://www.ofgem.gov.uk/system/files/docs/2016/04/sola_bristol_closedown _report_final.pdf. Accessed on February 2017.

- [111] V. Muenzel, I. Mareels, J. d. Hoog, A. Vishwanath, S. Kalyanaraman, and A. Gort, "PV generation and demand mismatch: Evaluating the potential of residential storage," in 2015 IEEE Power & Energy Society Innovative Smart Grid Technologies Conference (ISGT), 2015, pp. 1-5.
- [112] C. Long and L. F. Ochoa, "Voltage Control of PV-Rich LV Networks: OLTC-Fitted Transformer and Capacitor Banks," *IEEE Transactions on Power Systems*, vol. 31, no. 5, pp. 4016-4025, 2016.
- [113] P. Esslinger and R. Witzmann, "Regulated distribution transformers in lowvoltage networks with a high degree of distributed generation," in *Innovative Smart Grid Technologies (ISGT Europe), 2012 3rd IEEE PES International Conference and Exhibition on*, 2012, pp. 1-7.
- [114] A. Ballanti and L. F. Ochoa, "WP2 Part A Final Report 'Off-Line Capability Assessment'," The University of Manchester, Manchester, UKSept. 2015, Available: https://goo.gl/b8azYr.
- [115] J. J. Grainger and W. D. Stevenson, "Transformers," in *Power System Analysis*New York: McGraw-Hill, 1994, p. 816.
- [116] Automatic Voltage Regulator Function Block, Finland: ABB, 2007. [Online]. Available: https://library.e.abb.com/public/24f01501a3907fb3c125737500379a3f/appl_ RET54_COLTC_756421_ENa.pdf. Accessed on February 2017.
- [117] A. Berger, M. Hennig, Ko, x, and C. Rner, "Voltage control in smart distribution grids-Overview and practical experience of available solutions," in *Electricity Distribution (CIRED 2013), 22nd International Conference and Exhibition on*, 2013, pp. 1-4.
- [118] C. Korner, M. Hennig, K. Handt, and R. Schmid, "Gaining experience with a regulated distribution transformer in a smart grid environment," in *Integration of Renewables into the Distribution Grid, CIRED 2012 Workshop*, 2012, pp. 1-4.
- [119] F. Kupzog, H. Brunner, W. Pruggler, T. Pfajfar, and A. Lugmaier, "DG DemoNet-Concept - A new Algorithm for active Distribution Grid Operation facilitating high DG penetration," in 2007 5th IEEE International Conference on Industrial Informatics, 2007, vol. 2, pp. 1197-1202.
- [120] M. Stifter *et al.*, "DG DemoNet validation: Voltage control from simulation to field test," in 2011 2nd IEEE PES International Conference and Exhibition on Innovative Smart Grid Technologies, 2011, pp. 1-8.
- [121] A. Einfalt, A. Lugmaier, F. Kupzog, and H. Brunner, "Control strategies for smart low voltage grids: The project DG DemoNet-Smart LV Grid," in *CIRED 2012 Workshop: Integration of Renewables into the Distribution Grid*, 2012, pp. 1-4.
- [122] C. Reese, C. Buchhagen, and L. Hofmann, "Voltage range as control input for OLTC-equipped distribution transformers," in *Transmission and Distribution Conference and Exposition (T&D), 2012 IEEE PES,* 2012, pp. 1-6.
- [123] C. Reese, C. Buchhagen, and L. Hofmann, "Enhanced method for voltage range controlled OLTC-equipped distribution transformers," in 2012 IEEE Power and Energy Society General Meeting, 2012, pp. 1-8.

- [124] Y. P. Agalgaonkar, B. C. Pal, and R. A. Jabr, "Distribution Voltage Control Considering the Impact of PV Generation on Tap Changers and Autonomous Regulators," *Power Systems, IEEE Transactions on*, vol. 29, no. 1, pp. 182-192, 2014.
- [125] R. Kabiri, D. G. Holmes, B. P. McGrath, and L. G. Meegahapola, "LV Grid Voltage Regulation Using Transformer Electronic Tap Changing, With PV Inverter Reactive Power Injection," *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 3, no. 4, pp. 1182-1192, 2015.
- [126] M. Armendariz, D. Babazadeh, D. Brodén, and L. Nordström, "Strategies to improve the voltage quality in active low-voltage distribution networks using DSO's assets," *IET Generation, Transmission & Distribution*, vol. 11, no. 1, pp. 73-81, 2017.
- [127] (2015). *EFR: European Radio Ripple Control*. Available: http://www.efr.de/en/efr-system/
- E. Szabó, Control and devices for the feed-in of renewable energy generating systems into the distribution network: Elektrotechnika, 2013. [Online]. Available: http://www.prolan.hu/wp-content/uploads/2014/01/14-16ANGOL_web.pdf. Accessed on February 2017.
- [129] Z. Liposcak and M. Boskovic, "Survey of smart metering communication technologies," in *EUROCON*, 2013 IEEE, 2013, pp. 1391-1400.
- [130] Radio Ripple Control Receiver (LCR544), Germany: Elster GmbH. [Online]. Available: http://www.elster.com/assets/products/products_elster_files/LCR544_Flyer_ E.pdf. Accessed on February 2017.
- [131] Radio Ripple Control Receivers FTY23x/24x/26x, Switzerland: Landis+Gyr AG, 2014. [Online]. Available: http://www.landisgyr.it/webfoo/wpcontent/uploads/2012/09/LM_FTY-Product-News-Flyer-ENb.pdf. Accessed on February 2017.
- [132] BPT-S 5 Hybrid Storage Solution, Germany: Bosch, 2016. [Online]. Available: http://www.bosch-powertec.com/uk/bpte/produkte/storage_solutions/bpt_s_5_hybrid/vs_5_hybrid.
- [133] J. Gifford. (2015) Offloading the grid. PV Magazine. 15. Available: http://www.pv-magazine.com/fileadmin/PDFs/pvmagazine_Storage_Special_Jul_2015.pdf
- [134] F. Marra *et al.*, "A Decentralized Storage Strategy for Residential Feeders With Photovoltaics," *IEEE Transactions on Smart Grid*, vol. 5, no. 2, pp. 974-981, 2014.
- [135] A. Becker *et al.*, "Electricity Storage Systems in Medium- and Low-Voltage Networks," in *International ETG Congress 2015; Die Energiewende Blueprints for the new energy age; Proceedings of*, 2015, pp. 1-8.
- [136] T. Guena and P. Leblanc, "How Depth of Discharge Affects the Cycle Life of Lithium-Metal-Polymer Batteries," in *INTELEC 06 Twenty-Eighth International Telecommunications Energy Conference*, 2006, pp. 1-8.
- [137] B. Zhao, X. Zhang, J. Chen, C. Wang, and L. Guo, "Operation Optimization of Standalone Microgrids Considering Lifetime Characteristics of Battery

Energy Storage System," *IEEE Transactions on Sustainable Energy*, vol. 4, no. 4, pp. 934-943, 2013.

- [138] "Electrical Energy Storage White Paper," Switzerland: International Electrotechnical Commission (IEC), 2012. [Online]. Available: http://www.iec.ch/whitepaper/pdf/iecWP-energystorage-LRen.pdf.
- [139] "Aspen 24S-83 Battery," Aquion Energy, 2016. [Online]. Available: http://aquionenergy.com/homeowners/aspen-home-batterystorage/#Aspen_48S_Battery.
- [140] Tarifs reglementes de vente (EN: Regulated tariffs), France: Commission

de régulation de l'énergie (CRE), 2012. [Online]. Available: http://www.cre.fr. Accessed on March 2017.

- [141] I. Richardson and M. Thomson. (2011). Integrated domestic electricity demand and PV micro-generation model. Available: https://dspace.lboro.ac.uk/2134/7773
- [142] Standard EN50160 Voltage characteristics of electricity supplied by public electricity networks, 2013.
- [143] N. Maskey, S. Horsmanheimo, and L. Tuomimaki, "Analysis of latency for cellular networks for smart grid in suburban area," in *Innovative Smart Grid Technologies Conference Europe (ISGT-Europe), 2014 IEEE PES, 2014, pp.* 1-4.
- [144] R. C. Dugan and T. E. McDermott, "An open source platform for collaborating on smart grid research," in *Power and Energy Society General Meeting*, 2011 IEEE, 2011, pp. 1-7.
- [145] M. R. GMBH, GridCon, Transformer-The voltage regulated distribution transformer, Germany: Maschinenfabrik Reinhausen GMBH, 2015. [Online]. Available: http://www.reinhausen.com/desktopdefault.aspx/tabid-1516/1835_read-4652/. Accessed on March 2017.

University Library



MINERVA A gateway to Melbourne's research publications

Minerva Access is the Institutional Repository of The University of Melbourne

Author/s: Procopiou, A

Title: Active Management of PV-Rich Low Voltage Networks

Date:

2017

Citation: Procopiou, A, Active Management of PV-Rich Low Voltage Networks, 2017

Persistent Link: http://hdl.handle.net/11343/219852

File Description: PhD thesis